PUBLIC UTILITIES COMMISSION

505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3298



June 29, 2020

Sephra Ninow Director, Regulatory Affairs Center for Sustainable Energy® 3980 Sherman Street, Suite 170 San Diego, CA 92110 sephra.ninow@energycenter.org

Ray Ortiz
Tariff Manager
Southern California Gas Company
555 W. Fifth Street, GT14D6
Los Angeles, CA 90013-1011
rortiz@socalgas.com

Gary A. Stern, Ph.D.
Managing Director, State Regulatory Operations
Southern California Edison Company
8631 Rush Street
Rosemead, CA 91770
advicetariffmanager@sce.com

Erik Jacobson
Director, Regulatory Relations
Pacific Gas and Electric Company
77 Beale St., Mail Code B10C
P.O. Box 770000
San Francisco, CA 94177
PGETariffs@pge.com

SUBJECT: Center for Sustainable Energy® (CSE) Advice Letter (AL) 110-E/-A, Southern California Edison (SCE) 4192-E/-A, Southern California Gas Company (SoCalGas) AL 5619-G/-A, and Pacific Gas and Electric Company (PG&E) AL 4237-G/-A/5808-E/-A: Proposed Revisions to the Self-Generation Incentive Program Handbook to Further Incorporate Requirements Pursuant to Decision D.20-01-021

Dear Ms. Ninow, Dr. Stern, Mr. Ortiz, and Mr. Jacobson:

Pursuant to California Public Utilities Commission (CPUC) Decision (D.) 20-01-021 Ordering Paragraph (OP) 5, the Self-Generation Incentive Program (SGIP) Program Administrators (PAs) jointly filed Advice Letters (ALs) seeking approval of their proposed revisions to the SGIP Handbook: CSE AL 110-E/-A, SCE AL 4192-E/-A, SoCalGas AL 5619-G/-A, PG&E AL 4237-G/-A/5808-E/-A (collectively, the "Joint SGIP PAs' ALs").

After review and analysis, Energy Division staff (staff) has determined that the Joint SGIP PAs' ALs, as supplemented, are in compliance with OP 5 of D.20-01-21 and are approved, effective June 29, 2020.

On January 27, 2020 the CPUC issued D.20-01-021 (the SB 700 Decision) to authorize SGIP revisions, including annual ratepayer collections of \$166 million from 2020 to 2024, as stipulated in Senate Bill (SB) 700. This decision also clarified and expanded SGIP eligibility rules for the Equity Resiliency Budget. OP 5 of the SB 700 Decision directed the SGIP PAs to submit a Tier 2 Implementation AL no later than 90 days from the Decision's issuance to propose modifications to

June 29, 2020 Page 2 of 2

the SGIP Handbook that would implement all program and budget modifications adopted in the SB 700 Decision, not included in the other Advice Letter filings directed by the Decision.¹

The California Solar & Storage Association (CALSSA), the California Energy Storage Alliance (CESA), and Tesla, Inc. (Tesla) timely filed protests on May 5, 2020.

On May 12, 2020, CSE filed a reply on behalf of the SGIP PAs to the above-stated protests, followed by a joint supplemental AL filed on June 5, 2020.

Attachment 1 contains a detailed discussion of the procedural background, protests, reply, and staff's determination that the Joint SGIP PAs' ALs are compliant with D.20-01-21, OP 5.

Please contact Asal Esfahani of the Energy Division at <u>ae3@cpuc.ca.gov</u> if you have any questions.

Sincerely,

Som FOR

Edward Randolph

Deputy Executive Director for Energy and Climate Policy/

Director, Energy Division

cc: Alex Morris, Executive Director, California Energy Storage Alliance Scott Murtishaw, Senior Advisor, California Solar & Storage Association Andy Schwartz, Managing Policy Advisor, Tesla, Inc.

¹ Joint PG&E AL 4191-G-A/5714-E-A et al., approved February 26, 2020, implemented program revisions adopted in D.20-01-021 specific to Equity Resiliency Budget residential customers; and Joint PG&E AL 4218-G/-A/5764-E/-A et al., approved April 9, 2020, implemented revisions adopted in D.19-09-027 and D.20-01-021 specific to the Equity Resiliency Budget non-residential customers.

Attachment 1 Staff Technical Review and Disposition

Background

The CPUC adopted D. 20-01-021 ("SB 700 Decision") on January 27, 2020. This decision authorized ratepayer collections of \$166 million annually through 2024, allocated newly authorized funds across eligible SGIP technologies and customer sectors, established program and incentive modifications to expand and clarify eligibility requirements for customers impacted by the Public Safety Power Shutoff (PSPS) or other outage events, and modified the SGIP PAs' administrative budget allocations and requirements. OP 5 of the SB 700 Decision directed the SGIP PAs to submit a Tier 2 Implementation AL no later than 90 days from its issuance to propose modifications to the SGIP Handbook that would implement all program and budget modifications adopted in the SB 700 Decision. These modifications were in addition to those not included in the other Advice Letter filings directed by the Decision.

On April 15, 2020, pursuant to the SB 700 Decision, OP 5, the SGIP PAs filed CSE AL 110-E, SCE AL 4192-E, SoCalGas AL 5619-G, PG&E AL 4237-G/5808-E. On June 5, 2020, the SGIP PAs filed supplemental CSE AL 110-E/-A, SCE AL 4192-E/-A, SoCalGas AL 5619-G/-A, PG&E AL 4237-G/-A/5808-E/-A (the initial and supplemental ALs are collectively referred to as the "Joint SGIP PAs' ALs").

Protests to the Initial AL Filing and the PAs' Reply

Equipment Pre-approval Requirement

In response to the SB 700 Decision's directive to the SGIP PAs to ensure standard application review and incentive payment processing times of approximately 45-60 days,² the Joint SGIP PAs' AL includes, among other things, a proposed requirement for equipment pre-approval by the SGIP PAs' technical consultant Alternative Energy Systems Consulting (AESC) for an applicant to submit a reservation request. CESA's protest asserts that the SB 700 Decision does not explicitly order this action, and all three protestants (CALSSA, CESA, and Tesla) argue that the proposed equipment pre-approval requirement would hinder, rather than facilitate the application review process. CALSSA argues that requiring equipment approval prior to accepting reservation request forms may encourage program participants to submit all their potential equipment configurations, creating an undue administrative burden for the PAs and detracting from the time needed to review applications. In addition, Tesla's protest expresses concern that the proposed technical review process lacks a well-defined timeline for completion and could, therefore, result in some projects that usenew equipment missing the opportunity to reserve funding.

CALSSA's and CESA's protests provide recommendations for streamlining the proposed equipment pre-approval process. CALSSA recommends that the PAs allow potential applicants to apply with new equipment not yet on the pre-approved list. According to their proposal, once such an application is received, the time to review and vet the equipment would not count toward the PAs' overall application review target set by the CPUC in the SB 700 Decision, and the application processing timeline would only begin once the equipment is approved. CESA, on the other hand,

_

² D.20-01-021, OP 32 at 106.

recommends the additional equipment verification step be completed prior to the submission of the incentive claim form, "as it is done today".³

In their reply to protests, the SGIP PAs clarify that their equipment pre-approval proposal is in direct response to the SB 700 Decision's established standards for incentive application processing time and will enable improved timelines for reservation request form review. The PAs, however, concur that the lack of a clear timeline for the review and approval of new equipment is problematic. The PAs' reply confirms their willingness to work with industry to develop clear descriptions of the data needed and the appropriate timelines for equipment pre-approval reviews.

In their reply, the PAs express support for CALSSA's recommendation for streamlining the proposed equipment pre-approval process. Similarly, the PAs proposed a scheme that would allow projects using equipment that has not been pre-approved to still submit applications, but they would create a new "review status" for reservation request forms specific to applications using new equipment. This would ensure time spent on technical equipment review does not count against a PA's application processing target.

The PAs disagree with CESA's recommendation for an alternative to the proposed equipment preapproval requirement and clarify that the existing rules subject all equipment to review before a reservation is issued. The PAs contend that this enables appropriate incentive amounts to be determined early in the process, at the reservation stage, thus eliminating the risk that inflated or inappropriately reserved incentive funds are tied up in projects ultimately found to be ineligible, and thus, withheld from other eligible applicants.

Definition of "Discrete Public Safety Power Shutoff (PSPS)4 Event"

The SB 700 Decision directs the SGIP PAs to include a working definition of "discrete PSPS event" in the Joint SGIP PAs' ALs.⁵ The initial Joint SGIP PAs' ALs proposes two different definitions of "discrete PSPS event", as the definition used in the PG&E and SCE territories is different from the definition used in San Diego Gas & Electric (SDG&E) territory. CALSSA, CESA, and Tesla argue in their protests that a single statewide definition should be used to provide more simplicity and consistency for the SGIP Equity Resiliency Budget eligibility requirements.

In their reply, the PAs agreed to implement a statewide definition for "discrete PSPS event", for SGIP purposes only, and proposed that a supplemental filing would use the definition currently used in the PG&E and SCE service territories.

Tesla's protest letter further requests that the proposed SGIP Handbook omit the final sentence in the PG&E and SCE definition that states "PSPS event are counted per calendar year". Tesla argues that keeping this element of the definition in SGIP is inconsistent with the intent of the SB 700 Decision, which provided eligibility for Equity Resiliency Budget incentives to customers that have experiences two or more PSPS outages.

³ CESA Protest at 4.

⁻

⁴ The State's investor-owned electric utilities, notably PG&E, SCE, and San Diego Gas & Electric (SDG&E), may shut off electric power, referred to as "de-energization" or Public Safety Power Shut-offs (PSPS), to protect public safety under California law, specifically PU Code Sections 451 and 399.2(a).

⁵ D.20-01-021 at 44.

In their reply, the PAs agreed to remove the phrase from their proposed SGIP Handbook.

CESA's and Tesla's protests also seek clarification on how the adopted PSPS definition will apply to electric customers of a municipal utility, specifically the Los Angeles Department of Water and Power (LADWP). Should LADWP adopt a definition specific to its service territory, Tesla's protest suggests the PAs accept program applications to the Equity Resiliency Budget from LADWP customers in SoCalGas's service territory.

In their reply, the PAs agree that "when or if made available, the applicable electric provider's definition of a discrete PSPS event will be used for SGIP eligibility purposes where an application is submitted in unbundled territories." The PAs, however, note that LADWP's Wildfire Exposure Mitigation Plan, released as part of their Power System Update in July 2019, does not include power shutoffs, in accordance with LADWP's finding in September 2018 that their "overhead electrical lines and equipment do not pose a significant risk of causing catastrophic wildfire."⁷

Waitlist Procedures for Residential Steps 6 and 7

OP 12 of the SB 700 Decision orders the SGIP PAs to implement a "soft target" on the general market residential incentive budget, such that at least half of these funds will be used by residential customers living in Tier 3 or Tier 2 High Fire Threat Districts (HFTDs) or by residential customers who have been subject to two or more discrete PSPS events. The initial Joint SGIP PAs' ALs proposes to comply with this order by pausing acceptance of SGIP applications from residential customers who do not live in a Tier 3 or Tier 2 HFTD or who were not subject to two or more discrete PSPS events, once reservation requests from these types of customers equal 50 percent of available funds in each residential incentive step. The PAs also proposed that once the residential step 6 incentive budget reaches the said 50 percent soft target, waitlisted projects will be rejected and the waitlist closed to new applications.

CALSSA and Tesla protest the proposed waitlist procedure arguing that the PAs do not provide a good rationale for their proposal, which could lead to an increased administrative burden both for the PAs and for potential applicants who would have to resubmit their applications once step 7 opens. CALSSA and Tesla's protests recommend, instead, to allow projects on the waitlist to hold "their place in line" once the new incentive step opens. They argue that this alternative procedure would allow for a smoother transition between the incentive steps.

In their reply to these protests, the PAs cite the SB 700 Decision's use of a "soft target" rather than a distinct budget carve-out as an indication of the CPUC's intent to encourage installations in areas most in need of the resiliency benefits that energy storage can provide. Therefore, the PAs argue, allowing applications for customers without urgent resiliency needs as defined in the Decision would counter the Decision's intent. The PAs' reply requests that the CPUC adopt the waitlist rules as proposed in the Joint SGIP PAs' ALs.

Limits on the Number of Requests for Clarification or Missing Information

Tesla opposes the PAs' proposal to limit the number of suspensions allowed per project. Tesla expresses concern that the PAs' proposed SGIP Handbook changes regarding limits on the number

⁶ SGIP PAs Reply at 5.

⁸ CALSSA Protest at 5.

of suspensions are "overly restrictive" and could lead to the inappropriate cancellation of applications, where applicants, despite their best efforts and perhaps due to project complexities, are not able to provide requested information or address a deficiency. Tesla further argues that under the current SGIP Handbook, the PAs already have sufficient authority to determine when an abuse of process has occurred that justifies project disqualification.

In their reply, the PAs agree with Tesla that their intent in putting forth this proposal is not "to remove those program participants who act in good faith," but rather to provide clear language and rules to prevent "repeated requests for the same piece of information without reasonable and satisfactory progress on behalf of the applicant to provide the missing or incomplete information." Therefore, the PAs contend, the proposed language should not be interpreted as preventing communication between applicants and the PAs in the process of determining how best to provide missing information or documents.

Removal of Investment Tax Credit (ITC) Adjustment

CESA's protest requests that the PAs explicitly remove the Investment Tax Credit (ITC) adjustment for general market large-scale projects using equipment purchased after December 31, 2021, as was directed in D.20-01-021. CESA further states that the Joint SGIP PAs' ALs maintains the differential incentive structure for ITC and non-ITC general market large-scale storage projects. As an example, CESA points to the Table 3.1.2 of the proposed revised SGIP Handbook, which shows different incentive rates for ITC and non-ITC general market large-scale storage projects receiving step 5 incentives. CESA requests that the PAs remove this differential for ITC and non-ITC projects as shown in the table.

The PAs' reply confirms that per the SB 700 Decision, the language in the initial AL filing on the removal of the ITC adjustment for general market large-scale projects will apply, regardless of the incentive step open at the time of incentive reservation, for all equipment purchased before December 31, 2021.¹³

Multifamily Greenhouse Gas (GHG) Rules

CALSSA's protest disagrees with the PAs' proposal to apply non-residential GHG rules and operating requirements to multifamily energy storage projects that allocate 50 percent or more of their Virtual Net Energy Metering (VNEM) credits to the common area load account and that have a common load meter that is on a commercial rate. CALSSA argues that, regardless of whether the meter is on a residential or commercial rate, multifamily residential projects should be classified strictly as residential projects subject to residential GHG rules, because they serve "the residential needs of the tenants." CALSSA then cites precedent in federal housing law by pointing to the Low Income Housing Tax Credit program, whereby the common areas of a property are eligible for federal housing credits because they are deemed as "necessary for and appurtenant to the operation of the residential property". 15

¹⁰ SGIP PAs Reply at 4.

⁹ Tesla Protest at 2.

¹¹ D.20-01-021, OP 9 at 99-100.

¹² Joint SGIP PAs' ALs, Appendix A at 27.

¹³ Please see CSE 110-E et al at 6 and Appendix A at 27.

¹⁴ CALSSA Protest at 6.

¹⁵ CALSSA Protest at 6 referring to IRS Revenue Ruling 2004-82 at 1 − 3. https://www.irs.gov/pub/irs-drop/rr-04-82.pdf.

In their reply, the PAs contend that the proposed rules are in line with SGIP-specific policy, as established in D.19-08-001, "whereby the applicable GHG rules are determined by primary use and the type of electric rate of the benefitting account." Moreover, the PAs argue that their proposal provides for clarity of program rules and "eliminates[s] ambiguity and burden of verification in cases where commercial spaces may be co-located with multifamily common areas." ¹⁷

Equity Resiliency Budget Eligibility for Customers Who Depend on On-site Electric Pumps for Water CALSSA's protest requests that the PAs provide additional clarity on eligibility requirements for customers who rely on electric well pumps for their water service. CALSSA adds that since the introduction of this eligibility option in the SB 700 Decision, it has received numerous member inquiries into the details of the policy. CALSSA notes that there are several configurations of water service for households without water utility service, and it is not clear if all or some of these configurations meet the eligibility criterion established in the SB 700 Decision.

The PAs reply that they have also received requests for clarification on this rule and commit to working with Energy Division and stakeholders to provide further guidance to program participants. The PAs state, however, that, at this time, the request is out of scope since the SB 700 Decision directed the PAs to address Equity Resiliency Budget eligibility rules in a previous Advice Letter filed on February 18, 2020.

Errors and Omissions in the Proposed SGIP Handbook

CALSSA's protest includes a number of items categorized as "Miscellaneous Typos, Omissions, and Errors." 18

In their reply, the PAs separate these items into three categories. First, the PAs agree to correct errors found meriting correction in their supplemental filing. However, the PAs consider some items as constituting recommendations for general Handbook improvements, rather than typographical errors. The PAs state that they may consider these recommendations in a "future Handbook streamlining or organization improvement effort."¹⁹. Finally, the PAs note that there are additional items raised by CALSSA that the PAs identify as requests for substantive changes or interpretation of Decision language, which the PAs deem to be out of scope under the Joint SGIP PAs' ALs.

Supplemental Filing

On June 5, 2020, CSE, on behalf of the SGIP PAs, filed a supplemental advice letter to the Joint SGIP PAs' ALs.

In their reply to protests discussed above, the PAs express support for CALSSA's recommendation to allow projects using equipment that has not been pre-approved to submit applications. These projects will be designated with a new reservation request form review status specific to applications using new equipment, so that time spent in technical review does not count against a PA's

¹⁶ SGIP PAs' Reply at 7.

¹⁷ Ibid.

¹⁸ CALSSA Protest at 8.

¹⁹ SGIP PAs Reply at 8.

application processing timeline target. Accordingly, the PAs submitted a revision to Section 5.4.1 of the SGIP Handbook in the supplemental to the Joint SGIP PAs' ALs to clarify this point.

In response to protests by CALSSA, CESA, and Tesla, the PAs worked with SDG&E to align SDG&E's definition of a discrete PSPS event with that of PG&E and SCE, for SGIP purposes only. The supplemental to the Joint SGIP PAs ALs includes a statewide definition of a discrete PSPS event that applies to all three electric investor-owned utilities, for SGIP purposes only.

As noted in the reply, the PAs agree with Tesla's protest that the discrete PSPS event definition in the initial Joint SGIP PAs' ALs should be modified to exclude the final statement that "PSPS events are counted per calendar year." The supplemental filing revises the SGIP Handbook definition accordingly.

After reviewing items in CALSSA's protest categorized as miscellaneous typos, omissions, or errors, the PAs found some identified errors that merit correction and included those in their proposed revised SGIP Handbook in the supplemental to the Joint SGIP PAs' ALs.

Disposition

Staff finds that the Joint SGIP PAs' ALs as supplemented adequately address issues raised by parties in protests on the initial AL filing.

Staff finds that the PAs' proposal for streamlining SGIP application review times is in line with the SB 700 Decision's order that the PAs limit application processing times to approximately 45-60 days. The PAs' proposed measures, namely the equipment pre-approval requirement and rules limiting the number of application suspensions, adhere to steps taken in the SB 700 Decision to establish application processing targets to create greater transparency in the administration of SGIP incentives.

Staff agrees with the PAs that the rules established in D.19-08-001, which were designed to help ensure SGIP energy storage projects reduce GHG emissions, were determined, in large part, based on the type of electric rate of the benefitting account. Under the GHG rules, a key requirement for SGIP residential storage projects is to take service under an SGIP-approved time-varying rate, where available, and all current approved time-varying rates under SGIP are residential rates. Multifamily energy storage projects that allocate 50 percent or more of their VNEM credits to the common area load account and that have a common load meter that is on a commercial rate cannot adhere to this requirement. Therefore, staff finds the PAs' proposal to apply SGIP's non-residential GHG rules and operating requirements to these types of projects is reasonable and in accordance with D.19-08-001.

Staff finds the PAs' proposal for residential incentive steps 6 and 7 waitlist procedure is reasonable, given that the SB 700 Decision orders a soft target to be applied to the residential incentive budget to encourage installations in areas with the most urgent critical resiliency needs. The Decision is silent on the details of the implementation of this order. As the CPUC has delegated the day-to-day implementation of the SGIP to the PAs, it is reasonable that the PAs determine the most streamlined and effective means for compliance with newly established SGIP policy. In particular,

-

²⁰ D.20-01-021, OP 32 at 106.

we find the PAs' approach of requiring SGIP developers to focus on customers with urgent resiliency needs as a prerequisite for the next step to open align with the directives in the SB 700 Decision related to general market residential steps 6 and 7. If issues arise during implementation, however, Energy Division retains its right to work with the PAs to revise the waitlist procedure for future steps.

Staff agrees with the PAs that the language in the proposed SGIP Handbook revisions regarding the removal of the ITC incentive adjustment is correct. The SB 700 Decision did not tie the removal of the ITC adjustment to any given incentive step, but rather specified that the ITC adjustment should be removed for equipment purchased after December 31, 2021, to coincide with the ITC step-down to 10 percent. The ITC adjustment removal is not tied to any given incentive step, and it is possible that, depending on timing, different projects in the same incentive step will be subject to the ITC adjustment, or not, depending on when their equipment is purchased.

Staff appreciate CALSSA's request for further clarification of the Equity Resiliency Budget eligibility option for customers who rely on electric well pumps for their water service. Like CALSSA and the PAs, Energy Division has received a number of inquiries into the details of this eligibility pathway. The PAs should note that the SB 700 Decision defines eligible households as those "that rely on electric-pump wells for their water supply" and whose critical resiliency needs include water for drinking, sanitation, and fire response during a PSPS or other power outage event. ²¹ Concerning the question of how various water system configurations fit under this eligibility rule, the Decision is silent. Therefore, Energy Division and the PAs will work with stakeholders to ascertain whether further clarification on the current rules are needed. Staff recommends the PAs include this topic on the agenda for the upcoming SGIP Quarterly Workshop in July 2020.

Staff appreciates CALSSA's efforts in identifying errors and omissions in the SGIP Handbook and equally welcomes the PAs' initiative in reviewing and executing those recommendations they find valid. After reviewing CALSSA's comments in this regard, however, staff finds that certain suggestions deemed out of scope by the PAs would indeed help clarify and correct SGIP rules in the Handbook. Therefore, staff recommends that the PAs review both CALSSA's recommendations and the Handbook as a whole in order to submit an advice letter to streamline and improve the clarity of the Handbook within 60 days of the effective date of this disposition letter.

-

²¹ D.20-01-021 at 45.



June 5, 2020

Center for Sustainable Energy 3980 Sherman Street, Suite 170 San Diego, CA 92110 Tel 858-244-1177 Fax 858-244-1178 EnergyCenter.org

Advice No. 110-E-A (Center for Sustainable Energy®)

Advice 4192-E-A (Southern California Edison Company U 338-E)

Advice 5619-A (Southern California Gas Company U 904-G)

Advice Letter 4237-G/5808-E-A (Pacific Gas and Electric Company U 39 M)

PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA ENERGY DIVISION

SUBJECT: Supplemental: Proposed Revisions to the Self-Generation Incentive Program Handbook to Further Incorporate Requirements Pursuant to Decision (D.) 20-01-021

PURPOSE

Pursuant to Ordering Paragraph (OP) 5 of Decision (D) 20-01-021 (Decision), Center for Sustainable Energy® (CSE), on behalf of the Self-Generation Incentive Program (SGIP) Program Administrators (Joint PAs or PAs),¹ hereby submits this Supplemental Advice Letter (Supplemental AL) to the Joint PA Advice Letter CSE Advice Letter 110-E-A,² (Joint PA AL) submitted on April 15, 2020, which proposed revisions to the SGIP Handbook (Handbook) to incorporate further program changes pursuant to the Decision not previously included in the Advice Letters implementing the Equity and Equity Resiliency Budgets for residential customer applications by March 1, 2020 and non-residential customers by April 1, 2020.

In their May 12, 2020 Reply to Protests to the Joint PA AL (Reply), the Joint PAs agreed to submit this Supplemental AL at a later date to incorporate further revisions as discussed in the Reply. In addition, this Supplemental AL corrects several errors, typos, and/or omissions. As

¹ The SGIP PAs are: CSE, in the service territory of San Diego Gas & Electric Company (SDG&E); Southern California Edison Company (SCE); Southern California Gas Company (SoCalGas); and Pacific Gas and Electric Company (PG&E).

² CSE AL 110-E / SCE AL 4192-E / SoCalGas AL 5619 / PG&E AL 4237-G/5808-E.

such, this Supplemental AL supplements the original Joint PA AL but does not replace it in its entirety.

BACKGROUND

The SGIP was established in 2001 by the California Public Utilities Commission (Commission) in D.01-03-073 in response to Assembly Bill (AB) 970 (Stats. 2000, ch. 329). On October 13, 2017, the Commission issued D.17-10-004 and established the SGIP Equity Budget, requiring the SGIP PAs to reserve 25 percent of the energy storage incentive budget, beginning with Incentive Step 3, and allocate it to the SGIP Equity Budget for qualifying projects.

In 2018, Senate Bill (SB) 700 (Stats. 2018, ch. 839) was enacted and set in motion a new series of modifications to the SGIP. Among other things, SB 700 authorized the Commission to extend annual collections for the SGIP to December 31, 2024 and to extend program administration to January 1, 2026. To implement SB 700, the Commission sought Party input regarding necessary program modifications to the SGIP, including: 1) overall collection levels for years 2020-2024; 2) funding allocations among technology and customer sectors; 3) incentive levels; 4) incentive step-down structure; 5) administrative budget; 6) resiliency; 7) proposals from the San Joaquin Valley proceeding; 8) grid support; and 9) thermal energy storage and coordination with the Commission's new building decarbonization rulemaking.

Subsequently, the Commission issued D.19-09-027 on September 18, 2019, establishing a new Equity Resiliency Budget for the SGIP and adjusting eligibility rules and incentive levels for participants. D.19-09-027 ordered the SGIP PAs to carry over accumulated unused SGIP funds collected prior to or during 2017-2019 to fund new budgets and directed the following budget transfers:

- \$100 million to be transferred from generation technology funds and allocated to the new Equity Resiliency Budget;
- \$4 million to be transferred from large-scale storage technology funds and allocated to an equity Heat Pump Water Heater (HPWH) Budget; and
- \$10 million to be transferred from PG&E's and SCE's accumulated unused Non-Residential Equity Storage Budgets and allocated to the San Joaquin Valley Pilot Budget.

The Joint PAs submitted two Advice Letters to carry out the direction in OP 7(a) of D.19-09-027 in order to implement the Equity Resiliency Budget for residential customer applications by March 1, 2020 and non-residential customers by April 1, 2020.

On January 27, 2020, the Commission issued D.20-01-021, authorizing ratepayer collections of \$166 million annually for the years 2020 to 2024 to fund the SGIP consistent with the authorization established by SB 700, prioritizing allocation of 2020 to 2024 collections in accordance with AB 1144 (Stats. 2019, ch. 394) and to benefit customers impacted by Public Safety Power Shutoff (PSPS) events or located in areas of extreme or elevated wildfire risk and adopting further program revisions pursuant to SB 700. OP 5 directed the SGIP PAs to

submit a Tier 2 Implementation Advice Letter no later than 90 days from adoption of the Decision to modify the Handbook to implement all program and budget modifications adopted in the Decision not included in the Advice Letters directed in OP 3 and 4.

Accordingly, on April 15, 2020, the SGIP PAs submitted the Joint PA AL to seek approval of proposed revisions to the Handbook to incorporate further program changes pursuant to the Decision not previously included in the Advice Letters implementing the Equity and Equity Resiliency Budgets for residential customer applications by March 1, 2020 and non-residential customers by April 1, 2020.

On May 5, 2020, California Energy Storage Alliance (CESA), Tesla, Inc. (Tesla), and the California Solar and Storage Association (CALSSA) protested the Joint PA AL.

After further review of the redlined Handbook, the Joint PAs discovered several errors, typos, and/or omissions in need of correction.

SUPPLEMENTAL PROPOSED REVISIONS TO THE SGIP HANDBOOK TO FURTHER INCORPORATE REQUIREMENTS PURSUANT TO THE DECISION AND CORRECT ERRORS, TYPOS, AND/OR OMISSIONS

On May 12, 2020, the Joint PAs agreed to submit a Supplemental AL to incorporate further revisions and correct several errors, typos, and/or omissions. Outlined below are the supplemental proposed revisions to the Handbook, and a revised copy of the Handbook showing these supplemental edits and corrections in redline format is in Appendix A.

Equipment Pre-approval Requirement

While the Joint PAs continue to recommend that equipment be pre-approved by the SGIP technical consultant prior to an Reservation Request Form (RRF) submittal when reasonable to do so, the Joint PAs also support CALSSA's recommendation³ to allow projects using equipment that has not been pre-approved to submit applications and create a new status within RRF review specifically for vetting new equipment so that time spent in technical review does not count against a PA's target review timeline. Doing so will allow program participants to request funds while undergoing a timely review of the proposed equipment. The Joint PAs seek to revise section 5.4.1 of the SGIP Handbook, as follows with red/strikethrough text indicating the revised language:

1. <u>Equipment Specifications</u> (All Projects not currently listed on the SGIP Energy Storage Equipment List)

All reservation requests must list equipment that has been vetted by the SGIP Technical Working Group. Equipment and/or component pairings that have been evaluated by the Technical Working Group can be found on the SGIP Energy Storage Equipment List at www.selfgenca.com. For projects using equipment and/or component pairings not yet evaluated by the Technical Working Group, manufacturer equipment specifications for all major

³ CALSSA Protest at 2.

components of the system, such as the storage component, as well as the inverter, DC-DC converter, controller and/or additional system components when applicable, are required. Rated capacity (kW), energy capacity (kWh), and round-trip efficiency⁴ for the storage system must be provided. to a PA prior to submitting a reservation request form for that project. Equipment specifications may be submitted to any PA via email to the contact information listed on selfgenca.com.

Affected Handbook Section:

5.4.1 Required Documentation for Reservation Request

Single Statewide Definition of "Discrete PSPS Event"

CALSSA, CESA, and Tesla contend that a single statewide definition of a discrete PSPS event should be used, rather than the two definitions the PAs proposed in the Joint PA AL (i.e., one for PG&E's and SCE's service territories and another for SDG&E's service territory).⁵ The Joint PAs agree that a single statewide definition applied to Investor-Owned Utilities (IOUs) would provide greater consistency for SGIP Equity Resiliency eligibility requirements.

In light of the three protests, SDG&E indicated to CSE that it agrees to use the definition provided by PG&E and SCE for SGIP purposes only. Accordingly, the Joint PAs find it appropriate to implement a statewide IOU definition of a discrete PSPS event, for the purposes of SGIP only, using the definition of PG&E and SCE.

The Joint PAs agree with Tesla⁶ that the definition should be modified to exclude the final sentence stating that "PSPS events are counted per calendar year." The SGIP Handbook definition has been modified as follows:

For PG&E and SCE Customers: For the purposes of SGIP, if the utility de-energizes a customer for safety and then restores power after the weather event has passed, this would count as one PSPS event – whether that PSPS event endured for the customer for only a few hours or some number of days. If power is restored for the customer and another weather event subsequently requires that the utility de-energize the same customer again – whether this occurred days, weeks or months later – this would count as the customer's second PSPS event. PSPS events are counted per calendar year.

For SDG&E Customers: A discrete PSPS event is defined as an event with 72 hours or more between concurrent weather events, similar to a storm series.

Affected Handbook Section: Definitions and Glossary

⁵ Tesla Protest at 3-4.

⁶ Tesla Protest at 3.

⁷ In non-IOU territories, the applicable electric provider's definition of a discrete PSPS event will be used for SGIP eligibility purposes.

Errors or Omissions to be Corrected

In CALSSA's Protest,⁸ several items were identified as miscellaneous typos, omissions, or errors to the redlined Handbook. CESA and Tesla also proposed some of the same edits in their Protests. After reviewing each item, the Joint PAs have found some errors merit correction, while other items constitute recommendations. The Joint PAs will correct the errors outlined below. These edits can be found in Attachment A.

- Section 1: Budget
 - Page 14: Correct the figure for the SJV Pilot Budget from \$100 million to \$10 million.
- Section 2: Applications
 - Pages 20 and 22: CESA requests clarification regarding the statement: "An application may not receive multiple requests for clarification or more information for the same item." In an effort to reduce application processing time, the PAs are not required to suspend projects or to contact applicants more than once with a clarification request for the same item.
 - Clarification proposed by the PAs: "An application may not receive multiple requests for clarification or more information for the same <u>missing</u> or <u>uncorrected</u> item."
- Section 3: Incentives
 - Page 27: Add "for small residential storage" after "seven steps."
 - Page 28: Delete the paragraph directly under Table 3.1.2. With the restructuring of the Equity Budget, this remaining text is no longer applicable.
 - Page 28: Add "is reserved" after "general market residential incentive budget" under heading for Section 3.1.2.2.
 - o Page 28: Delete "Residential" in heading of Section 3.1.2.3.
- Section 4: Program Eligibility
 - o Page 36: Under Section 4.1.1.1, add a bullet for "tribal governmental agency".
 - Page 36: Under Section 4.1.1.1, make the following revisions: "Additionally, the project site must be located in a disadvantaged, <u>tribal</u>, or low-income community <u>or document that at least 50% of the census tracts it serves are disadvantaged</u>, <u>tribal</u>, <u>or low-income communities</u>."
- Section 5: Energy Storage Technologies
 - o Page 55: Delete "Equity" from the caption for Table 5.3.3.
 - o Page 55: Correct the incentive duration step-down example as follows:

Example 1: Four-hour system at \$.40/Wh incentive 100 kW discharge rating 400 kWh discharge capacity 400,000 Wh * \$.40/Wh = \$160,000

_

⁸ CALSSA Protest at 8-11.

Example 2: Six-hour system at \$.40/Wh incentive 100 kW discharge rating 600 kWh discharge capacity 400,000 Wh * \$.40/Wh = \$160,000 200,000 Wh * \$.20/Wh = \$80,000 \$160,000 + \$80,000 = \$240,000

- Page 57: Revise Table 5.4.1 to delete "Proof of Income Qualification (Single-family only)" from under Item 6 "Proof of Equity Budget Eligibility (Equity Projects Only)".
- Page 58: Add "or Equity Resiliency incentive" after "applying for a Resiliency Adder" in Item 7.
- Page 59: Delete "All reservation requests must list equipment that has been vetted by the SGIP Technical Working Group" and "to a PA prior to submitting a reservation request form for that project" from Item 3 "Equipment Specifications".
- Page 64: Add "or Equity Resiliency incentive" after "applying for a Resiliency Adder" in Item 7.
- Page 66: In Table 5.4.3, revise Item 3 to conform with edits for page 58 (add "Projects applying for Resiliency adder").
- Page 67: Revise headings of Item 3 to conform with edits for page 58 (add "Projects applying for the Resiliency Adder").

• Appendices

- Appendix C: Revise the text at the top of the page as follows: "The following information (Responses to Questions 1 5) is required by the Developer of the project applying for all energy storage systems with a longer than two-hour discharge duration and all projects with a discharge duration greater than two hours applying for SGIP's Equity Resiliency Budget or General Market Storage Resiliency Adder.
 - Revision proposed by the PAs: "This information is required for all Equity and Non-equity Resiliency projects and all other projects with a discharge duration greater than two hours."

TIER DESIGNATION

Pursuant to General Order (GO) 96-B, Energy Industry Rule 5.2, and the Decision, this Supplemental AL is submitted with a Tier 2 designation.

PROTESTS

The Joint PAs respectfully request that the Commission maintain the original protest period designated in the Joint PA AL pursuant to GO 96-B, General Rule 7.5.1, and not reopen the protest period.

EFFECTIVE DATE

The Joint PAs request that the Joint PA AL and Supplemental AL become effective on regular notice, May 15, 2020, which is the date requested in the Joint PA AL submitted on April 15, 2020.

NOTICE

CSE is providing a copy of this Supplemental AL to service list R.12-11-005.

Sephra A. Ninow, J.D.

Director, Regulatory Affairs

Center for Sustainable Energy®

Syphia A. Minou

Attachments: Appendix A – Revised Redlined SGIP Handbook

Appendix C – Storage Customer Attestation

cc: Service List R.12-11-005



California Public Utilities Commission

ADVICE LETTER



LINLINGTOTILIT	CAU
MUST BE COMPLETED BY UT	ILITY (Attach additional pages as needed)
Company name/CPUC Utility No.: Center for Su	stainable Energy® (CSE)
Utility type: ELC GAS WATER PLC HEAT	Contact Person: Sephra Ninow Phone #: (858) 244-1186 E-mail: sephra.ninow@energycenter.org E-mail Disposition Notice to: sephra.ninow@energycenter.org
EXPLANATION OF UTILITY TYPE ELC = Electric GAS = Gas WATER = Water PLC = Pipeline HEAT = Heat	(Date Submitted / Received Stamp by CPUC) June 5, 2020
Advice Letter (AL) #: 110-E-A	Tier Designation: 2
Subject of AL: Supplemental: Proposed Revi Incorporate Requirements Pursuant	sions to the Self-Generation Incentive Program Handbook to Further to Decision D.20-01-021
Keywords (choose from CPUC listing): Compliant AL Type: Monthly Quarterly Annual Annua	
	on order, indicate relevant Decision/Resolution #:
Does AL replace a withdrawn or rejected AL? I	f so, identify the prior AL: $_{ m No}$
Summarize differences between the AL and th	e prior withdrawn or rejected AL: $ m N/A$
Confidential treatment requested? Yes	√ No
	nation: $\mathrm{N/A}$ vailable to appropriate parties who execute a ontact information to request nondisclosure agreement/
Resolution required? Yes V No	
Requested effective date: 5/15/20	No. of tariff sheets: $_{ m 0}$
Estimated system annual revenue effect (%): $_{ m 0}$	
Estimated system average rate effect (%): 0	
When rates are affected by AL, include attach (residential, small commercial, large C/I, agric	nment in AL showing average rate effects on customer classes ultural, lighting).
Tariff schedules affected: $_{ m N/A}$	
Service affected and changes proposed $^{1:}$ $_{ m SUI}$	PPLEMENTAL: Proposed Revisions to the Self-Generation Incentive
Pending advice letters that revise the same ta	iff sheets: CSE AL 110-E

Protests and all other correspondence regarding this AL are due no later than 20 days after the date of this submittal, unless otherwise authorized by the Commission, and shall be sent to:

CPUC, Energy Division Attention: Tariff Unit 505 Van Ness Avenue San Francisco, CA 94102

Email: <u>EDTariffUnit@cpuc.ca.gov</u>

Name: Sephra Ninow

Title: Director, Regulatory Affairs

Utility Name: Center for Sustainable Energy®

Address: 3980 Sherman St., Suite 170 Zip Code: 92110 City: San Diego State: California

Telephone (xxx) xxx-xxxx: (858) 244-1186 Facsimile (xxx) xxx-xxxx: (858) 244-1178 Email: sephra.ninow@energycenter.org

Name:

Title:

Utility Name:

Address:

City: State: California

Telephone (xxx) xxx-xxxx: Facsimile (xxx) xxx-xxxx:

Email:

ENERGY Advice Letter Keywords

Affiliate	Direct Access	Preliminary Statement	
Agreements	Disconnect Service	Procurement	
Agriculture	ECAC / Energy Cost Adjustment	Qualifying Facility	
Avoided Cost	EOR / Enhanced Oil Recovery	Rebates	
Balancing Account	Energy Charge	Refunds	
Baseline	Energy Efficiency	Reliability	
Bilingual	Establish Service	Re-MAT/Bio-MAT	
Billings	Expand Service Area	Revenue Allocation	
Bioenergy	Forms	Rule 21	
Brokerage Fees	Franchise Fee / User Tax	Rules	
CARE	G.O. 131-D	Section 851	
CPUC Reimbursement Fee	GRC / General Rate Case	Self Generation	
Capacity	Hazardous Waste	Service Area Map	
Cogeneration	Increase Rates	Service Outage	
Compliance	Interruptible Service	Solar	
Conditions of Service	Interutility Transportation	Standby Service	
Connection	LIEE / Low-Income Energy Efficiency	Storage	
Conservation	LIRA / Low-Income Ratepayer Assistance	Street Lights	
Consolidate Tariffs	Late Payment Charge	Surcharges	
Contracts	Line Extensions	Tariffs	
Core	Memorandum Account	Taxes	
Credit	Metered Energy Efficiency	Text Changes	
Curtailable Service	Metering	Transformer	
Customer Charge	Mobile Home Parks	Transition Cost	
Customer Owned Generation	Name Change	Transmission Lines	
Decrease Rates	Non-Core	Transportation Electrification	
Demand Charge	Non-firm Service Contracts	Transportation Rates	
Demand Side Fund	Nuclear	Undergrounding	
Demand Side Management	Oil Pipelines	Voltage Discount	
Demand Side Response	PBR / Performance Based Ratemaking	Wind Power	
Deposits	Portfolio	Withdrawal of Service	
Depreciation	Power Lines		

What's New Self-Generation Incentive Program (SGIP)

The 2020 V6V5 Handbook has been updated to reflect the following changes:

- Updated incentive collections and allocations
 - Statewide Program Budget and Administrator Allocations (§1.1)
- Updated administrative allocations statewide
 - Budget Allocation (§1.2)
- Added Fund Shifting Authority
 - Budget Allocation (§1.2)
- Updated waitlist procedures
 - Wait List and Program Closure (§2.3.6)
- Updated rates and budget structure for renewable generation projects
 - Generation Incentive Rates (§3.1.1)
- Updated missing information request parameters
 - Incomplete Reservation Request (§2.3.4)
 - o Incomplete Proof of Project Milestone (§2.4.2)
 - o Incomplete Incentive Claim (§2.5.2)
- Created new section on resiliency adder for generation technologies
 - o Renewable Generation Resiliency Adder (§3.1.1.1)
- Updated ITC requirements and effective date as well as added new steps for small residential budget category
 - Energy Storage Incentive Rates (§3.1.2)
 - Required Documentation for Incentive Claim (§5.4.3)
- Created section to discuss resiliency adder for general market
 - General Market Non-Residential Storage Resiliency Adder (§3.1.2.1)
- Created new section on the new residential storage "soft target"
 - Residential Storage "Soft Target" (§3.1.2.2)
- Added duration step-down incentive structure expanded to general market storage projects
 - Incentive Declines Based on Storage Duration (§5.3.3)
 - Deleted Incentive Declines Based on Storage Incentive (§5.3.2)
- Revised requirements for equipment approvals in the program
 - Required Documentation for Reservation Request (§5.4.1)

- Added Islanding RRF requirements for New Residential Projects
 - Required Documentation for Reservation Request (§5.4.1)
- Revised requirements for equipment approvals in the program
 - Required Documentation for Reservation Request (§5.4.1)
- Clarified classification for VNEM projects
 - Required Documentation for Reservation Request (§5.4.1)
 - Required Documentation for Incentive Claim (§5.4.3)
- Removed low-income verification document requirement for single-family equity projects
 - Required Documentation for Reservation Request (§5.4.1)
- Added customer attestation for Generation systems
 - o Required Documentation for Reservation Request (§6.10.1)
 - Required Documentation for Incentive Claim (§6.10.3)
- Added Islanding ICF requirements for Generation systems
 - Required Documentation for Incentive Claim (§6.10.3)
- Added AB1144 requirements for renewable generation projects
 - Required Documentation for Reservation Request (§6.10.1)
- Created definition for "discrete PSPS event"
 - Definitions and Glossary
- Added D. 20-01-021 to SGIP regulatory history
 - Legislative and Regulatory History

Contents

V	What's New Self-Generation Incentive Program (SGIP)1			
Ρ	Program Administrator Contact Information1			
Р	rogram O	verview	11	
1	Budge	t	12	
	1.1 S	statewide Program Budget and Administrator Allocations	12	
	1.2 E	Sudget Allocation	12	
2	Applica	ations	15	
	2.1 A	pplication Process	15	
	2.1.1	Application Submission	15	
	2.1.2	Signatures	15	
	2.1.3	File Retention	16	
	2.2 Ir	ncentive Process Flowcharts	16	
	2.3 F	Reservation Request	18	
	2.3.1	Submitting the Reservation Request	18	
	2.3.2	Lottery Process	18	
	2.3.3	Pause Period	19	
	2.3.4	Incomplete Reservation Request	20	
	2.3.5	Approval of Reservation Request	20	
	2.3.6	Wait List and Program Closure	21	
	2.4 F	Proof of Project Milestone	21	
	2.4.1	Submitting Proof of Project Milestone	21	
	2.4.2	Incomplete Proof of Project Milestone	21	
	2.4.3	Approval of Proof of Project Milestone	22	
	2.5 Ir	ncentive Claim	22	
	2.5.1	Submitting Incentive Claim	22	
	2.5.2	Incomplete Incentive Claim	22	
	2.5.3	Inspections	23	
	2.5.4	Approval of Incentive Claim	24	

	2.6 N	Modifications and Extensions	24
	2.6.1	Modifications Pre-ICF	24
	2.6.2	Modifications Post-ICF	25
	2.6.3	Extensions and Exceptions	25
3	Incentive	s	26
	3.1 In	centive Rates	26
	3.1.1 F	Renewable Generation Incentive Rates	26
	3.1.1.1	1 Renewable Generation Resiliency Adder	27
	3.1.2 E	Energy Storage Incentive Rates	27
	3.1.2.	1 General Market Non-Residential Storage Resiliency Adder	28
	3.1.2.2	2 Residential Storage "Soft Target"	28
	3.1.2.3	3.4 Residential Equity and Equity Resiliency Incentive Levels	28
	3.1.3 I	ncentives for Technologies from a California Manufacturer	29
	3.2 Ince	entive Limitations	31
	3.2.1 N	Maximum Incentive Amount	31
	3.2.2	Total Eligible Project Costs	31
	3.2.3 I	ncentive Calculation for Site with Multiple Systems	32
	3.2.4	Calculating Incentives with Existing Systems	33
	3.2.5	Calculating Incentives for Replacement Generation	33
	3.2.6	Incentives from other sources	33
	3.2.7 [Developer Cap	34
	3.3 PBI /	Assignment	35
4	Progra	am Eligibility	36
	4.1 F	Program Participant Criteria	36
	4.1.1	Host Customer	36
	4.1.2	System Owner	39
	4.1.3	Applicant	39
	4.1.4	Payee	39
	415	Developer	30

4.2 E	Equipment Eligibility	43
4.2.1	Commercial Availability	43
4.2.2	Interconnection	43
4.2.3	Permanent Installation	44
4.2.4	Ineligible Equipment	45
4.2.5	System Sizing for Projects without Peak Demand Information	45
4.2.6	Eligibility for New Technologies	45
4.2.7	Program Modification Guidelines	46
5 Energ	y Storage Technologies	48
5.1 F	Rating Criteria for Energy Storage Projects	48
5.1.1	Rated Capacity (W)	48
5.1.2	Energy Capacity (Wh)	49
5.2 E	Eligibility Requirements for Energy Storage Projects	49
5.2.1	Greenhouse Gas Signal	49
5.2.2	Greenhouse Gas Emission Standards for new non-residential applications	50
5.2.3	Greenhouse Gas Emission Standards for non-residential legacy applications	51
5.2.4	System Size Parameters	51
5.2.5	Operational Requirements	52
5.2.6	Paired with On-site Renewables	52
5.2.7	Demand Response Dual Participation	52
5.2.8	Upfront Eligibility Requirements for New Residential Projects	53
5.2.9	Greenhouse Gas Emission Standards for Legacy Residential Projects	53
5.3 I	ncentive Calculation for Energy Storage Projects	53
5.3.1	Incentive Declines Based on GHG Performance	53
5.3.2	Incentive Declines Based on Storage Duration	54
5.3.3	Equity Budget Incentive Declines Based on Storage Duration	54
5.3.4	Incentive Declines and Caps Based on Energy Capacity (Wh)	56
5.3.5	Performance-Based Incentive Payment (PBI)	56
5.4 A	Application Documentation Requirements for Energy Storage Projects	57

	5.4.1	Required Documentation for Reservation Request	57
	5.4.2	Required Documentation for Proof of Project Milestone	64
	5.4.3	Required Documentation for Incentive Claim	66
	5.5 N	Metering & Monitoring Requirements for Energy Storage Projects	68
	5.5.1	Minimum Electrical Meter Requirements	69
	5.6 S	San Joaquin Valley Affordable Energy Pilot Projects, Budget, Incentive Design, and Eligibili	ty 71
3	Genera	ation Technologies	73
	6.1 C	Operational Eligibility Requirements for Projects Operating on Blended Fuel	73
	6.1.1	Minimum Operating Efficiency Requirements	73
	6.1.2	NOx Emission & Minimum System Efficiency Standards	74
	6.1.3	Greenhouse Gas Emission Standards	75
	6.1.4	Reliability Criteria	76
	6.1.5	Rating Criteria for System Output	77
	6.2 C	Capacity Factors	77
		Operational Eligibility Requirements for Renewable Technologies and Generation Pro	-
	6.3.1	Rating Criteria for System Output of Renewable Technologies	78
	6.4 S	Sizing Requirements for all Generation Systems	78
	6.4.1	System Sizing for Wind Turbines	79
	6.4.2	System Sizing for PRT, Waste Heat to Power, CHP and Fuel Cells	79
	6.4.3	System Sizing for Projects Exporting Power to the Grid	79
	6.4.4	System Sizing for RES-BCT Customers	79
	6.4.5	System Sizing Limitations - Ineligible Host Customer Loads	79
	6.5 E	ligible Fuel Requirements	80
	6.5.1	Renewable Fuel Blending Requirements	80
	6.5.2	Directed Biogas Project Requirements	80
	6.5.3	Directed Biogas Renewable Fuel Audits	81
	6.5.4	Renewable Fuel Commitment Modifications	81
	655	Pressure Reduction Turbine Requirements	81

	6.6 Ir	ncentive Calculation for Generation Projects	82
	6.6.1	Incentive Declines Based on Generation Capacity	82
	6.7 F	Performance-Based Incentive Payment (PBI)	82
	6.7.1	PBI Payments for Export to the Grid Projects	83
	6.8 F	Renewable Fuel Annual Payment Requirements	84
	6.8.1	Directed Renewable Fuel Verification	84
	6.8.2	On-site Renewable Fuel Verification	85
	6.9 Ir	ncentive Limitations for Projects using Renewable Fuel	85
	6.9.1	Limitations on PBI based on GHG Emissions Reductions	85
	6.9.2	Limitations on PBI Adjustments based on Renewable Fuel Verification	86
	6.9.3	Incentive Limit for the Renewable Fuel Adder	86
	6.9.4	Blended Fuel Generating Systems Converted to 100% Renewable Fuel	86
	6.10 A	application Documentation Requirements for Generation Projects	87
	6.10.1	Required Documentation for Reservation Request	87
	6.10.2	Required Documentation for Proof of Project Milestone	95
	6.10.3	Required Documentation for Incentive Claim	99
	6.11 N	Netering & Monitoring Requirements for Generation Projects	101
7	Meteri	ng & Data Collection	106
	7.1	Oata Reporting and Transfer Rules – Contract for PDP Services	106
	7.1.1	Data Format	107
	7.1.2	Meter Reading and Data Submission Timeline	107
	7.1.3	Online Submission Process	108
	7.1.4	PDP Data Validation	108
	7.1.5	Data Audits & Payment Validation	108
	7.1.6	PDP Performance Exemptions	109
	7.1.7	PDP Non-Performance	109
	7.1.8	Data Retention	110
	7.1.9	Technical and Customer Support.	110
	7.1.10	Program Administrator Liability	110

	7.2	PDP Application Process	.111
	7.2.1	Data Transfer Test	.111
	7.3	Data Privacy and Security	.111
	7.4	Measurement & Evaluation (M&E) Activities	.112
	7.4.1	M&E Field Visits	.112
	7.4.2	M&E Metering Requirements	.112
	7.4.3	Disposition of SGIP Metering Equipment	.113
8	Disp	ute Resolution	.114
9	Parti	cipant Performance and Infractions	115
	9.1	Participant Performance	115
	9.1.1	Application	115
	9.1.2	Inspection	115
	9.1.3	Attrition and Extensions	115
	9.1.4	Data Reporting	116
	9.1.5	SGIP Online Application Database Operation	.116
	9.1.6	Developer	116
	9.1.7	Residential GHG Emissions Compliance	.116
	9.2	Infractions	.117
	9.2.1	Issuance of Warnings and Infractions	.117
D	efinition	s and Glossary	.118
Le	egislatio	n and Regulatory Background	125
Α	ppendix	A - System Calculation Example	133
	Efficier	cy Calculations	133
Α	ppendix	B – Combustion Emission Credit Calculation	. 134
Α	ppendix	C - Conversion of Emissions PPM to Lb/MWH	. 136
Α	ppendix	D – Conversion Tables for HVAC-Integrated S-TES	. 138
Α	ppendix	E – Updates to the GHG Emissions Factor Section 379.6(b)(2) as Amended by Senate Bill	861
(L	egacy I	Projects Only)	.140
9	CID CH	G Emissions Fligibility Factor – The Equation	140

Appendix F – Incentive rate for large-scale storage projects claiming the ITC at all incentive levels 142

Program Administrator Contact Information

Potential program participants can obtain information and apply for incentive funding through the following Program Administrators:1

Pacific Gas & Electric (PG&E)

Website: www.pge.com/sgip Email Address: selfgen@pge.com Telephone: 1 (877) 743-4112

Mailing Address: Self-Generation Incentive Program

PO Box 7433

San Francisco, CA 94120

Overnight Mailing Address: 245 Market Street

Mail Code N9K

San Francisco, CA 94105-1797

Center for Sustainable Energy® (CSE)

Website: www.energycenter.org/sgip Email Address: sgip@energycenter.org Telephone: (858) 244-1177

Mailing Address: Center for Sustainable Energy

Attn: Self-Generation Incentive Program

3980 Sherman Street, Suite 170

San Diego, CA 92110

Southern California Edison (SCE)

Website: www.sce.com/SGIP Email Address: SGIPgroup@sce.com Telephone: (626) 302-0610

Mailing Address: Self-Generation Incentive Program

Southern California Edison

P.O. Box 800

Rosemead, CA 91770-0800

Southern California Gas Company (SoCalGas)

Website: https://www.socalgas.com/for-your-business/power-

generation/self-generation-incentive Email Address: selfgeneration@socalgas.com Mailing Address: Self-Generation Incentive Program

Southern California Gas Company 555 West Fifth Street, GT20B8 Los Angeles, CA 90013-1011

¹ Potential eligible projects located in the service territory of both Southern California Edison and the Southern California Gas Company can apply for incentive funding to either Program Administrator, but not to both.

Program Overview

The Self-Generation Incentive Program (SGIP) provides financial incentives for the installation of new qualifying technologies that are installed to meet all or a portion of the electric energy needs of a facility. The purpose of the SGIP is to contribute to Greenhouse Gas (GHG) emission reductions, demand reductions and reduced customer electricity purchases, resulting in the electric system reliability through improved transmission and distribution system utilization; as well as market transformation for distributed energy resource (DER) technologies.

This handbook establishes the policies and procedures of the SGIP for potential program participants and other interested parties. The SGIP has been approved by the California Public Utilities Commission (CPUC) and is subject to change in whole or in part at any time without prior notice. Any changes made to the SGIP will be published in revisions to this Handbook and/or posted at each Program Administrator's (PA's) website. The Program Administrators are Pacific Gas and Electric (PG&E), Southern California Edison (SCE), the Southern California Gas Company (SoCal Gas) and the Center for Sustainable Energy® (CSE).

⁻

² CSE is the Program Administrator for SDG&E customers.

1 Budget

1.1 Statewide Program Budget and Administrator Allocations

Authorized incentive collections through the end of 2019 2024—total \$813,400,000 \$501,735,000.3 Additional funds made available through attrition will be added to Program Administrators' budgets as they become available. Authorized incentive collections for each Program Administrator are as follows:

 Pacific Gas and Electric Company
 \$217,620,000
 \$360,000,000⁴

 Southern California Edison Company
 \$169,260,000
 \$280,000,000⁵

 Center for Sustainable Energy
 \$66,495,000
 \$99,000,000

 Southern California Gas Company
 \$48,360,000
 \$74,400,000

The SGIP shall be administered on a continuous basis. Program Administrators will issue incentive reservations until all incentive funds have been fully allocated.⁶

The current budget, incentive rates, and incentive steps in each Program Administrator territory are posted at www.selfgenca.com.

1.2 **Budget Allocation**

The budget is divided accordingly and described in detail below:

Energy Storage Technologies – 88% 80% of funds

- Beginning in Step 3, 75% of energy storage funds will remain in the general budget, including the largescale and small residential carve outs.
- Beginning in Step 3, 25% of energy storage funds are directed to the Equity Budget, including the nonresidential and residential carve outs.
- Large-scale storage (greater than 10 kW) 10%
- Small residential storage (equal to or less than 10 kW) 7%
- Residential equity 3%

³ Available authorized incentives include comprise of ratepayer collections from 2020 to 2024 pursuant to D.20-01-021 50% of 2016 collections (D.15-12-027) plus authorized incentive collections for 2017, 2018 and 2019 (D.14-12-033 and D.17-04-017).

⁴ Please note, PG&E's funding allocation per D.20-01-021 does not include funding for administrative expenses.

⁵ Please note, SCE's funding allocation per D.20-01-021 does not include funding for administrative expenses.

⁶ Total available funding includes authorized incentive collections, funds from cancelled projects, and application fee forfeitures.

- Non-residential equity 0%⁷
- Equity resiliency (residential and non-residential) 63%
- Heat pump water heaters (general) 5%
- SJV Pilot budget 0%⁸

Generation Technologies – 12% 20% of funds

Energy Storage General Budget

The incentive budget allocates 88% 80% to energy storage technologies, with 7% 13% of the energy storage category carved out for small residential projects less than or equal to 10 kW. This budget category has added Steps 6 & 7 (with equal budget allocations of \$28 million each per step) and continues the \$0.05/Wh incentive step-down structure. The small residential storage carve out is set per each Program Administrator step and operates independently of the large-scale carve out. Once the funds in the residential carve-out fifth step in an individual PA's territory are exhausted, PAs may use funds from the large-scale storage category, if available, to fund additional small residential projects as they are submitted. Furthermore, the PAs may file a Tier 2 advice letter to change the allocation to residential projects less than or equal to 10kW before the opening of Steps 2, 3 or 4 if the PAs find a modification is justified by the ratio of demand to budget for such projects exhibited in an earlier step relative to the ratio of demand to budget for projects greater than 10kW in size.

The incentive budget allocates 10% to large-scale storage (greater than 10 kW) for Steps 3 through 5.

Energy Storage Equity Budget

The incentive budget allocates 3% for residential energy storage equity budget for qualifying residential (single family and multi-family low-income housing) regardless project size. The incentive budget allocates 0% for non-residential energy storage equity budget until existing carryover budget is exhausted.

For energy storage incentive steps 3-5, each PA's energy storage general budget (large-scale and small residential) is reduced by 25% in Steps 3-5 to create an energy storage Equity Budget for eligible projects. Additionally, within the Equity Budget, 10% is carved out for qualifying residential projects (single family and multi-family low-income housing) regardless of the size of the project. The residential equity carve-out is set per each Program Administrator step and operates independently of the Equity Budget for non-residential customers.

⁷ 2020-2024 authorized collections suspends further collections for non-residential equity storage once existing carryover is exhausted.

⁸ Pursuant to D.19-09-027, SJV Pilot Program has a \$10 million set-aside funded from SCE and PG&E's unused non-residential equity budget.

Renewable Generation Budget

The incentive budget allocates 12% 20% to generation technologies, with a minimum of 40% of the generation category carved out for renewable generation projects. With no incentive step-down structure. The minimum amount of incentives set aside for renewable generation technologies is set statewide, across all four Program Administrators.

Equity Resiliency Budget

A \$100 million Equity Resiliency Budget has been established using unspent funds from the Generation Technology budget.¹¹

Heaty Pump Water Heater Budget

\$4 million has been established for equity projects. Additionally, the incentive budget allocates 5% to general market HPWHs.

SJV Pilot Budget

\$100 million has been set aside from SCE and PG&E's unspent non-residential equity budget.

Fund Shifting Authority

Program Administrators have the option to transfer funds between technology incentive budgets after December 31, 2022 if it is likely that funds will remain unspent, allowing PAs the flexibility to better respond to market demands in SGIP.¹²

⁹ The sum of the renewable generation budget per D.17-04-017 combined with 40% of authorized collections in the generation budget per D.16-06-055.

⁴⁹ Defined as wind, waste heat to power, pressure reduction turbines, and 100% biogas as defined by the most recent RPS guidelines.

¹¹ Funds were transferred sequentially in order from the highest budgeted incentive step to the lowest. If any PA had insufficient funds in the Generation Technology budget to reach the budget levels required, the shortfall was transferred from the Large-Scale storage budget.

¹² Implemented pursuant to Ordering Paragraph 34 in D.20-01-021.

2 Applications

2.1 Application Process

Applications are subject to the incentive rates of the Program Administrator to which they apply. Generally, applications will be assigned an incentive rate and reviewed in the order in which they are received. However, in the event that application submissions on a single day exceed available funding in a given Program Administrator's territory for a given budget and step, a lottery will be initiated. Lottery details are found in *Section 2.3.2*. Please refer to www.selfgenca.com for the most up-to-date information on current incentive steps, rates, and available funds.

2.1.1 Application Submission

All SGIP applications and required documents¹³ at all stages of the application process must be submitted via the SGIP online application database at www.selfgenca.com. Mailed, emailed, faxed or hand delivered applications will not be accepted.

In order to submit an application and/or project documentation, companies or individuals must create an account and register users at www.selfgenca.com. Once the account has been confirmed, registered Applicants may create and edit applications.

Only complete applications may be assigned incentive funds or be placed into a lottery. Only complete applications may receive an approved reservation. Duplicate applications or multiple submissions for the same project will be rejected. ¹⁴ Applicants must agree to the Terms of Use pertaining to the SGIP online application database in order to submit an application. The Terms of Use can be found at www.selfgenca.com.

Once an application is entered into the SGIP database and submitted for consideration within a given step, it will be retained in the database. In the event a lottery is implemented and the application is *not* selected for the current step, the Applicant must update relevant documentation and resubmit the application in order to be considered in the next incentive step.

2.1.2 Signatures

Original signed documents or scanned copies of original signed documents are required for all program provided forms. ¹⁵ Electronic signatures and/or acknowledgments are acceptable for documents created by

¹³ With the exception of the application fee check that is to be mailed directly to the Program Administrator. See *Section 5.4.1* and *Section 6.10.1* for more information.

¹⁴ Duplicative applications are considered a program infraction. See Section 9 for information on program infractions.

¹⁵ Includes: Reservation Request Forms, Proof of Project Milestone Forms, Incentive Claim Forms, and all affidavits. All forms requiring signatures from multiple parties must have all signatures submitted on one document.

the Contractor or Host Customer, such as the installation contract. ¹⁶ The signature on any signed document submitted to the Program Administrator shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such signature page were an original thereof.

2.1.3 File Retention

Although "wet" signatures are not required on submitted documents, original signed documentation must be retained by the Applicant, Host Customer and/or System Owner for at least five years from the date of submission. Program Administrators reserve the right to request original signed documents within the five-year period.

2.2 Incentive Process Flowcharts

There are two application processes illustrated below:

- Three-Step Application Process Figure 2.2-1
- Two-Step Application Process Figure 2.2-2

All residential projects and non-residential projects less than 10 kW must follow the two-step application process. Non-residential projects greater than or equal to 10 kW must follow a three-step process.

¹⁶ E-signatures may be acceptable for other program documents depending on Program Administrator territory. Please contact your Program Administrator for more information.

Figure 2.2-1: Three-Step Application Process for Public and Non-Public Entities ≥ 10 kW

Reservation Request

- •Completed Reservation Request Form submitted to the Program Administrator along with the required attachments. (See Section 5.4.1 and 6.10.1)
- •Once a project is assigned funding in a given step (first-come or lottery), Program Administrator reviews Reservation Request documents. Once the documentation has been approved and eligibility requirements have been met, a Conditional Reservation Letter is issued.
- Proof of Project
 Milestone (PPM)
 documents are due
 within 90 calendar days
 of the Conditional
 Reservation date for
 Non-Public entities, and
 within 240 calendar days
 for Public Entities (copy
 of RFP due within 90
 days). (See Section 5.4.2
 and 6.10.2)

Proof of Project Milestone

- •Completed Proof of Project Milestone documents submitted to the Program Administrator by the PPM due date.
- Program Administrator reviews PPM documents.
 Once the documentation has been approved a Confirmed Reservation Letter is issued.
- •Incentive Claim Form (ICF) documents are due within 18 months of the Conditional Reservation date (see Section 5.4.3 and 6.10.3).

Incentive Claim

- Completed Incentive Claim Form documents submitted to the Program Administrator once the system is installed, interconnected and operational.
- Program Administrator reviews ICF documents and may schedule a site inspection to verify system eligibility.
- •Upon approval, the incentive payment process begins. Upfront incentive will be issued, and (where applicable) the Performance Based Incentive process begins.

Figure 2.2-2: Two-Step Application Process for All Residential and Non-Residential Entities <10 kW

Incentive Claim **Reservation Request** • Completed Reservation Request Form submitted Completed Incentive Claim Form documents to the Program Administrator along with the submitted to the Program Administrator once required attachments. (See Section 5.4.1 and the project is completed. 6.10.1) Program Administrator reviews ICF documents • Once a project is assigned funding in a given and may schedule a site inspection to verify step (first-come or lottery), Program system eligibility. Administrator reviews Reservation Request documents. Once the documentation has been • Upon approval, the incentive payment process approved and eligibility requirements have been begins. Upfront incentive will be issued, and met, a Confirmed Reservation Letter is issued. (where applicable) the Performance Based Incentive process begins. • Incentive Claim Form (ICF) documents are due within 12 months of the Confirmed Reservation date (See Section 5.4.2 and 6.10.2)

2.3 **Reservation Request**

To request an incentive, an online Reservation Request Form (RRF) must be submitted with required attachments. Non-residential projects are required to submit an application fee upon submittal of the RRF. (See *Section 5.4.1* and *6.10.1* for required attachments specific to storage and generation technologies.) Projects that include multiple technologies must submit separate applications for each technology.

2.3.1 Submitting the Reservation Request

After all Reservation Request documents have been uploaded to the SGIP online application database, Applicants must submit their complete application package to the appropriate Program Administrator. Applicants may not submit applications in excess of the Developer cap for the active step. Once an application is assigned an incentive rate, the Program Administrator will review the application package to ensure the project meets all program requirements. Incentive funds may only be reserved after the Program Administrator receives and approves the Reservation Request documents.

2.3.2 Lottery Process

A lottery will be triggered only in the event that applications submitted on a single day exceed funds available for a given budget and step. Lotteries are to be conducted separately for large scale energy storage technologies, small residential energy storage technologies less than or equal to 10 kW, and generation technologies by Program Administrator territory, as necessary. All applications not selected in the lottery will be rejected and must reapply in the next funding step in order to receive funding.

2.3.2.1 Priority Projects

The following energy storage projects shall have priority in the SGIP lottery process:

- Energy storage projects located within the service territory of Los Angeles Department of Water and Power.¹⁷
- Energy storage projects located within the West Los Angeles Local Reliability Area of Southern California Edison's service territory.
- Energy storage systems paired with an on-site renewable generator and claiming the Investment Tax Credit (ITC) or, if not claiming the ITC, charging a minimum of 75% from the on-site renewable generator.

Energy storage projects that meet more than one criterion shall be given the highest priority. A lottery will be held for the projects in the priority or non-priority category that exceeds available funding in the active step.

Generation projects shall have priority in the SGIP lottery in the following order:

- 1) Renewable projects using wind, waste heat to power, pressure reduction turbines, or 100% on-site biogas will be given first priority.
- 2) 100% directed biogas will be given second priority.
- 3) Blended on-site biogas will be given third priority.
- 4) Blended directed biogas will be given fourth priority.

A lottery will be held for the priority category that exceeds available funding in the active step.

2.3.3 Pause Period

When a budget category changes to the next incentive step the Program Administrator will initiate a pause period of no less than 20 days, whereby:

- No new applications within the budget category are accepted.
- The Program Administrator may perform a pre-screen to reject applications with missing documentation or applications submitted above the Developer cap, and to verify projects identified with a locational priority.
- If required, the lottery is conducted.
- After 10 days, Program Administrators will determine if the incentive level reduction for energy storage technologies shall increase from \$0.05/Wh to \$0.10/Wh between incentive steps based on statewide oversubscription for a given step.

¹⁷ All projects interconnecting into LADWP's electrical grid must abide by LADWP interconnection rules.

- If a lottery is conducted, a notification of the results of the lottery is sent to Applicants. Applications that were not selected for funding in the current step through the lottery will be instructed on how to reapply for funding in the next step.
- Projects that are only able to be partially funded within a certain step must choose to reapply for funding in the next step or claim the remaining funds in the current step. 18
- The SGIP public website is updated with information on the new incentive rate(s), available funds and the date of the next application submission opportunity.

2.3.4 Incomplete Reservation Request

Reservation Request documents must be complete and submitted as part of the application package. If an application is found to be missing any required information or requires additional clarification, the Program Administrator or their representative will request the information necessary to process that application further. Applicants have 15 calendar days to respond with the necessary information. If after 15 calendar days the Applicant has not submitted the requested information or satisfied the eligibility requirement in question, the application may be cancelled. An application may not receive multiple requests for clarification or more information for the same missing or uncorrected item. Cancelled applications may be resubmitted and will be treated as a new application (i.e. all required documents must be resubmitted) and processed in sequence along with other new applications. Funds from cancelled projects will be allocated to the currently active incentive step in the Program Administrator's SGIP incentive budget. If the Program Administrator is in a pause period when attrition occurs, the funds will be placed in the next incentive step.

2.3.5 Approval of Reservation Request

Upon Program Administrator approval of the Reservation Request package (Reservation Request Form and required attachments) the Applicant and Host Customer will receive a reservation letter. The two types of reservation notice letters are based on the type of application:

Conditional Reservation Letter (for Three-Step applications)

Upon approval of the Three-Step Reservation Request package, a Conditional Reservation Letter will be issued confirming that a specific incentive amount is conditionally reserved for project. The letter will list the approved incentive amount, the Proof of Project Milestone date and the reservation expiration date. All reservations are conditional pending receipt of the Proof of Project Milestone documentation on or before the Proof of Project Milestone date.

Confirmed Reservation Letter (for Two-Step applications)

Upon approval of the Two-Step Reservation Request package, a Confirmed Reservation Letter will be issued. The Confirmed Reservation Letter will list the approved incentive amount and the reservation expiration date (12 months after the date of the Confirmed Reservation Letter). Upon project completion

¹⁸ Projects are not allowed to be assigned a "split incentive" across two or more incentive steps.

and prior to the reservation expiration date, the completed Incentive Claim Form must be submitted along with all of the necessary documentation to request an incentive payment.

2.3.6 Wait List and Program Closure

Once funds have been fully allocated in the final incentive step of a Program Administrator's given budget, applications will be placed on a wait list to be funded as incentive funds become available throughout the remainder of the program. When there is enough attrition to fund wait-listed projects, wait listed projects will be assigned an incentive rate in the last step and reviewed in the order in which they were submitted. In the event that there are available funds and all wait-listed projects have been allocated funding, new applications will be subject to normal program procedures specified in *Section 2.1*. Program Administrators may continue accepting new applications until all incentive funds have been fully paid or until December 31, 2024 2020 19, whichever comes first. 20

2.4 **Proof of Project Milestone**

<u>Two-Step Applications</u> must submit all Proof of Project Milestone (PPM) documents as part of their Reservation Request. For <u>Three-Step Applications</u>, <u>Non-Public Entities</u> have 90 calendar days from the date of the Conditional Reservation Letter to satisfy all Proof of Project Milestone criteria. For <u>Three-Step Applications</u>, <u>Public Entities</u> must submit a copy of the issued Request for Proposal (RFP) or equivalent for purchase or installation of the system within 90 calendar days of the date of the Conditional Reservation Letter; Proof of Project Milestone documentation must then be submitted within 240 days of the date of the Conditional Reservation Letter.

2.4.1 Submitting Proof of Project Milestone

Once the Online Proof of Project Milestone is completed and all the required attachments are uploaded, the PPM package must be submitted to the appropriate Program Administrator via the online application database.

2.4.2 Incomplete Proof of Project Milestone

If the Proof of Project Milestone package is not received by the Proof of Project Milestone date, the application may be cancelled by the Program Administrator.

If the Proof of Project Milestone documentation is incomplete and/or requires clarification, the Program Administrator will request the information necessary to process that application further. Applicants will have 15 calendar days to respond with the necessary information. If after 15 calendar days the requested

¹⁹ SB 700 861 extended SGIP administration through 2024 2020

²⁰ Applications that remain on the waitlist for step 5 when step 6 opens will be assigned to step 6 until the soft target is reached. Projects that remain on the waitlist once the target is met will be removed and the waitlist closed. Projects are welcome to apply to the next step provided they meet the current eligibility requirements.

information has not been submitted or satisfied the eligibility requirement in question, the application may be cancelled. An application may not receive multiple requests for elarification or more information for the same missing or uncorrected item, the application may be cancelled. Any project attrition and forfeited application fees will be allocated to the current incentive step in the Program Administrator's SGIP incentive budget. If the Program Administrator is in a pause period when attrition occurs, the funds will be placed in the next incentive step.

2.4.3 Approval of Proof of Project Milestone

Once Proof of Project Milestone requirements have been successfully met, the Program Administrator will issue a Confirmed Reservation Letter. The Confirmed Reservation Letter will list the reservation dollar amount and the reservation expiration date (18 months after the date of the original Conditional Reservation Letter). Upon project completion and no later than the reservation expiration date, the completed Incentive Claim Form must be submitted along with all of the necessary documentation to request an incentive payment.

2.5 Incentive Claim

Once the project is complete, Applicants must request payment of the incentive amount by submitting the Incentive Claim Form (ICF) and all applicable Incentive Claim documents to the Program Administrator via the online application database. A project is considered complete when the system is completely installed, interconnected (if applicable), permitted, and capable of operating in the manner for which it was designed. Payment will be dispersed after the Program Administrator verifies by Post-Installation inspection²¹ (if the project is selected for a Post-Installation inspection) that the system meets all the eligibility requirements of the SGIP. The completed Incentive Claim Form must be submitted to the Program Administrator on or before the reservation expiration date together with the required attachments.

2.5.1 **Submitting Incentive Claim**

Once the Incentive Claim Form is complete and all the required attachments are submitted, Applicants may submit their Incentive Claim package to the Program Administrator via the online application database.

2.5.2 Incomplete Incentive Claim

If the complete Incentive Claim package is not received by the reservation expiration date, the application may be cancelled by the Program Administrator.

If submitted Incentive Claim documentation is incomplete and/or requires clarification, the Program Administrator will request the information necessary to process that application further. Applicants have 30 calendar days to respond with the necessary information. If after 30 calendar days the requested information has not been submitted or satisfied the eligibility requirement in question, the application may be cancelled. An application may not receive multiple requests for clarification or more information for the

-

²¹ See Definition for *Inspections*.

same missing or uncorrected item, the application may be cancelled. Any project attrition and forfeited application fees will be allocated to the current incentive step in the Program Administrator's SGIP incentive budget. If the Program Administrator is in a pause period when attrition occurs, the funds will be placed in the next incentive step.

2.5.3 Inspections

Upon receipt of a complete Incentive Claim Form package, the Program Administrator may organize an On-site Post-Installation inspection (or virtual Post-Installation inspection for select residential applications) ²² to verify that the project's system is installed as represented in the application, is operational, is interconnected and conforms to the eligibility criteria of the SGIP. Verification includes but is not limited to:

- If the project is subject to PBI payments, the metering system will be inspected, and it will be verified that it follows the proposed monitoring plan and meets the metering requirements of the SGIP.
- If the project uses renewable fuel, the availability and flow rate of the renewable fuel will be demonstrated by the Host Customer and/or System Owner.
- If the project uses waste energy, the availability, temperature and production rate of the waste energy will be demonstrated by Host Customer and/or System Owner.
- Energy storage systems will be tested to validate the discharge energy capacity. Energy storage projects will be inspected according to the Energy Storage Field Verification Protocol. HVAC-integrated S-TES systems will be tested to show they can provide enough thermal energy to turn off the compressor of the accompanying HVAC unit for the specified discharge duration period. Refrigeration TES systems will be tested to show they can provide enough thermal energy to turn off the compressor(s) and condenser(s) of the accompanying refrigeration system(s) for the specified discharge duration period.
- If the eligible system size depended on new construction or load growth, the required load will be confirmed.
- Verify system capacity rating to confirm the final incentive amount.

Energy storage inspections will follow the latest version of the Energy Storage Post-Installation Inspection and Discharge Testing Protocol found on www.selfgenca.com.

Failed Post-Installation Inspection

If the results of the Post-Installation inspection indicate that the project is not ready for approval, the Program Administrator will notify the Applicant, Host Customer and System Owner of the reasons for

²² Refer to Energy Storage Field Inspection and Discharge Testing Protocol and Field Inspection Sampling Protocol at www.selfgenca.com

failure. The Applicant, Host Customer and System Owner will have up to 60 calendar days to bring the project into compliance. A subsequent inspection visit may be required to determine final approval. If the Applicant, Host Customer and System Owner fail to bring the project to full eligibility within the requested time, the application may be cancelled.

If the site load, renewable fuel or waste energy forecast has not yet materialized, the Applicant will be given two options:

- 1. Receive payment based on the site load, renewable fuel or waste energy availability (whichever is less) demonstrated at the time of initial Post-Installation inspection or;
- 2. Wait for the site load, renewable fuel or waste energy to materialize within 12-months from the date the Incentive Claim Form and documents were initially received. If the site load, renewable fuel or waste energy has not materialized within the 12-month period, the project will be paid based on the site load or system operating capacity available at the end of the 12-month period, whichever is less.

2.5.4 Approval of Incentive Claim

Upon final approval of the Incentive Claim documentation and completed virtual or On-Site Post-Installation Inspection (if the project is selected for a Post-Installation inspection), the Program Administrator will issue a final ICF approval letter. The incentive payment will be made approximately 30 days after the final approval letter is sent. Payment will be made to the assigned Payee as indicated on the Incentive Claim Form and will be mailed to the address provided.

2.6 Modifications and Extensions

All projects are expected to be installed as described on the Conditional and Confirmed Reservation Letter. In the event that changes are made during the development of the project and/or during the installation it is the responsibility of the Host Customer and/or Applicant to notify the Program Administrator as soon as possible. Changes to the Host Customer or project site are generally not permitted and must be approved on a case-by-case basis by the Program Administrator. Unapproved changes may result in project cancellation.

2.6.1 Modifications Pre-ICF

Changes pertaining to System Owner, Payee, equipment type, or system capacity must be approved by the Program Administrator before the application can proceed. If the step to which a project is assigned has closed, modifications to the project will not result in additional incentive funding. At the Program Administrator's discretion, additional incentive funding for an application may be allowed only when a project is assigned to the currently active step and adequate funding is available.

Changes in equipment type, system capacity, Applicant, installer or other substantial changes may require new RRF and/or PPM documentation. Once the request has been approved, a new reservation letter may be issued. Changes do not extend the reservation expiration date.

2.6.2 Modifications Post-ICF

In general changes to the completed project are not allowed. In the event that a system needs to be upgraded or changed due to poor performance the Applicant must notify the Program Administrator of new equipment information and provide updated documentation to help support performance and measurement and evaluation activities. For projects adding generation see *Section 3.2.4*.

2.6.3 Extensions and Exceptions

Extension requests will be reviewed on a case-by-case basis and should be submitted in writing to the appropriate Program Administrator for review. Any extension granted to either the Proof of Project Milestone or Request for Proposal due date will not extend the reservation expiration date.

All projects will be limited to a maximum of three 6-month extensions of the reservation expiration date, after which the reservation expires automatically. Extensions will be for special circumstances only. Additionally, extensions will not be granted to projects that have not made satisfactory progress toward completion in compliance with established milestones and requirements. Any request for a second or third extension of the reservation expiration date requires unanimous SGIP Working Group approval, and the SGIP Working Group shall notify Applicants of the SGIP Working Group's decision in writing within 30 days. When considering a request for a second or third reservation expiration extension, the SGIP Working Group will consider:

- 1) Whether the project's delay is outside the control of the Host Customer;
- 2) Whether the project has made significant progress toward completion, and a timeline is provided showing the expected date of commissioning of the project and that interconnection of the project will fall within the third six-month extension of the project's reservation expiration date;
- 3) Whether the extension of the project's reservation expiration date will affect the Program Administrator's ability to incentivize other projects.

Eligible energy storage SGIP projects may rely on the longer of the extension granted under D.14-05-033 or the third six-month extension granted in D.15-06-002.

Any other procedure or documentation exceptions should be submitted to the appropriate Program Administrator and will be subject to Working Group approval.

²³ D.15-06-002 granted a petition for modification to increase the number of six-month extensions from two to three.

3 Incentives

3.1 Incentive Rates

The incentive rates for the two budget categories of the SGIP, generation and energy storage, are provided below.

3.1.1 Renewable Generation Incentive Rates

Renewable generation projects incentive levels are \$2.00/W with no step-down structure.

Upon program opening, total generation incentive funds are divided across three steps. Generation incentives decline by \$.10/W between incentive steps, according to the following schedule:

Table 3.1.1 Generation Incentives per Watt (W)

	Step 1		Step 2		Step 3	
Technology Type	Initial Incentive Rate	Max Incentive w/ biogas adder	Initial Incentive Rate	Max Incentive w/ biogas adder	Initial Incentive Rate	Max Incentive w/ biogas adder
Generation Technologies	\$/W	\$/W	\$/W	\$/W	\$/W	\$/W
Wind	\$0.90	n/a	\$0.80	n/a	\$0.70	n/a
Waste Heat to Power	\$0.60	n/a	\$0.50	n/a	\$0.40	n/a
Pressure Reduction Turbine ²⁴	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
Internal Combustion Engine CHP	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
Microturbine CHP	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
Gas Turbine CHP	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
Steam Turbine - CHP ²⁵	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
Fuel Cell CHP	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00

²⁴ Pressure reduction turbine includes but is not limited to, any small turbine generator installed in an existing, man-made channel for delivery of water, steam or natural gas. Decision 16-06-055 allows pressure reduction turbines to be eligible for the same renewable fuel adder which is available to other generators that directly or indirectly use fuel.

²⁵ Steam Turbines were included in the program as a conventional topping cycle CHP technology in 2014 per Energy Division's Disposition Letter approving CSE Advice Letter (AL) 47-A, PG&E AL 3474-GA/4417-E-A, SCE AL 3038-E-A, and SoCalGas AL 4644-A.

Fuel Cell Electric Only	\$0.60	\$1.20	\$0.50	\$1.10	\$0.40	\$1.00
-------------------------	-------------------	-------------------	-------------------	-------------------	-------------------	-------------------

3.1.1.1 Renewable *Generation Resiliency Adder*

A renewable generation resiliency adder incentive of \$2.50/W for renewable generation projects. Combined with the base incentive of \$2.00/W, this results in an incentive of \$4.50/W for renewable generation projects intended for resiliency purposes. Customers eligible for the equity resiliency budget and general market customers with critical resiliency needs are eligible for the renewable generation resiliency adder.²⁶

The renewable generation resiliency adder is not limited to only equity budget or equity resiliency customers and is also available to critical resiliency facilities that provide critical resiliency services to serving their communities during PSPS events.

3.1.2 Energy Storage Incentive Rates

Total energy storage incentive funds are divided across five steps for large storage and five seven steps for small residential storage. Energy storage incentives generally decline by \$0.05/Wh between incentive steps. However, if the previous incentive step becomes fully subscribed within 10 calendar days across all Program Administrator territories, the incentive decline to the next step for large-scale storage projects not claiming the ITC and the small residential carve-out will be \$0.10/Wh rather than \$0.05/Wh. The incentive for large-scale storage projects claiming the ITC will maintain a 72% differential from the incentive rate for non-ITC large-scale storage projects in all steps for equipment purchased before December 31, 2021.²⁷

The table below has been updated to reflect the accelerated incentive decline beginning in Step 2. Please visit www.selfgenca.com for a list of current incentive rates.

Step 1 Step 2 Step 3 Step 4 Step 5 Step 6 Step 7 **Energy** Storage \$/Wh \$/Wh \$/Wh \$/Wh \$/Wh \$/Wh \$/Wh General **Budget** Large Storage \$0.50 \$0.40 \$0.30 \$0.25 N/A \$0.35 N/A (>10 kW) Large Storage \$0.36 \$0.29 \$0.25 \$0.22 \$0.18 N/A N/A Claiming ITC

Table 3.1.2 Energy Storage Incentives per Watt-hour (Wh)

²⁶ As defined in D.20-01-021.

²⁷ A complete table of incentive rates for ITC and non-ITC-claiming large-scale projects is given in Appendix F.

Residential Storage (<=10 \$0.50 kW)	\$0.40	\$0.35	\$0.30	\$0.25	\$0.20	\$0.15
--	--------	--------	--------	--------	--------	--------

The PA's Equity Budget and general budget will operate independently of one another. Within a given incentive step, if the Equity Budget is exhausted, any project eligible for the Equity Budget is eligible for the general SGIP budgets (i.e., non-Equity budgets) within that same incentive step, provided there are remaining funds.

3.1.2.1 General Market Non-Residential Storage Resiliency Adder

General market non-residential customers with critical resiliency needs that provide critical facilities or infrastructure as outlined in section 4.1.1.3, with the exception of the equity requirement, qualify for the general market resiliency adder.

The non-residential general market resiliency adder is \$0.15/Wh in addition to the current step incentive rate.

3.1.2.2 Residential Storage "Soft Target"

A "soft target" is adopted such that half of the general market residential incentive budget is reserved for residential customers living in Tier 3 or Tier 2 HFTDs or residential customers whose electricity has been turned off during two or more discrete PSPS events.

The "soft target" shall be implemented by pausing acceptance of SGIP applications from residential customers who do not live in a Tier 3 or Tier 2 HFTD, or who did not have their electricity turned off in two or more discrete PSPS events prior to applying for SGIP incentives, once the PA's have received reservations requests from such customers that have reached 50 percent of that PA's available funds for each residential incentive step.

3.1.2.3 4 Residential Equity and Equity Resiliency Incentive Levels

The Equity Budget incentive for residential and non-residential customers is \$0.85/Wh. 28

For Residential and Non-Residential customers that meet the eligibility requirements of the Equity Resiliency Budget, as outlined in section 4.1.1.3, the incentive rate is \$1.00/Wh.

²⁸ The equity budget step-up incentive structure and the linkage between the equity budget and step three of the general storage incentives adopted in D.17-10-004 is eliminated.

Table 3.1.2.3 4 Energy Storage Incentives per Watt-hour (Wh)

Energy Storage Equity Budget	\$/Wh
Equity	\$0.85
Equity Resiliency	\$1.00

Vendors/developers shall not sell a residential storage system that receives incentives for a total price (before incentives) that is greater than the price they sell a comparable system that does not receive incentives.

3.1.3 Incentives for Technologies from a California Manufacturer

An additional incentive of 20 percent will be added to the technology incentive for projects in which the equipment used is manufactured in California.²⁹ In order for a project to be eligible for the 20 percent adder, it must demonstrate that at least 50% of its capital equipment value is manufactured by an approved California Manufacturer.

3.1.3.1 California Manufacturer Eligibility Criteria and Verification

All California Manufacturers will be required to submit an application for California Manufacturer status and proof to support each criterion below.

California Manufacturers must meet the following requirements:

- Operate a manufacturing facility in California
- Licensed to conduct business in California
- Registered with a primary or secondary manufacturing NAICS code

3.1.3.2 Project Equipment Verification

Equipment is deemed to be manufactured in California if at least 50% of the value of the capital equipment has been made in a dedicated production line by an approved California Manufacturer. For the purposes of determining eligibility, the SGIP recognizes the following equipment types:

Generation	Energy Storage
Generator/Prime Mover	Storage medium (i.e. battery)
Ancillary equipment	Inverter
	Controller

²⁹ The incentive for each project including the California Manufacturer adder shall be capped based upon the Incentive Limitations outlined in *Section 5.2 and Section 6.6, 6.7 and 6.9.*

3.1.3.3 How to Determine Value

Value is based on the capital cost of a single equipment type as listed above. The California Manufacturer supplying capital equipment component(s) with the largest cost percentage is the one whose California credentials will be considered. The largest cost percentage is the total value of the eligible capital equipment.

Example:

An energy storage project requests the California Manufacturer incentive adder. The project provides the following cost breakdown:

Equipment type	Company	Cost	Location Manufactured	Approved CA Manufacturer?
Advanced lithium ion batteries	ABC Company	\$12,000	111 Fake Street, Los Angeles, CA 90011	Yes
Bidirectional AC-DC inverter	Lizard Inverters	\$3,000	333 Jon Street, Phoenix, AZ 81234	No
Operating Controller	Nick Controllers	\$2,000	4444 Real Street, San Francisco, CA 92222	Yes

- Total system cost= \$17,000
- Battery cost percentage = 71%
- Inverter cost percentage = 18%
- Controller cost percentage = 11%

The capital equipment with the highest percentage cost is the battery. Since the battery was made by an SGIP approved California manufacturer, this project will be eligible to receive the 20% adder.

Beginning June 23, 2017, Program Administrators will deny requests for California Manufacturer status for manufacturers that have not met the above requirements, including suppliers which were previously approved. Also, beginning June 23, 2017, projects will receive the adder only when using equipment from an approved California Manufacturer under the above requirements. New projects that apply before June 23, 2017 with a previously approved "California Supplier" may retain the adder only if that manufacturer is re-approved under the above requirements by the Incentive Claim stage.

3.2 Incentive Limitations

Incentive amounts for both generation and storage projects can be limited by a number of factors, including maximum incentive amount, total eligible project costs, sizing limitations per site, and incentives from other sources. Please refer to *Section 5.2* and sections 6.6, *6.7* and 6.9 for incentive calculations and limitations specific to energy storage and generation technologies, respectively.

3.2.1 Maximum Incentive Amount

The maximum incentive amount per project shall not exceed \$5 million.

3.2.2 Total Eligible Project Costs

No project can receive total incentives (to include any combination of the technology incentive, biogas adder, and/or California Manufacturer) that exceed the Total Eligible Project Costs. Submittal of Project cost details is required to report Total Eligible Project Costs and to ensure incentive limits are not exceeded. Equipment and other costs outside of the project envelope are considered ineligible project costs but must also be reported.

The following costs may be included in Total Eligible Project Cost:

- 1. Engineering feasibility study costs.
- 2. Engineering and design costs.
- 3. Environmental and building permitting costs.
- 4. Equipment capital costs.
- 5. Primary heat recovery equipment, i.e. heat recovery equipment directly connected to the generation system whose sole purpose is to collect the waste heat produced by the power plant. For example, a heat exchanger or heat recovery boiler (a.k.a., heat recovery steam generator, or HRSG) used to capture heat from a gas turbine is an eligible cost.
- 6. Heat recovery piping and controls necessary to interconnect the generating equipment to either the Primary Heat Recovery Equipment or the heat recovery piping and controls within the space primarily occupied by the generator partitioned by a fence or wall, whichever cost is less. If there is no identifiable Primary Heat Recovery Equipment and no identifiable space primarily occupied by the generator, eligible heat recovery piping and control costs shall be limited to the generator skid.
- 7. Construction and installation costs. For projects in which the equipment is part of a larger project, only the construction and installation costs directly associated with the installation of the energy equipment are eligible.
- 8. Interconnection costs, including:
 - a. Electric grid interconnection application fees
 - b. Natural gas grid interconnection costs
 - c. Metering costs associated with interconnection

- Warranty and/or maintenance contract costs associated with eligible project cost equipment.
 The cost of this component is capped at 10% of the total claimed project costs.
- 10. System metering, monitoring and data acquisition equipment as well as additional on-board monitoring equipment and costs associated with the PDP contract.
- 11. Air emission control equipment capital cost.
- 12. Gas line installation costs, limited to the following:
 - a. Costs associated with installing a natural gas line on the customer's site that connects
 the serving gas meter or customer's natural gas infrastructure to the distributed
 generation unit(s).
 - b. Customer's cost for an additional (second) gas service to serve the distributed generation unit if this represents a lower cost than tying to the existing meter or gas service.
 - c. Customer's cost for any evaluation, planning, design, and engineering costs related to enhancing/replacing the existing gas service specifically required serving the distributed generation unit.
- 13. For Renewable fuel projects (except wind turbines), the cost of equipment to remove moisture and other undesirable constituents from Renewable Fuels that would damage the generation equipment. Such equipment includes but is not limited to "gas skids", dryers/moisture removal and siloxane removal towers.
- 14. Electricity storage devices
- 15. Electrical and critical loads panel and wiring upgrades are allowable costs for equity resiliency and equity budget projects.
- 16. Renewable fuel projects (except wind turbines) may claim the cost associated with securing a bond to certify use of renewable fuel, described in the SGIP Contract, as eligible costs.
- 17. Sales tax and use tax.
- 18. Cost of capital included in the system price by the vendor, contractor or subcontractor (the entity that sells the system) is eligible if paid by the System Owner.
- 19. For Steam Turbine CHP projects where new or existing boiler capacity is being increased to generate power with a steam turbine, only the incremental costs directly associated with the increased capacity is considered an eligible project cost. If the boiler or any ancillary equipment directly associated with the increased capacity received an incentive or rebate from another source, the incentive or rebate amount is an ineligible project cost and must be deducted from the eligible cost of the project.

3.2.3 Incentive Calculation for Site with Multiple Systems

Program participants can apply for incentives for multiple types of systems installed at one site. The total SGIP incentive is the sum of the incentive for each type of technology. When calculating the total eligible incentive for generation technologies, the incentives are to be calculated sequentially until the 3 MW limit is reached, with the lowest incentive rate (\$/Watt) technology portion calculated first. For multiple

generation technologies within a single incentive level, the incentives are calculated in the order in which they appear in Table 3.1.1, from top to bottom. When calculating the total eligible incentive for energy storage technologies, the incentives are to be calculated sequentially until the 6 MWh limit is reached.

3.2.4 Calculating Incentives with Existing Systems

A system may be installed in addition to existing systems if all program eligibility requirements are met by the project. Backup generators are not considered "existing on-site generation".

Sites with existing systems that have met their permanency requirements: The existing SGIP system capacity is not accounted when determining the current incentive. However, if the existing system is still in operation the existing capacity must be accounted in order to best determine new system size.

Sites with existing SGIP systems that have not met their permanency requirements: The existing SGIP system capacity is accounted first at the highest incentive rate and then the proposed system capacity incentive is added on top of the existing capacity to determine in which incentive capacity tier the proposed system falls. Energy storage system capacity is non additive with generation capacity for purposes of calculating the tiered incentive. The incentive calculation and capacity limits are treated separately for energy storage and generation technologies.

3.2.5 Calculating Incentives for Replacement Generation

Installation of a new system intended to replace an existing system is allowed if all program eligibility requirements have been met and the replaced system has either never received incentives from the SGIP or the Energy Commission's Emerging Renewables Program (ERP), or has received incentives from the SGIP, or ERP programs but has been in service for at least the applicable program's permanency requirement. Systems that did receive incentives but have not met the appropriate program's permanency requirements may only receive incentive on the incremental increase above the existing system's rated capacity (kW for generation or kWh for energy storage). ³⁰

The replaced system must also be fully decommissioned and removed from the site. The Program Administrator will confirm this has been completed as part of the Post-Installation Inspection.

3.2.6 Incentives from other sources

Host Customers, Applicants, and System Owners are required to disclose information about all other incentives they have received, plan to receive or have applied for.

³⁰ All applicable Incentive Limitations apply. See Section 5.2 and Section 6.6, 6.7, 6.9.

For all projects, the sum of the SGIP incentive and any other incentive received for the project may not exceed the total eligible project costs for the SGIP-eligible system. If the total exceeds the eligible costs, the SGIP incentive will be reduced until the sum of the SGIP and other incentive equal the total eligible costs.

For projects (excluding Equity Budget projects) receiving incentives under other programs, the SGIP incentive may be reduced depending on the source of the other incentive.

- For other incentives funded 100% by Investor Owned Utility (IOU) ratepayers, the total SGIP incentive will be reduced by the full amount of the other incentive.
- For other incentives funded by Non-IOU Ratepayers, the total SGIP incentive will be reduced by 50% of the amount of the other incentive.

In order to protect against entities creating governance structures or affiliations that would allow them to achieve more funding than the capped amount, it is required that Host Customers, Applicants, and System Owners disclose information about all other incentives and eligible tax credits taken advantage of by them or any of their affiliates applicable to the project. **Failure to disclose such information will be considered an infraction and is subject to the penalties indicated in Section 9.**

3.2.7 Developer Cap

Any single Developer is limited to 20% of the SGIP incentive funding for a given budget category in each statewide incentive step. The Developer cap will be calculated separately for generation projects, large scale energy storage projects, small residential energy storage projects, and equity budget projects. Applicants may not submit applications for Developers in excess of the statewide Developer cap for the active step, and Program Administrators shall not issue conditional reservations to projects by a Developer that has already applied for reservations in a given step that exceed 20%. The Developer cap will be established by budget step and posted prior to program opening. The Developer cap will remain fixed for each budget step once the step is opened even if total available funds change.³¹, ³² Please see *Section 4.1.5* for the definition of a Developer.

The Equity Budget and Equity Resiliency Budgets are exempt from the Developer Cap.

³¹ The developer cap will take into account all authorized collections approved in D.14-12-033 and D.17-04-017 as well as all other funds made available through project cancellation before the step opens.

³² Energy Division Disposition of Advice Letters (CSE Advice Letter 97-E-A, SoCalGas Advice Letter 5548, SCE Advice Letter 4114-E, and PG&E Advice Letter 4187-G/5699-E) approved the reallocation of funds from the Step 5 large-scale energy storage budgets to the Step 5 small residential budget carve-outs within each PA Territory resulting in an authorized, one-time adjustment of the Step 5 small residential budget Developer Cap on January 2, 2020.

3.3 PBI Assignment

If there is a change in ownership of the property which hosts the SGIP equipment, the new owner may continue to receive the Performance-Based Incentives (PBI) if the owner completes a new interconnection agreement. If the seller relocates the equipment, the owner may continue to receive the PBI incentive payments if the equipment is relocated within the same Program Administrator's service territory within six months and the owner completes an interconnection agreement at the new address. In either case, the PBI payment sunset date will not be extended.

4 Program Eligibility

In order to qualify for incentives, all program eligibility criteria must be satisfied. The following section details these requirements.

4.1 Program Participant Criteria

4.1.1 Host Customer

Any retail electric or gas distribution class of customer (industrial, agricultural, commercial or residential) of PG&E, SCE, SoCalGas, or SDG&E is eligible to be the Host Customer and receive incentives from the SGIP.³³ The Host Customer must be the utility customer of record at the site where the SGIP system is or will be located. In the event that the Host Customer's name is not on the utility bill, a letter of explanation is required that addresses the relationship of the Host Customer to the named utility customer.

The Host Customer is the exclusive incentive reservation holder who is party to the SGIP Contract. The Host Customer has the authority to designate the Applicant, energy service provider, and/ or Developer and change any of these parties at any time with prior written notice to the Program Administrator. For multifamily buildings that are installing a system on behalf of tenants and are enrolled in a VNEM tariff, the property owner may be the Host Customer.

4.1.1.1. Host Customer Eligibility for the Equity Budget

Non-residential projects:

For the Equity Budget, a non-residential project must meet one of the following customer criteria:

- Local governmental agency³⁴
- State governmental agency³⁵
- Tribal government agency
- Educational institution³⁶
- Non-profit organization³⁷
- Small business³⁸

Additionally, the project site must be located in a disadvantaged, tribal, or low-income community or document that at least 50% of the census tracts it serves are disadvantaged, tribal, or low-income communities. A disadvantaged community is defined as any census tract that ranks in the statewide top

³³ "...retail electric or gas distribution class of customer..." means that the Host Customer pays for and receives distribution services, as defined by their respective utility rate schedule.

³⁴ Public Contracts Code § 22161(f)

³⁵ Government Code §11000

³⁶ Any institution that would otherwise be eligible for funding through the California Clean Energy Jobs Act (Proposition 39), or a college or university accredited to operate in California.

³⁷ An organization that is registered and in good standing with the California Secretary of State as a domestic non-profit entity.

³⁸ A business or manufacturer, including affiliates, with average annual gross receipts of \$15 million of less, over the last three tax years.

25% most affected census tracts in the most recently released version of the environmental health screening tool, CalEnviroScreen, plus census tracts that score within the highest 5% of CalEnviroScreen's pollution burden, but do not receive an overall CalEnviroScreen score. A low-income community is defined as census tracts with median household incomes at or below 80% of the statewide median income, or with median household incomes at or below the threshold designated as low-income by the Department of Housing and Community Development's list of state income limits adopted pursuant to Section 50093.

Residential Projects:

For the Equity Budget, residential projects are classified as multi-family low-income housing or single-family low-income housing.

Eligible multi-family housing is defined as a multi-family residential building of at least five rental housing units that is operated to provide deed-restricted low-income residential housing³⁹, and is either located in a disadvantaged community, or is a building where at least 80% of the households have incomes at or below 60% of the area median income⁴⁰. Any customer account in such buildings will be eligible for the Equity Budget.

Additionally, the Equity Budget is available to customers living in single-family low-income residences sold at an affordable housing cost to a lower income household that is subject to:

- a) a resale restriction⁴¹, or;
- b) an equity sharing agreement, for which the homeowner does not receive a greater share of equity than described in paragraph (2) of subdivision (c) of Section 65915 of the Government Code, with a public entity or nonprofit housing provider organized under Section 501(c)(3) of the Internal Revenue Code that has as its stated purpose in its articles of incorporation on file with the office of the Secretary of State to provide affordable housing to lower income households⁴².

Additionally, the Equity Budget is available to residential customers who have participated or who are eligible for the SASH or DAC-SASH programs and for multi-family properties that have reserved funds in the MASH or SOMAH programs. Customers must provide a copy of their incentive reserved status for the SASH, er-DAC-SASH, MASH, or SOMAH program to demonstrate that they are eligible for one of these programs.

4.1.1.2. Inclusion of Indian Country in California

a) Indian Country in California, as defined in 18 USC 1151, is considered as being a disadvantaged community for purposes of the SGIP Equity Budget. Thus, projects in these areas are eligible for

³⁹ As described in clause (i) of subparagraph (A) of paragraph (3) of subdivision (a) of § 2852 of the Public Utilities Code.

⁴⁰ As defined in subdivision (f) of §50052.5 of the Health and Safety Code.

⁴¹ Certain and specific "presumed resale restrictions" meet this requirement. See section 5.4.1 for more information.

⁴² Subparagraph (c) of paragraph (3) of subdivision (a) of 2852 of the Public Utilities Code

the Equity Budget. However, non-Indian residences or businesses on privately owned fee land in Indian Country are not eligible under this expanded definition. If the in-holding has multiple owners, at least one owner must be a tribe or tribal member for the project to be eligible.

4.1.1.3. Eligibility for the Equity Resiliency Budget

Residential customers are eligible for the Equity Resiliency Budget if they meet each of the following:

- a. Are located in a Tier 3 or Tier 2 HFTD⁴³ or were subject to two or more discrete PSPS events prior to the date of application for SGIP incentives, and
- b. Are one of the following:
 - i. Eligible for the SGIP Equity Budget; or
 - ii. Medical Baseline customer44; or
 - A customer that has notified their utility of serious illness or condition that could become life-threatening if electricity is disconnected⁴⁵; or
 - iv. Rely on electric pump wells for water supplies.

Non-Residential customers are eligible for the Equity Resiliency Budget if they meet each of the following:

Located in a Tier 3 or Tier 2 HFTD;⁴⁶ or were subject to two or more discrete PSPS events prior to the date of application for SGIP incentives, and

- a. Provides critical facilities or critical infrastructure during a PSPS event to at least one community⁴⁷ that is located at least partially in a Tier 3 or Tier 2 HFTD or were subject to two or more discrete PSPS events prior to the date of application for SGIP incentives, and
- b. The community is also eligible for the equity budget; 48 and
- c. The Non-Residential customer is one of the following:
 - i. Police stations; or
 - ii. Fire stations; or
 - iii. Emergency response providers⁴⁹ with the addition of tribal government providers; or

⁴³ https://ia.cpuc.ca.gov/firemap/

⁴⁴ Defined in D.86087, 80 CPUC 182

⁴⁵ Defined in D12-03-054

⁴⁶ https://ia.cpuc.ca.gov/firemap/

⁴⁷ Defined in D.17-10-004 (Table1)

⁴⁸ Non-residential customers with critical resiliency needs do not need to meet eligibility criteria for the SGIP equity budget themselves. Non-residential customers with critical resiliency needs are not required to exclusively serve communities that are located in a Tier 3 or Tier 2 HFTD and are also eligible for the equity budget.

⁴⁹ Defined in D.19-05-042

- iv. Emergency operations centers; or
- v. 911 call centers (also referred to as Public Safety Answering Points); or
- vi. Medical facilities including hospitals, skilled nursing facilities, nursing homes, blood banks, health care facilities, dialysis centers and hospice facilities; or
- vii. Public and private gas, electric, water, wastewater or flood control facilities; or
- viii. Jails and prisons; or
- ix. Locations designated by the IOUs to provide assistance during PSPS events (CRCs);
 or
- x. Cooling centers designated by state, local or tribal governments; or
- xi. Homeless shelters supported by federal, state, or local, or tribal governments; or
- xii. Grocery stores, corner stores, markets and supermarkets that have average annual gross receipts of \$15 million or less as calculated at the single location applying for SGIP incentives; or
- xiii. Independent living centers; or
- xiv. Food banks.

4.1.2 System Owner

The System Owner is the owner of the SGIP incentivized equipment. In the case when a vendor sells a system to a Host Customer, the Host Customer is the System Owner. In the case of a leased system, the lessor is the System Owner. The System Owner shall be designated on the Reservation Request Form.

In the event that the System Owner is not the Host Customer and withdraws from the project, the Host Customer will retain sole rights to the incentive reservation and corresponding incentive reservation number. To preserve such incentive reservation and corresponding reservation number, the Host Customer must submit revised application documentation to the Program Administrator.

4.1.3 Applicant

An Applicant is the person or entity that is responsible for completing and submitting the SGIP application and serves as the main point of contact for the SGIP Program Administrator throughout the application process. Host Customers may act as the Applicant, or they may designate a third party to act as the Applicant on their behalf. Host Customers may elect to change the Applicant at any time with prior written notice to the Program Administrator.

4.1.4 *Payee*

A Payee is the person or entity to whom the SGIP incentive check is made payable. The Payee may be any person or entity designated by the Host Customer on the Incentive Claim Form.

4.1.5 **Developer**

For applications received in Steps 1 and 2:

A Developer is the corporate entity that holds the contract for purchase and installation of the system, and/or alternative System Ownership Agreement (such as a Power Purchase Agreement) with the host customer and handles the project's development activities. The Developer must fully disclose their participation in developing the project and/or ownership in the project, or that of a combination of affiliated installers/developers. The customer contract will be verified at Proof of Project Milestone to confirm the Developer's representations. When applicable, the Developer cap will apply to any combination of affiliated developers under the same majority ownership.

For applications received in Step 3 or later:

A Developer is, if not individual homeowners applying for SGIP incentives for systems located on their own property, the corporate entity registered and in good standing with the Secretary of State of California that handles a substantial amount of the project's development activities.

An exclusive list of a project's development activities to be used for the purpose of applying this definition to the Program's developer cap is as follows:

- 1. Approaching or communicating with the host customer about the project and learning about its needs and energy profile
- 2. Developing the specifications for a system based on the customer's needs and interests
- 3. Soliciting bids from multiple manufacturers for the specified system
- 4. Gaining the customer's commitment to purchase or lease the specified system, usually but not necessarily by signing a purchase order with a customer or other form of agreement
- 5. Purchasing the specified system from the manufacturer to fulfill the obligation to provide a system to the customer
- 6. Securing permits for the system on behalf of the customer
- 7. Securing interconnection permission for the system on behalf of the customer
- 8. Submitting SGIP applications on behalf of the customer
- 9. Liaising with the SGIP administrators on incentive reservations
- 10. Liaising with the SGIP administrators on data reporting requirements
- 11. Supplying project data to SGIP evaluators
- 12. Physically constructing the system at the customer's premises
- 13. Installing the system at the customer's premises

For the purpose of this definition, the Program Administrators, with oversight from Energy Division, have discretion to determine whether a participant handles such activities, and whether a participant handles a substantial amount of these activities for a given project. If a reassignment of Developer status for a given

project occurs after an SGIP application is submitted but before a confirmed reservation is issued, and leads to a participant retroactively exceeding its developer cap for a given incentive step, a sufficient number of reservation requests for that participant shall be cancelled to bring the participant under its developer cap.

4.1.5.1 Developer Application Process

All SGIP projects must select a pre-approved Developer before the application may be submitted. A list of approved Developers will be available at www.selfgenca.com. Entities interested in becoming an approved SGIP Developer must submit the Developer Eligibility Application to the Program Administrators, containing the following information:

- If a parent company for the participant exists, the ultimate owner of the parent company if the
 parent company is nested within other parent-child corporate relationships revelation of the
 ultimate owner is required
- Any ownership interest including a minority interest in the developer held by another developer or ultimate owner of another developer
- The identity of the entity, if reasonably foreseeable, that will execute any of the following development activities if it is not the developer named in the Developer Eligibility Application:
 - Approaching or communicating with the host customer about the project and learning about its needs and energy profile
 - o Developing the specifications for a system based on the customer's needs and interests
 - Soliciting bids from multiple manufacturers for the specified system
 - Gaining the customer's commitment to purchase the specified system, usually by signing
 a purchase order with a customer or other form of agreement; or owning the system
 during its lifetime and signing a lease or other similar agreement with the customer
 - Purchasing the specified system from the manufacturer to fulfill the obligation to provide a system to the customer
 - Securing permits for the system on behalf of the customer
 - Securing interconnection permission for the system on behalf of the customer
 - Filing SGIP applications
 - Liaising with the SGIP administrators on incentive reservations
 - Liaising with the SGIP administrators on data reporting requirements
 - Supplying project data to SGIP evaluators
 - Physically constructing the system at the customer's premises
 - Installing the system at the customer's premises
- If the developer plans to divide project development activities with another entity in different ways for different projects, the developer must note that on the application form and submit

- supplemental information with the Developer Eligibility Application describing those other ways for other projects.
- To the extent not described in the section asking for information on execution of project
 development activities, a description of relationships with other SGIP participants, including
 commercial or financial relationships between developers that do not share the same majority
 ownership, in developing and/or owning any existing projects funded in whole or in part by SGIP
 or reasonably foreseeable future projects that may be funded in whole or in part by SGIP.

Additionally, if requested by the Program Administrator, Developers must provide the following documentation:

- 1. Articles of incorporation,
- 2. Certificate of incorporation,
- Operating agreements or similar applicable organizational document of Developer and most current financial statements.

Only one participant per project can be determined by the Program Administrators or Energy Division to handle a substantial portion of the project's development activities.

Acceptance by the Program Administrators of a Developer Eligibility Application does not guarantee that a participant qualifies as a Developer of any given project.

The Developer must fully disclose its participation in developing the project and/or ownership in the project, or that of a combination of affiliated installers/developers. The Developer must disclose relationships with other SGIP participants in developing and/or owning any existing projects funded in whole or in part by SGIP or reasonably foreseeable future projects that may be funded in whole or in part by SGIP to the Program Administrators on an ongoing basis and in the Developer Eligibility Application. For example, if a Developer purchases the contract for purchase and installation of a system from another Developer, each Developer must describe the transaction in its Developer Eligibility Application, or inform the Program Administrators of the transaction if the Developer Eligibility Application has already been accepted. Note that the relationships between Developers to be disclosed to the Program Administrators include commercial or financial relationships between Developers that do not share the same majority ownership.

The customer contract will be verified at Proof of Project Milestone to confirm the Developer's representations. When applicable, the Developer cap will apply to any combination of affiliated developers under the same majority ownership. Majority ownership may be evidenced by the ownership interest in different Developers by: a single parent corporation, the distinct subsidiaries of a single parent corporation, a private equity firm, an investment bank, a group of individual shareholders, or any other entity that the

Program Administrators or Energy Division in their discretion determine to constitute the same majority ownership of different Developers.

In order to ensure that majority ownership can be accurately assessed by the Program Administrators, all Developers will be required, upon request, to disclose the identity of all of their equity holders, including the ultimate ownership interests in those equity holders.

Failure to disclose relationships with other participants in a Developer Eligibility Application shall be considered an infraction.

4.2 Equipment Eligibility

The SGIP intends to provide incentives for reliable, permanent and safe systems that are professionally installed and that comply with all applicable Federal, State and local regulations. Host Customers and System Owners are strongly encouraged to become familiar with applicable equipment certifications, design, and installation standards for the systems they are contemplating. The following section describes the specific equipment eligibility criteria for systems that want to participate in the SGIP.

4.2.1 Commercial Availability

Commercially available, factory-new equipment is eligible for incentives. "Commercially available" means that the major system components are acquired through conventional procurement channels.

All eligible technologies must be certified for safety by a nationally recognized testing laboratory (NRTL). An application may be submitted for a device that has not yet received certification from an NRTL if the certification process is underway, however proof of certification must be submitted at the latest with the Incentive Claim documents. Failure to submit proof of certification with the Incentive Claim documents will result in cancellation of the project.

If NRTL certification is not applicable for the technology type, it must be supported by a 10-year warranty as consistent with Rule 21 interconnection standards. In the event that Rule 21 interconnection standards or NRTL certification do not require a separate manufacturer's warranty, a single 10-year service warranty is sufficient. If Rule 21 interconnection standards or NRTL certification require a separate 10-year manufacturer's warranty in addition to the 10-year service warranty, then the obligation for dual warranties must be met by the project.

4.2.2 Interconnection

All systems receiving incentives under the SGIP that discharge electricity⁵⁰ must be connected to the local electric utility's distribution system and must be installed on the Host Customer's side of the electric utility

⁵⁰ For example, TES does not discharge electricity and therefore does not require an interconnection agreement.

meter. The interconnection, operation, and metering requirements for the systems shall be in accordance with the local electric utility rules for customer generating facility interconnections. Energy storage systems must also be configured to operate in parallel with the grid. SGIP eligibility includes systems that interconnect to the local electric utility's distribution system under the requirements of a virtual net metering tariff.

In order to connect a system to the electric utility distribution system, Host Customers and/or System Owners will be required to execute certain documents such as, but not limited to, an "Application to Interconnect a Generating Facility" and a "Generating Facility Interconnection Agreement" with the local electric utility. Written certification of interconnection and parallel operation to the Program Administrator prior to the Reservation Expiration date will be required. Applicants, Host Customers, and System Owners are solely responsible to submit interconnection applications to the appropriate electric utility interconnection department as soon as the information to do so is available to prevent any delays in system parallel operation. For more information on electric grid and/or natural gas pipeline interconnections, please contact your local utility.

Systems will be eligible to receive a reservation up to 12 months after receiving authorization to operate in parallel with the grid from the electric utility.

4.2.3 **Permanent Installation**

The intent of the SGIP is to provide incentives for equipment installed and functioning for the duration of its useful life. Only permanently installed systems are eligible for incentives. This means that the system must demonstrate to the satisfaction of the Program Administrator adequate assurances of both physical and contractual permanence prior to receiving an incentive.

Physical permanence is to be demonstrated by electrical, thermal and/or fuel connections in accordance with industry practice for permanently installed equipment and be secured to a permanent surface (e.g. foundation). Any indication of portability, including but not limited to temporary structures, quick disconnects, unsecured equipment, wheels, carrying handles, dolly, trailer or platform, will deem the system ineligible.

Contractual permanence, corresponding to a minimum of the length of the applicable warranty period (10 years), is to be demonstrated as follows:

- System Owner agrees to notify the Program Administrator in writing a minimum of 60 days prior to any change in either the site location or change in ownership of the system if the change(s) takes place within the applicable warranty period.
- All agreements involving the system receiving an incentive are to be provided to the Program Administrator for review as soon as they become available. These agreements include, but are not

limited to, system purchase and installation agreements, warranties, leases, energy or services agreements, energy savings guarantees and system performance guarantees.

4.2.4 Ineligible Equipment

The following equipment is not eligible for participation in the SGIP:

- Backup systems intended solely for emergency purposes⁵¹
- Any system/equipment that is capable of operating on, or switching to, diesel fuel or diesel cycle for start-up or continuous operation
- Field demonstrations for proof-of-concept operation of experimental or non-conventional systems partially or completely paid by research and development funds
- Rebuilt, refurbished or relocated equipment (e.g. second life batteries)
- Equipment that has been interconnected for more than 12 months

4.2.5 System Sizing for Projects without Peak Demand Information

Sites with 12-months of previous energy usage data (kWh) but without peak demand (kW) information available (e.g., customers on rate schedules without a demand component) will have an equivalent peak demand calculated using the highest amount of energy consumed (kWh) in a given interval in the previous 12 months of consumption at the project site divided by the duration of that interval, in hours. The most granular interval for which there is 12 months of available data should be used.

For example, if a residential customer's meter collected hourly interval data for a 12-month period, and this data revealed that the greatest hourly consumption data was 5.5kWh during the previous 12 months, then the peak demand estimate would be 5.5 kW. If the data revealed that the greatest consumption within a 15-minute interval was 5.5kWh during the previous 12 months, then the peak demand estimate would be 22 kW.

For customers with less than 12 months of history with the utility or customers that have meters incapable of recording interval usage data, methodology from Section 220 of the National Electric Code (NEC) may be used to determine instantaneous peak demand.

4.2.6 Eligibility for New Technologies

Systems consisting of new technologies not already included in the list of eligible SGIP technologies listed in Section 3.1.1 and Section 3.1.2 may become eligible for the SGIP if its first commercial installation

⁵¹ Operate as short-term temporary replacement for electrical power during periods of Electric Utility power outages. In addition to emergency operation they ordinarily only operate for testing and maintenance. Backup generators do not produce power to be sold or otherwise supplied to the grid or provide power to loads that are simultaneously serviced by the Electric Utility grid. Backup generators only service customer loads that are isolated from the grid either by design or by manual or automatic transfer switch. Adopted Affidavit Form in CPUC Resolution E-4717.

occurred less than ten years prior to SGIP funding. New technologies must meet all applicable eligibility and program requirements. Developers of such technologies seeking eligibility must follow the Program Modification Guidelines as outlined in *Section 4.2.7*.

4.2.7 Program Modification Guidelines

For the consideration of new technologies and/or SGIP rule changes, the following Program Modification Guidelines as outlined in Decision 03-08-013 should be followed:

All Program Modification Requests (PMRs) must be submitted in writing, using the current PMR format to the SGIP Working Group for review at least 10 business days prior to the SGIP Working Group meeting or the request will roll over to the next monthly SGIP Working Group meeting.

All parties requesting a program modification will be required to meet with the SGIP Working Group during a monthly SGIP Working Group meeting to determine if the Working Group would support the PMR.

The SGIP Working Group will first determine whether or not the proposed PMR requires a modification to a prior Commission Decision. If the PMR is minor and non-substantive and does not require modifications to prior Commission orders, then:

- The Working Group will review the PMR. If accepted, the Working Group will make the appropriate changes to the Handbook.
- If the Working Group needs more information, the party proposing the PMR would have the
 opportunity to present at the following Working Group meeting with additional information which
 supports its request for a program change.
- The Working Group will make a decision to accept or deny the PMR based on the new information presented in the follow-up presentation.
- The proposed program change and the Working Group recommendation(s) and rationale will be captured in the Working Group meeting minutes.
- If the party objects to the Working Group's decision to deny the PMR, the party may write a
 letter to Energy Division stating why their program change should be included in the SGIP.
 Information that supports the party's reasons to accept the program change must be included
 in the letter.
- Energy Division will then make a final decision on whether to approve the PMR.
- Energy Division will report its final decision at the following SGIP Working Group meeting,
 which will be captured in the SGIP Working Group meeting minutes.
- If the PMR is accepted, appropriate revisions to the SGIP Handbook will be made.

If the proposed change requires modification to a prior Commission order or if the PMR addresses large programmatic or substantive issues, then:

- The Working Group will review the PMR and make a recommendation to support or oppose the PMR in the same meeting.
- The Working Group recommendation and rationale will be captured in the Working Group meeting minutes.
- Subsequent to the meeting, the Working Group will write up a summary of the discussion of the PMR at the Working Group meeting, a list of comments in support or against the PMR, as well as the Working Group's overall recommendation with rationale, which will be presented to the Applicant.
- The party proposing the PMR has the choice to move forward and submit a Petition to Modify (PTM) for Commission review regardless of the Working Group's recommendation, but the Working Group's summary must be included in the PTM.
- The Energy Division participates in Working Group meetings and is welcome to participate in the discussion related to the PMR as well as in generating the "list of issues." The Energy Division does not need to participate in the "recommendation" portion of the Working Group's PMR review.
- Once the PTM is filed with the Commission, the normal PTM process will transpire, only it will
 have the benefit of the idea being somewhat vetted before submittal. All parties have a chance
 to comment on PTMs according to the Commission's Rules of Practice and Procedure.
- The Commission will review and address the PTM in a decision.

5 **Energy Storage Technologies**

The following sections outline the eligibility requirements, rating criteria, incentive calculation guidelines and application documentation requirements and metering requirements specific to SGIP energy storage projects. Additionally, all projects are subject to the general program requirements as outlined in sections 2, 3 and 4.

5.1 Rating Criteria for Energy Storage Projects

5.1.1 Rated Capacity (W)

The rated capacity (W) for energy storage technologies is calculated as follows:

- DC/AC systems: The nominal voltage multiplied by the amp-hour capacity multiplied by the applicable efficiency divided by the duration of discharge ((V_{DC} x Amp-Hours x (1 kW/1000W) x Applicable Efficiency) / Duration of Discharge).
 - The following specifications must be provided to calculate rated capacity:
 - Duration of discharge (hours)
 - DC dischargeable amp-hour capacity, associated with the duration of discharge specified, including all losses and ancillary loads (such as power conditioning and thermal management)
 - Nominal voltage (V_{DC})
 - Applicable efficiency (if necessary), which accounts for conversion, transformation, or other efficiency losses (e.g. Inverter CEC weighted efficiency, DC-DC converter efficiency, etc.)
 - The continuous maximum power output capability of the system.⁵² For DC systems, this is rated in DC, and for AC systems, this is rated in AC.
- HVAC-integrated S-TES: Calculated using the Conversion Tables for HVAC-integrated S-TES
 found in Appendix D, based on the SEER rating and tonnage of the HVAC unit with which the STES system will be integrated and the climate zone in which the project is located.
- Refrigeration TES: Calculated using the Refrigeration TES kW Worksheet, and is based on the
 following parameters of the refrigeration system(s) with which the Refrigeration TES system will be
 integrated: Energy Efficiency Ratio (EER) of the compressor group, nameplate BTUh capacity of
 the compressor group, and maximum recorded output of the compressor group (as a percentage
 of nameplate compressor group horsepower), the number of condenser fans, nameplate
 condenser fan power, and maximum recorded output of the condenser fan group (as a percentage
 of nameplate condenser fan group output).

⁵² Maximum continuous output might be determined by the maximum continuous discharge power/current of the system or by the continuous rated output of the inverter.

5.1.2 Energy Capacity (Wh)

The energy capacity (Wh) for energy storage technologies is calculated as follows:

- DC/AC systems: The nominal voltage multiplied by the amp-hour capacity multiplied by the applicable efficiency (V_{DC} x Amp-Hours x (1 kW/1000W) x Applicable Efficiency).
 - o The following specifications must be provided to calculate energy capacity:
 - Duration of discharge (hours)
 - DC dischargeable amp-hour capacity, associated with the duration of discharge specified, including all losses and ancillary loads (such as power conditioning and thermal management)
 - Nominal voltage (V_{DC})
 - Applicable efficiency (if necessary), which accounts for conversion, transformation, or other efficiency losses (e.g. Inverter CEC weighted efficiency, DC-DC converter efficiency, etc.)
- HVAC-integrated S-TES: The rated capacity (W) multiplied by the avoided compressor(s) run-hours that the system can provide in a single discharge.
- Refrigeration TES: The rated capacity (W) multiplied by the avoided compressor(s) and condenser(s) run-hours that the system can provide in a single discharge.

5.2 Eligibility Requirements for Energy Storage Projects

Energy storage projects may be stand-alone or paired with generating systems and must be capable of discharging fully at least once per day. Energy storage systems paired with wind generation must have the ability to handle hundreds of partial discharge cycles each day. Residential energy storage projects, whether stand-alone or paired, must comply with additional requirements specified in the Residential Energy Storage Eligibility Affidavit designed to ensure that all residential energy storage systems participating in the SGIP will be used for more than just back-up emergency purposes.

5.2.1 **Greenhouse Gas Signal**

A Greenhouse Gas (GHG) signal provides energy storage developers with information they need to charge the storage system during low-GHG emission periods and to discharge during high-GHG emission periods to reduce GHGs and remain in compliance with the statute.⁵³

- The signal is a digitally-accessible (in API format), real-time, marginal GHG emissions factor for the appropriate California Independent System Operator (CAISO) sub-regions, at five-minute intervals, in units of kgCO2/kWh;
- The signal is calculated using the same heat rate-based methodology as in the most recent Avoided Cost Calculator (ACC), but with updated parameters and data sources more suitable for real-time use;

_

⁵³ Pub. Util. Code Section 379.6(b)(3).

- This signal provides the marginal emissions per kWh calculated based on a natural gas-fired power plant producing energy at a price equaling the real-time (five-minute) CAISO Locational Marginal Price with costs equal to the most recent publicly available data on gas prices, CO₂ prices, and variable operating costs constrained by reasonable maximum and minimum efficiencies. When the calculated heat rate is zero or below, instead it is assumed that the marginal generator is renewable and the marginal emissions rate is zero; and,
- The marginal GHG emission signal includes non-CAISO territories where SGIP incentives are available, using the same methodology for non-CAISO regions as for CAISO areas, using the closest representative input data.

A five-minute real time GHG signal is used as the basis of determining compliance with the GHG requirements adopted in the decision; as such, this signal is referred to as the "GHG compliance signal". The GHG compliance signal is used for verification of both project and fleet GHG emission compliance. The following forecasting tools are also available:

- Fifteen-minute forecast, with five-minute granularity and updated every fifteen minutes;
- One hour-ahead forecast, with five-minute granularity and updated every fifteen minutes;
- Day-ahead forecast, with five-minute granularity and updated every fifteen minutes;
- Longer Term Forecasts: 72-Hour Ahead, Month-Ahead and Year-Ahead

The GHG signal and forecasting tools are available online at selfgenca.com.

5.2.2 Greenhouse Gas Emission Standards for new non-residential applications

During the PBI term, non-residential projects are required to reduce GHGs by a minimum of five kilograms of CO₂ per rated energy capacity (kg/kWh) annually to recoup full payment.

In order to calculate the GHG emissions, all new non-residential projects are to submit PBI data, including cycling data and data on net energy charged and discharged (kWh) and net real power charged and discharged (kW), in 15-min intervals on a monthly basis using the SGIP online application database PDP Portal. The SGIP PAs shall verify each new non-residential project's GHG reductions annually using PBI data in combination with the five-minute real time GHG signal and will provide each project with monthly feedback on GHG performance.

After the last year that a PBI payment is issued, new non-residential projects should continue to annually reduce GHG emissions by five kg/kWh through the tenth year of their permanency period verified on a fleet basis. New non-residential projects must submit cycling data on a quarterly basis to the SGIP online application data PDP Upload Portal. The PAs will use this data in combination with the GHG signal to provide quarterly feedback to developers on their GHG emissions. The annual SGIP storage impact evaluation will report the fleet GHG emissions performance of new non-residential projects in years six

through ten of their permanency periods, by developer. Large non-residential projects that are not otherwise part of a developer fleet are also subject to these requirements.

5.2.3 Greenhouse Gas Emission Standards for non-residential legacy applications

All non-residential energy storage systems with applications that were approved prior to April 1, 2020 will have three options to comply with the SGIP GHG emission standards:

Option 1: RTE Pathway- Operational Requirements

• The project will continue to comply with the operational requirements and the SGIP handbook procedures in place at the time of project approval, including the annual RTE requirement of 66.5 percent, with the exception that projects with a 260 per year cycling requirement may reduce this to 130 cycles per year.

Option 2: Storage Rate/DR Pathway- Operational Requirements

- These are identical to those of Option 1 with the exception that the project may substitute the RTE requirement in place when the project was approved with one of the following:
 - Enrollment in an economic DR program that is integrated into the CAISO or the DRAM mechanism; or,
 - Enrollment in an approved storage rate including:
 - PG&E: Option S or A-1-STORE;
 - SCE: Option E or TOU-GS-1; and,
 - A rate available to non-residential customers with storage systems that reduces, eliminates or otherwise does not include non-coincident demand charges and that is approved for use through a Tier-2 advice letter process.⁵⁴

Option 3: GHG Pathway- Operational Requirements

- Projects that select Option 3 that are within their ten-year permanency requirement are required to emit zero kg/kWh GHGs or less at the developer fleet level on an annual basis; and,
 - o Must cycle a minimum of 130 times per year; and,
 - o the annual RTE requirement is eliminated

Regardless of what option a non-residential project selects, all verification and enforcement standards will fall under the SGIP Handbook version that was approved by the CPUC at the time of application approval. If a project wishes to change which pathway they are on, they must notify the appropriate PA by December 15th of any calendar year. Any changes made will go into effect the forthcoming calendar year.

5.2.4 System Size Parameters

System size requirements are applicable for systems that are rated above 10 kW. Systems that are rated at 10 kW or less are exempt from the sizing requirements.

⁵⁴ The SGIP PAs will update the Handbook with approved storage rates that reduce or eliminate non-coincident demand charges periodically. For the latest information regarding available rates, please reach out to the appropriate IOU for where the project is based.

Energy storage projects that are not receiving equity budget resiliency incentives or receiving the resiliency adder, whether paired or stand-alone, may be sized up to the Host Customer's previous 12-month annual peak demand (kW). Energy storage systems that are receiving equity budget resiliency incentives or that are receiving the resiliency adder may be sized greater than the Host Customer's previous 12-month peak demand if this is necessary due to modular component sizes to accommodate the customer's peak load, and provided they demonstrate proof of this need before the incentives are paid. For new construction or projects with future load growth, projects can be sized up to future peak demand, but the load must be substantiated before the incentive can be paid.

Projects sizing to tenant loads may aggregate each tenant unit at an assumed 5 kW load. For example, a building with 10 residential units may assume a load of 50 kW for sizing purposes. If the system is sized greater than the assumed aggregated load, the project is subject to normal sizing requirements and must justify the actual load the system will be serving.

HVAC-integrated S-TES systems must be sized no larger than the tonnage of their accompanying HVAC unit and cannot be integrated with HVAC units greater than 20 tons. Refrigeration TES systems must be sized no larger than the tonnage of their accompanying refrigeration system(s).

5.2.5 **Operational Requirements**

New non-residential systems are required to discharge a minimum of 104 full discharges per year. Residential systems are required to discharge a minimum of 52 full discharges per year. A "full discharge" is the equivalent of discharging the SGIP-incentivized energy capacity, whether it is during a single or multiple discharges.⁵⁵

Legacy projects required to discharge a minimum of 260 full discharges a year may opt to discharge a minimum of 130 full discharges a year by notifying their respective SGIP PA.

5.2.6 Paired with On-site Renewables

To be considered paired with and charging from on-site renewables, energy storage systems must either be claiming the Investment Tax Credit (ITC) or, if not claiming the ITC, charge a minimum of 75% from the on-site renewable generator.

5.2.7 **Demand Response Dual Participation**

Energy storage projects funded through the SGIP are eligible to provide demand response services or participate in demand response programs.

⁵⁵ Each discharge does not have to be a 100% depth of discharge, but the aggregate amount of discharges over the year must equate to 104 full discharges.

5.2.8 Upfront Eligibility Requirements for New Residential Projects

All new residential systems, regardless of system size, must have a single cycle round trip efficiency (SCRTE) of 85% or greater.

Additionally, all new residential IOU and non-IOU customers are required to enroll in a time-varying rate with a peak period starting at 4 pm or later and with a summer peak to off-peak price differential of 1.69 or more, if such rate is available. SGIP eligible customers may not utilize time-of-use (TOU) bill protection measures. However, for California Alternate Rates for Energy (CARE) eligible customers, if a CARE TOU rate meeting the above criteria is not available at the time of SGIP reservation request submittal, the customer may enroll in any CARE TOU rate. CARE TOU customers may not utilize TOU bill protection measures.

A list of SGIP-approved residential rates can be found on www.selfgenca.com

Customers that do not have access to an approved TOU rate may be eligible if they can demonstrate that the energy storage system charges only from an onsite solar photovoltaic system (defined as a battery that begins charging at dawn and continues charging from the solar unit until fully charged, and that never charges from the grid) or has set relevant storage parameters to the appropriate manufacturer-certified functionalities at the time of installation to enable solar-self-consumption mode (defined as systems using a battery to store solar generation and discharging the stored power only to meet onsite load).

5.2.9 Greenhouse Gas Emission Standards for Legacy Residential Projects

Residential energy storage systems with an application approval date prior to April 1, 2020 are no longer subject to the round-trip efficiency requirements in place at the time the reservation letters were issued. The annual cycling requirement of 52 cycles per year must still be maintained for all legacy residential project completed. These projects are encouraged to enroll in an SGIP-approved rate as described in section 5.2.8.

5.3 Incentive Calculation for Energy Storage Projects

Incentives for a proposed energy storage system are calculated by multiplying the energy capacity (Wh) of the system by the incentive rate for the appropriate step.

Incentive = Energy capacity (Wh) * incentive rate

5.3.1 Incentive Declines Based on GHG Performance

Non-residential energy storage systems must reduce GHG emissions by 5 kg CO₂/ rebated energy capacity (kWh) on an annual basis. Projects that do not reduce GHG by 5 kg CO₂/kWh will have their PBI payment reduced by \$1/kg CO₂ emitted above this level. The reduction is capped at 100% of the annual PBI

payment. PBI payment deductions are permanently forfeited and returned to the SGIP incentive budget. The SGIP PAs have the discretion to decrease the amount of any given PBI payment deduction in exceptional circumstances and with the written approval of CPUC Energy Division. Exceptional circumstances include, but are not limited to, causes not reasonably under the developer or customer's control, and causes that were not reasonably foreseeable.

5.3.2 Incentive Declines Based on Storage Duration

Energy storage incentives are reduced as the duration of energy storage (Wh) increases in relation to the rated capacity (W) according to the following schedule:

Table 5.3.2.1 Energy Storage Incentive Duration Decrease

Storage Duration	Incentive Rate (Pct. Of Base)
	, ,
0-2 hours	100%
Greater than 2 hours to 4 hours	50%
Greater than 4 hours to 6 hours	25%
Greater than 6 hours	0%

Example 1: 2-Hour System

200 kWh storage system at \$.40/Wh = 200.000 Wh * \$.40/Wh = \$80.000

Example 2: 4-Hour System

400 kWh storage system at \$.40/Wh =
200,000 Wh * \$.40/Wh = \$80,000
200,000 Wh * \$.20/Wh = \$40,000
\$80,000 + \$40,000 = \$120,000

5.3.3 Equity Budget Incentive Declines Based on Storage Duration

Energy storage incentives are reduced as the duration of energy storage (Wh) increases in relation to the rated capacity (W) according to the following schedule. For the equity budget, the The-duration step-down incentive structure is as follows, such that systems with discharge duration capacities of four to six hours

receive 50 percent of the base incentive rate. No incentive is provided for the discharge capacity greater than 6 hours.

Table 5.3.3 Energy Storage Equity Incentive Duration Decrease

Discharge Duration	Percent of Base
(hours)	Incentive
0 to 2 hours	100
Greater than 2 to 4 hours	100
Greater than 4 to 6 hours	50
Greater than 6 hours	0

Example 1: Four-hour system at \$.40/Wh incentive

100 kW discharge rating

400 kWh discharge capacity

400,000 Wh * \$.40/Wh = \$160,000

Example 2: Six-hour system at \$.40/Wh incentive

100 kW discharge rating

600 kWh discharge capacity

400,000 Wh * \$.40/Wh = \$160,000

200,000 Wh * \$.20/Wh = \$80,000

\$160,000 + \$80,000 = \$240,000

Example 1: 2-Hour System

200 kWh storage system at \$.40/Wh =

200,000 Wh * \$.40/Wh = \$80,000

Example 2: 4-Hour System

400 kWh storage system at \$.40/Wh =

200,000 Wh * \$.40/Wh = \$80,000

200,000 Wh * \$.20/Wh = \$40,000

\$80,000 + \$40,000 = **\$120,000**

The Commission requires all equity budget, and equity resiliency, and general market projects to meet all SGIP GHG emission reduction, cycling, and other system and operational requirements as these ensure that storage systems receiving incentives will not be used only or primarily to provide backup power.⁵⁶

Incentive Declines and Caps Based on Energy Capacity (Wh)

Energy storage incentives are paid up to 6 MWh of capacity with tiered incentive rates. For energy storage projects that are greater than 2 MWh, incentives decline according to the following schedule:

Table 5.3.4.1: Tiered Incentive Rates

Energy Capacity	Incentive Rate (Pct. of Base)
0 to 2 MWh	100%
Greater than 2 MWh to 4 MWh	50%
Greater than 4 MWh to 6 MWh	25%

5.3.5 Performance-Based Incentive Payment (PBI)

All new non-residential energy storage projects will be paid 50% of the incentive upon project completion and verification. The remaining 50% will be paid on a performance-based incentive (PBI). Annual kilowatt hour-based payments will be structured so that under the expected annual operational requirements⁵⁷ a project would receive the entire stream of performance payments in five years.

To calculate the basis (\$/kWh) of the annual PBI payments the following calculation is made:

\$/kWh = remaining 50% of incentive / total anticipated kWh discharge/offset

Total anticipated kWh discharge/offset = energy capacity (kWh) * 104 full discharges⁵⁸ * five vears

For a 5-year period the PBI payment will be paid annually based on recorded kWh of electricity discharged or offset over the previous 12 months.

PBI Payment = \$/kWh * actual annual kWh discharged/offset

⁵⁶ D.19-09-027, Conclusion of Law 19.

⁵⁷ See Section 5.3.3.

⁵⁸ For new non-residential projects only. If this was a residential project sized 30 kW or greater, 52 full discharges would be required.

Example PBI Payment for a two-hour system:

100 kWh system at \$.50/Wh

Total incentive: 100,000 Wh * \$.50/Wh = \$50,000

Upfront payment: \$50,000 / 2 = \$25,000

Remaining payment to be recuperated through PBI: \$50,000 / 2 = \$25,000

Total anticipated kWh discharged/offset: 100 kWh * 104 full discharges * 5 years = 52,000 kWh

\$/kWh for anticipated kWh discharged/offset: \$25,000 / 52,000 kWh = \$0.480769230/kWh

PBI payment per year assuming 104 full discharges: 10,000 kWh * \$0. 480769230/kWh = \$5,000

For new non-residential projects defined as submitted on or after April 1, 2020, annual PBI payments are subject to PBI payment reductions based on GHG emissions. See section 5.3.1 for more details.

5.4 Application Documentation Requirements for Energy Storage Projects

5.4.1 Required Documentation for Reservation Request

Energy storage applications must provide a copy of the following:

Table 5.4.1: Reservation Request Requirements

Required Materials

- 1. Completed Reservation Request Form (All Projects)
- 2. Application Fee (Non-Residential Projects only)
- 3. Equipment Specifications (All Projects not currently on the SGIP Energy Storage Equipment List)
- 4. Proof of Utility Service/Load Documentation (All Projects)
- 5. Preliminary Monitoring Plan (All Non-Residential Applications and/or Residential projects paired with on-site renewable generators)
- 6. Proof of Equity Budget Eligibility (Equity Projects Only)
 - Certification of Tax-Exempt Status (Gov't and non-profit only)
 - Proof of Eligibility under Proposition 39 or an Accredited Institution (Educational Institutions Only)
 - Small Business Affidavit (Small business projects only)
 - SGIP Multi-Family Low-Income Housing Documentation Cover Sheet (Multi-family only)
 - Copy of Deed-Restriction or Regulatory Agreement (Multi-family only)
 - Proof of Income Qualification (Single-family only)
 - Proof of PU Code Section 2852 Compliance (Single-family only)
 - Copy of SASH or DAC-SASH Reservation Letter
 - Proof of Indian Country Qualification
 - Proof that at least 50 percent of census tracts served are DACs or low-income communities (Public agencies and tribal government agencies only)

Required Materials

- Customer Resiliency Attestation (All projects with a longer than two-hour duration and projects applying for a Resiliency Adder or Equity Resiliency Incentive) Equity Resiliency and Equity Projects >2hrs)
- 8. Proof of Coordination with Local Government and California Office of Emergency Services (Non-Residential Equity Resiliency and Resiliency Adder Applications only)

Two-Step Applications should also include all applicable Proof of Project Milestone Documents (as outlined in Section 5.4.2) as part of their Reservation Request

1. Online Reservation Request Form (All Projects)

All applicants are required to complete the online Reservation Request Form (RRF). This online form is used to provide project details, contact information, and signed declarations. Residential energy storage projects must also fill out the Residential Energy Storage Eligibility Affidavit portion of the RRF.

As part of the RRF, the project Host Customer must attest that an Energy Efficiency Audit (EEA)⁵⁹ has been performed for the project site and they received and reviewed the findings of the audit.

The RRF must be printed and signed by the Applicant, Host Customer and System Owner (if not Host Customer). A copy of the signed document must be uploaded to the online application database and the information contained on the signed copy of the RRF must match the information provided on the online RRF.

2. Application Fee (Non-Residential Projects only)

The application fee is equal to 5% of the requested incentive amount, payable by check, cashier check or money order, and should reference the project by site address.⁶⁰ The application fee is the only required document that must be mailed directly to the Program Administrator.⁶¹ After a project is assigned to an incentive step, the application fee check must be mailed within 7 calendar days. If the check is not mailed within 7 calendar days, the project may be cancelled. Application fee checks returned by the financial institution without payment may result in cancellation of the application.

The application fee will be refunded upon completion and verification of the installed SGIP project. Prior to project completion, application fees are non-refundable once Confirmed Reservation has been issued.⁶² All forfeited application fees will be allocated to the Program Administrator's current incentive step.

⁵⁹ An Energy Efficiency Audit (EEA) report issued within the past five years. A Title 24 energy efficiency compliance report issued within the past three years may also be used in lieu of an Energy Efficiency Audit.

⁶⁰ Application fees are specific to an application, not a site. If the same site reapplies to the program, they will need to submit a new application fee.

⁶¹ A single application fee check for each project is required.

⁶² Application Fees will not be altered due to project changes that may result in a different incentive.

3. <u>Equipment Specifications</u> (All Projects not currently listed on the SGIP Energy Storage Equipment List)

All reservation requests must list equipment that has been vetted by the SGIP Technical Working Group.

Equipment and/or component pairings that have been evaluated by the Technical Working Group can be found on the SGIP Energy Storage Equipment List at www.selfgenca.com. For projects using equipment and/or component pairings not yet evaluated by the Technical Working Group, manufacturer equipment specifications for all major components of the system, such as the storage component, as well as the inverter, DC-DC converter, controller and/or additional system components when applicable, are required. Rated capacity (kW), energy capacity (kWh), and round-trip efficiency⁶³ for the storage system must be provided to a PA prior to submitting a reservation request form for that project Equipment specifications may be submitted to any PA via email to the contact information listed on selfgenca.com.

DC/AC systems must provide all specifications necessary to calculate the rated capacity and energy capacity, such as:

- Duration of discharge (hours)
- DC dischargeable amp-hour capacity, associated with the duration of discharge specified, including all losses and ancillary loads (such as power conditioning and thermal management),
- Nominal voltage (V_{DC}),
- Inverter CEC-AC efficiency
- DC-DC converter efficiency
- Inverter continuous power output rating (kW)
- The continuous maximum power output capability of the system. ⁶⁴ For DC systems, this is rated in DC, and for AC systems, this is rated in AC.

HVAC-integrated S-TES must provide the TES system equipment specifications, HVAC system equipment specifications identifying the SEER and tonnage of the HVAC unit and the climate zone in which the project is located.

Refrigeration TES systems must provide TES system equipment specifications, refrigeration system equipment specifications, the Refrigeration TES kW Worksheet and backup documentation of any site-specific conditions, if relevant.

⁶³ AC-AC for AC systems and DC-DC for DC systems

⁶⁴ Maximum continuous output may be determined by the calculated maximum continuous discharge power of the energy storage system or by the continuous rated power output of the inverter depending on which is less.

Resiliency projects requesting to receive incentives for a system that is sized above peak load necessary due to modular component sizes to accommodate the customer's peak load must demonstrate proof of this need.

4. Proof of Utility Service & Load Documentation (All Projects)

Participation in the SGIP is restricted to customers who are located in PG&E, SCE, SoCal Gas or SDG&E service territories and physically connected to the Electric Utility transmission and distribution system. All applications must include a copy of a recent electric utility bill indicating the account number, meter number, Site address, and Host Customer name. Customers applying in PG&E and SoCalGas territory must also submit a copy of a recent gas utility bill. For new construction, the Host Customer must submit confirmation from the serving utility that their Site is within the Program Administrator's service territory. For projects applying in CSE and SoCalGas territory, applications for technologies that discharge electricity to the onsite load must include a copy of the previous 12-months of electric consumption including maximum demand and kWh consumption to confirm that the participating generation system meets the program sizing requirements. Electric utility customers of SCE and PG&E are not required to submit their 12 month electric consumption and demand data. SDG&E customers are also required to submit an Authorization to Receive Customer Information form, signed by the utility customer of record that authorizes CSE to access utility account information for systems greater than 10 kW or any other instance under CSE discretion if needed.

If the energy storage system is being sized based on new or future load growth (i.e. new construction or load growth due to facility expansion or other load growth circumstances) applications must include an engineering estimate with appropriate substantiation of the Site's annual peak demand forecast. Suggested methods of demonstrating load growth include Application for Service with corresponding equipment schedules and single line diagram; building simulation program reports such as eQUEST, EnergyPlus, EnergyPro, DOE-2, and VisualDOE; or detailed engineering calculations.

For multi-family projects, proof of number of accounts benefiting from the incentivized system must be provided.

Projects interconnecting under a VNEM tariff must submit a VNEM Generation Credit Allocation Request Form to determine the project's primary use case (residential or non-residential). If 50% or greater of VNEM credits are allocated to the common load account, the system's primary use will be considered "common load". If the common meter is on a residential rate, residential GHG and operating rules will apply. If the common meter is on a commercial rate, non-residential GHG rules and operating requirements will apply.

If the primary use case is determined to be residential, all benefiting tenant accounts will be required to be enrolled in an SGIP-eligible TOU rate.⁶⁵

5. <u>Preliminary Monitoring Plan</u> (All Non-Residential Applications and/or Residential Projects Paired with On-Site Renewable Generators)

The preliminary monitoring plan should demonstrate the following components:

Description of the proposed SGIP system:

Description of the system with an overview of the energy services to be provided by the system to the host site; the major components making up the system; and the intended operation of the system (e.g., demand charge management of the facility or specific end-use equipment, TOU energy arbitrage, shifting excess renewable generation, etc.); Include photos and/or diagrams of the system if available.

A description of the existing load at the Site and identification of the sources of the fuel that would be displaced by operation of the SGIP system (i.e., electricity provided by XYZ utility or natural gas provided by ABC utility) and photos of the interface locations where the SGIP system would be located to displace the load.

Description of the metering system and metering approach:

An overview of the performance data to be collected and a simplified layout of the system showing major components and location of the proposed metering points and data to be collected at those points is required.

Description of the approach to be used for collecting, storing and transferring the necessary performance data:

- For example, if electrical energy performance data is to be collected, the reasoning behind the selected metering points
- Frequency with which the data is to be collected (e.g., 15 min intervals)
- Data storage capability and approach for transfer of data (e.g., cell modem)
- For non-residential systems 30 KW and larger, the frequency of reporting to PDP (e.g., daily, weekly) [this could also include frequency for reporting of data to Program Administrators, such as monthly]

⁶⁵ For projects located in front of the customer's electric meter, each customer may be automatically transferred to an SGIP-eligible TOU rate (if one is available to them) by the electric IOU, if one was not selected at interconnection.

Identification of the metering system components by performance data type (including manufacturer and model number)

- Electrical metering equipment (AC meters must meet the criteria for the applicable project type (see section 5.5)
- Thermal energy metering equipment
- Data acquisition (i.e., logger) system

Systems Pairing with On-site Renewable Generators:

Energy storage systems paired with on-site renewable generators must provide a description of:

- The anticipated charge and discharge schedule of the system demonstrating that the system complies with ITC operational requirements or, for projects not claiming the ITC, will be charged at least 75% from renewables;
- The metering that will be used to verify that the system is being charged from renewables;
- The ability to provide data to verify operation in the event of an audit.

6. **Proof of Equity Budget Eligibility** (Equity Projects Only)

Certification of Tax-Exempt Status (Gov't and Non-profit Projects Only)

State and local government and non-profit entities are required to submit verification of their tax-exempt status.

Proof of Eligibility under Proposition 39 or an Accredited Institution (Educational Institutions Only)

For educational institutions, applicants must submit documentation proving eligibility under the California Energy Commission's Proposition 39 Program⁶⁶, or which confirm they are an accredited educational institution.

Small Business Affidavit (Small Business Projects Only)

Application documentation must include a signed affidavit affirming that the host customer is a business or manufacturer, including affiliates, with average annual gross receipts of \$15 million or less over the last three tax years.

Deed Restriction or Regulatory Agreement (Multi-family Projects Only)

For a multi-family residential complex financed with low-income housing tax credits, tax-exempt mortgage revenue bonds, general obligation bonds, or local, state or federal loans or grants, a deed restriction on the

⁶⁶Proposition 39 eligibility guidelines can be found at http://www.energy.ca.gov/efficiency/proposition39/

property, or a regulatory agreement signed and executed between the property owner/developer and entity issuing financing must be submitted.

SGIP Multi-Family Low-Income Housing Documentation Cover Sheet (Multi-family Projects Only)

To clearly outline how a project achieves Public Utilities Code Section 2852 low-income eligibility, applicants are required to submit a cover sheet identifying key details within the deed restriction or regulatory agreement such as, but not limited to, property owner name, public entity/non-profit agency name, expiration date of low income documentation, rent restriction description, and percent of households with incomes at or below 60% of the area median income. This will allow Program Administrators to more accurately and efficiently determine eligibility.

Proof of Income Qualification (Single-family Projects Only)

Documentation showing the host customer's household income is 80% of the area median income or less based upon a copy of the most recently available federal income tax return. Area Median Income is subject to annual changes based upon Housing and Urban Development's income guidelines.

Proof of PU Code Section 2852 Compliance (Single-family Projects Only)

To demonstrate compliance with PU Code section 2852⁶⁷, applicants may submit a copy of a resale restriction or equity sharing agreement for which the homeowner does not receive a greater share of equity than described in paragraph (2) of subdivision (c) of Section 65915 of the Government Code. The resale restriction or equity sharing agreement must be between the homeowner and a public entity or a qualifying non-profit affordable housing provider. Additionally, a presumed resale restriction may also be demonstrated to meet the resale restriction eligibility requirement, such as single-family homes located in IRS-defined Qualified Census Tracts⁶⁸ and other designated areas eligible for the Single-Family Affordable Solar Homes (SASH) program.⁶⁹

Copy of SASH or DAC-SASH Reservation Letter

Customers must provide a copy of their incentive reserved status of the SASH or DAC-SASH program that they are eligible for the program.

Proof of Indian Country Qualification

_

⁶⁷ As described in subparagraph (c) of paragraph (3) of subdivision (a)

⁶⁸Qualified Census tracts are defined by Internal Revenue Code Section 143(j)(2) as "a tract in which 70 percent or more of the families have income which is 80 percent or less of the statewide median family income" https://www.huduser.gov/portal/sadda/sadda_qct.html
⁶⁹ Decision 15-01-027 allows presumed resale restrictions within the SASH program, such as those found in federally-designated Empowerment Zones, Enterprise Communities, certain Neighborhood Revitalization Areas, Targeted Employment Areas, and Qualified Census Tracts.

A Native American tribal card or official document must be provided to be eligible for the expanded DAC requirement.

Proof that at least 50 percent of census tracts served are DACs or low-income communities

Customers have the burden of providing any information that demonstrates a facility's eligibility.

Proof of Coordination with Local Government and California Office of Emergency Services

Non-Residential Customers must demonstrate that they are coordinating with their local government and the California Office of Emergency Services if applying for Equity Resiliency or for a Resiliency Adder Incentive. Customers that cannot demonstrate this, will have their application deprioritized in favor of applicants that can provide proof.

7. <u>Customer Resiliency Attestation (</u>All projects with a longer than two-hour duration and projects applying for a Resiliency Adder or Equity Resiliency Incentive)

Customer attestation including the following information to be provided by the Developer:

- (1) Estimate of how long a project's fully charged battery will provide electricity for the relevant facility average load during an outage
- (2) Indicate whether a project's critical loads can and will be isolated
- (3) Provide an estimate of how long the project's fully charged battery will provide electricity to critical uses during an outage
- (4) Provide an estimate of how long the project can operate in less-than favorable circumstances, such as if an outage occurs when the battery has been discharged or during the winter (if paired with solar)
- (5) Summarize information given to the customer about how the customer may best prepare the storage system to provide backup power, in the case of a PSPS event announced in advance
- (6) Attest to the truth of the information provided; and
- (7) Provide an attestation from the customer indicating that he or she received this information prior to signing a contract with the Developer.

Additional Requirements for Two Step Applications

All 2 step applications must include as part of their Reservation Request materials, all applicable requirements of the Proof of Project Milestone. See Section 5.4.2.

5.4.2 Required Documentation for Proof of Project Milestone

Energy storage applications must provide a copy of the following:

Table 5.4.2: Proof of Project Milestone Requirements

_	Table 5.4.2. I Tool of Froject Milestone Requirements
	Required Materials
	1. Completed Proof of Project Milestone Form (All 3-Step Projects)
	2. Copy of RFP or equivalent after 90 days (Public Entity Projects Only)

Required Materials

- 1. Completed Proof of Project Milestone Form (All 3-Step Projects)
- 3. Copy of Executed Contract or Agreement for Installation (All Projects)
 - Includes Required Warranty Documentation

1. Online Proof of Project Milestone Form (All 3-Step Projects)⁷⁰

The Online Proof of Project Milestone Form must be completed and signed by the Applicant and representatives with signature authority for both the System Owner and Host Customer (if not Host Customer). The online form must identify updated project information including the installation contractor's name, telephone number and contractor license number. All systems must be installed by an appropriately licensed California contractor in accordance with rules and regulations adopted by the State of California Contractors' State Licensing Board (CSLB). Installation contractors must have an active A, B, or C-10 license, or a C-46 license for the combined installation of solar photovoltaics and energy storage.

2. Request for Proposals (RFP) Documentation (Public Entities Only)

Notice to Invite Bids, or similar solicitation issued for the installation, lease and/or purchase for systems proposed for the SGIP. The RFP must include sufficient project details such as the scope of work, schedule, terms, budget, and/or system components desired. For Public Entities not issuing an RFP, alternative documentation such as an executed letter of intent to engage with a contractor on the Host Customer letterhead, an executed contract/agreement for system installation/lease, an equipment purchase order, or alternate system ownership agreement must instead be submitted within 90 calendar days of the date the Conditional Reservation Letter. Proof of Project Milestone documentation must then be submitted within 240 days of the date the Conditional Reservation Letter.

3. Executed Contract and/or Agreement for System Installation (All Projects)

A copy of the executed contract for purchase and installation of the system, and/or alternative System Ownership Agreement (such as a Power Purchase Agreement) is required. The contract/agreement must be legally binding and clearly spell out the terms and scope of work. Purchase and/or installation agreements must also include system equipment and eligible system costs. All contracts/agreements must be signed by appropriate representatives (Host Customer, Installer, and/or System Owner) who are a party to the agreements of the SGIP reservation.

As part of the Executed Contract, all storage systems are required to include a minimum <u>10</u>
 <u>year service warranty</u>. A service warranty ensures proper maintenance and continued project

65

⁷⁰ Not required for 2-Step Applications as part of the Reservation Request Package.

performance. The service warranty must cover the system maintenance to include (but not limited to) system support, problem diagnosis, on-site repair and preventative maintenance. The warranty should also include language to guarantee the continued performance of the system over the warranty period. The System Owner must provide proof of warranty and maintenance contract, and specify the warranty and maintenance contract start and end dates.

5.4.3 Required Documentation for Incentive Claim

Energy storage applications must provide a copy of the following:

Table 5.4.3: Incentive Claim Requirements

Required Materials

- 1. Completed Incentive Claim Form (All Projects)
- 2. Proof of Authorization to Interconnect (Projects that interconnect with the electrical grid)
- Building Permit Inspection Report (All Projects)
 Additional AHJ Grid Islanding Plan (Projects applying for Resiliency adder, Equity Resiliency, and Equity >2hr Projects Only)
- 4. **Substantiations New or Expanded Load (if applicable):** Projects Without Access to Approved TOU Rates (All Residential Projects and VNEM Projects (Proof of GHG Compliance)
- 5. Final Monitoring Schematic (All Projects)

1. Online Incentive Claim Form (All Projects)

The ICF information must be complete, accurate and represent the actual system, project costs and/or fuel information as installed (including system size and type). It must also be signed by the Applicant, Host Customer and System Owner (if not the Host Customer).

2. Proof of Authorization to Interconnect (Projects that interconnect with the electrical grid)

Host Customers and/or System Owners will be required to execute certain documents such as, but not limited to, an "Application to Interconnect a Generating Facility" and a "Generating Facility Interconnection Agreement" with the local Electric Utility. A copy of the signed letter from their Electric Utility granting the Host Customer and/or System Owner permission to interconnect and operate in parallel with the local grid should be submitted as proof of Authorization to Interconnect.

Applicants, Host Customers and System Owners are solely responsible to submit interconnection applications to the appropriate Electric Utility interconnection department as soon as the information to do so is available to prevent any delays in system parallel operation.

3. Building Permit Inspection Report (All Projects)

A copy of the final building inspection report (or proof of exemption) demonstrating that the Project has been inspected by the local Authorities having Jurisdiction (AHJ), meets all codes and standards of the permitting jurisdiction, and has been authorized for operation. Contact your local permitting jurisdiction to learn about permitting requirements.

<u>Projects applying for Resiliency adder, Equity Resiliency Projects, and Equity Projects with longer than two-hour discharge duration</u>

A copy of the final building Inspection Report (or proof of exemption) demonstrating that the Project has been inspected by the local Authority (AHJ), meets all codes and standards of the permitting jurisdiction and has been authorized for operation. Additionally, these projects must also show that an AHJ has approved plans showing that the system can operate in island mode and identifying if the project's critical loads are currently isolated.

4. Substantiations (All Projects)

New Construction or Expanding Load: For Projects where Host Customer estimated the future load to justify system size, applications must include documentation demonstrating that the load forecast has materialized.

Residential Projects Without Access to Approved TOU Rates: Customers that do not have access to an approved TOU rate may submit documentation as part of the incentive claim package that demonstrates the system is compliant with either solar-only charging (defined as a battery that begins charging at dawn and continues charging from the solar unit until fully charged, and that never charges from the grid) or has set relevant storage parameters to the appropriate manufacturer-certified functionalities at the time of installation to enable solar-self-consumption mode (defined as systems charging from solar generation and discharging the stored power only to meet onsite load).

<u>Projects claiming the ITC for equipment purchased after December 31, 2021:</u> Proof of purchase dated after December 31, 2021 for the energy storage equipment and its main components (battery, inverter, controls).

VNEM Projects: Proof that project complies with the applicable residential or non-residential operational and GHG emissions reduction requirements must be provided.

5. Final Monitoring Schematic (All Projects)

The final monitoring schematic is an electrical single line diagram (SLD) that includes the energy storage system, the inverter, the utility meter, the load panel and, when applicable, the PBI meter and on-site renewable generator with which the energy storage system is paired. Documentation must also be provided if there is a change in the make and model of the meters to be used (from what was submitted with the Preliminary Monitoring Plan at the Reservation Request).

For PBI projects, the Applicant must also provide the name of the Performance Data Provider (PDP) they are contracting with. A copy of the contract between the PDP and the Applicant may be requested at the Program Administrator's discretion.

5.5 Metering & Monitoring Requirements for Energy Storage Projects

PBI projects must install metering and monitoring equipment that measures net electrical output or offset from the system(s). Energy storage systems that discharge electricity must measure the net electrical energy during charge and discharge cycles. TES must measure electrical energy when charging. HVAC-integrated S-TES systems must monitor and report the power (kW offset) and energy (kWh offset) that would have been consumed by the HVAC unit to provide the same amount of cooling provided by the S-TES system by monitoring outside air temperature and when the S-TES system turned off the compressor of the HVAC unit. Refrigeration TES systems must report the power (kW offset) and energy (kWh offset) that would have been consumed by the refrigeration system(s) to provide the same amount of cooling provided by the Refrigeration TES system by monitoring the operating set points of the refrigeration system(s), cooling load on the refrigeration system(s), and when the Refrigeration TES system turns off the compressor(s) and condenser(s) of the refrigeration system(s).

System owners must install and maintain metering and monitoring equipment at their own cost and are responsible for the choice and installation of the metering hardware. For new non-residential projects 30 kW and larger, all AC electric meters must be listed on the CEC's list of Eligible System Performance and Revenue Grade Meters to be found on http://www.gosolarcalifornia.ca.gov/equipment/index.html. New non-residential projects less than 30 kW must comply with the meter standards in section 5.5.1.2

It is also the responsibility of the System Owners to contract with a Performance Data Provider (PDP) for a minimum of five years. PDPs will be tasked with recording performance data from the installed metering and monitoring equipment and submit it to the Program Administrators on a monthly basis.

All installed electrical meters must fulfill the following requirements:

- Protocols for the minimum required performance/output data must enable any PDP to communicate
 with the meter and obtain the minimum required performance/output data from the meter and/or
 logger.
- All meters must have the capability to report their data remotely.
- Data reporting must occur on a daily basis.
- All meters must have the ability to retain collected data in the event of a power outage. Meters must have the capability to store 7 days' worth of data.
- All meters must provide the PDP provider or defined list of authorized users with the ability to access and retrieve the minimum required data from the meter using the Meter Communication / Data Transfer Protocols described in Section 7.1.

The meters will be inspected as part of the project inspection process.

The following section details the minimum meter requirements per type of meter installed.

5.5.1 *Minimum Electrical Meter Requirements*

Electrical meters installed on the SGIP project provide data used to assess performance of the system, to analyze impact of the SGIP system on utility distribution systems, the peak system demand of the utility and net GHG emission impacts.

5.5.1.1 Large Non-Residential Projects (30kW and larger)

All systems 30 kW and larger and/or storage projects paired with and charging from on-site renewable generators must be installed with a meter or metering system which allows the System Owner and Program Administrator to determine the amount of net system energy charge and discharge and allows the System Owner to support proper system operation and maintenance.

5.5.1.2 Small Non-Residential Projects (less than 30kW)

All systems less than 30 kW with an on-board inverter-based meter, receiving a PBI incentive must have a meter accurate to within \pm 3% of actual system output.

 \pm 3% AC on-board inverter electrical meters required for PBI must be tested according to all applicable ANSI C-12 testing protocols pertaining to the monitoring of power (kW) and energy (kWh). The accuracy rating of \pm 3% meters must be certified by an independent testing body (i.e., a NRTL such as UL or TUV). DC electrical meters must also have a \pm 3% accuracy pertaining to the monitoring of power (kW) and energy (kWh), as specified by the DC-DC converter manufacturer.

Meter Type

All systems are allowed to use on-board electrical meters, however, the meter must meet the minimum meter requirements for the applicable system size. For all systems receiving PBI payments, the installed meter(s) may be a separate Interval Data Recording (IDR) meter(s), or a complete onboard system that is functionally equivalent to an IDR meter, recording data no less frequently than every 15 minutes. Program Administrators may have additional meter functionality requirements for systems receiving PBI payments, as the Program Administrators will use these meters to process PBI payments, and system compatibility may be required. For example, meters and service panels must meet all local building codes and utility codes. The meter serial number must be visible after installation.

Acceptable Electrical Metering Points

For AC energy storage systems, a meter(s) must be installed on the AC side(s) of the energy storage device and account for power delivery to all parasitic loads including thermal management and power conditioning. For DC electrical energy storage systems, a meter(s) must be installed at a point within the electrical system to measure the charge and discharge of the energy storage device and account

for power delivery to all parasitic loads, including thermal management and power conditioning (such as DC to DC conversion). For TES systems, one or more meters must be installed to directly meter or measure the data points necessary to calculate the rate and quantity of charge (in kW/kWh) and the rate and quantity of discharge (in kW/kWh offset).

Meter Accuracy

All systems receiving a PBI incentive must install a meter accurate to within ± 2% of actual system output. This applies to on-board electrical meters as well as external IDR meters.

Meter Measurement and Time Granularity of Acquired Data

Electric meters must measure the net energy charged and discharged (kWh) and net real power charged and discharged (kW)⁷¹. The PDP must log all required energy storage performance / output data points no less frequently than once every 15 minutes. The PDP must measure 15 minute energy and real power for the energy storage system during charging and discharging and account for all ancillary loads. The meter needs to generate an accurate time/date stamp.

Meter Testing and Certification

 \pm 2% AC electrical meters required for PBI must be tested according to all applicable ANSI C-12 testing protocols pertaining to the monitoring of power (kW) and energy (kWh). The accuracy rating of \pm 2% meters must be certified by an independent testing body (i.e., a NRTL such as UL or TUV). DC electrical meters must also have a \pm 2% accuracy pertaining to the monitoring of power (kW) and energy (kWh), as specified by the meter manufacturer.

Meter Display

All meters must provide a display capable of showing the measured net charge and discharge energy output and measured instantaneous power during charging and discharging. This display must be easy to view and understand and must be physically located either on the meter or on a remote device. For PBI, if a remote device is the only visible access, the PA may ask for verification.

⁷¹ For AC systems, measured in AC. For DC systems, measured in DC

5.6 San Joaquin Valley Affordable Energy Pilot Projects, Budget, Incentive Design, and Eligibility

\$10 million in SGIP incentives is available for the San Joaquin Valley (SJV) pilot host communities within PG&E's and SCE's service territories.⁷² Of the \$10 million set-aside, \$9.76 million is allocated for residential system incentives, and \$240,000⁷³ is allocated for non-residential system incentives.⁷⁴.

Eligible SJV customers are located in Allensworth, Alpaugh, Cantua Creek, Ducor, Fairmead, Lanare, Le Grand, La Vina, Seville, West Goshen, and California City.

Eligibility for the approved residential SJV pilot incentives is limited to households that participate in the pilot by replacing one or more propane, wood-burning, or inefficient or inoperable major electric appliances with the efficient electric appliances offered by the pilot. Eligibility for the approved non-residential SJV pilot incentives is limited to nonresidential customers providing critical facilities or infrastructure as outlined in Handbook Section 4.1.1.3.

All participating SJV pilot customers must to adhere to the GHG reduction requirements. CARE-eligible SJV pilot households that wish to access SGIP incentives must enroll in an SGIP-approved rate, if one is available, or in any CARE TOU rate if an SGIP-approved rate is not available, regardless of the date of submittal of the SGIP application.

The incentive level for SJV pilot communities is \$1.00/Wh.

Approved project incentives must not exceed actual project costs.

SJV Budget Allocations by Program Administrator

PA	SJV Residential	SJV Non-Residential		Total	
PG&E	\$ 4,880,000.00	\$	120,000.00	\$ 5,000,000.00	
SCE	\$ 4,880,000.00	\$	120,000.00	\$ 5,000,000.00	
Total	\$ 9,760,000.00	\$	240,000.00	\$ 10,000,000.00	

⁷³ If applications for non-residential projects exceed \$240,000 within four years of issuance of this decision, then PGE and SCE may allocate an additional \$240,000 towards SJV pilot non-residential system incentives.

⁷² D.19-09-027, p. 78.

⁷⁴ Unreserved funds remaining in the SJV set-aside after four years shall be reallocated to the residential equity budget.

Vendors / developers participating in SJV pilot projects shall not sell a residential storage system that receives incentives for a total price (before incentives) that is greater than the price they sell a comparable system that does not receive incentives.

6 Generation Technologies

The following sections outline operational eligibility, system size, fuel blending, incentive calculation, application documentation requirements and metering requirements specific to SGIP generation projects. Additionally, all projects are subject to the general program requirements as outlined in sections 2, 3 and 4.

6.1 Operational Eligibility Requirements for Projects Operating on Blended Fuel

The following section describes the operational eligibility requirements for participating generation technologies using any amount of fossil fuel.

6.1.1 Minimum Operating Efficiency Requirements

All generation technologies using blended fuel must meet or exceed a minimum operating efficiency requirement without the inclusion of renewable fuel. The systems can satisfy this requirement by either meeting:

- 1. Waste heat utilization requirements **OR**
- 2. Minimum electric efficiency

To facilitate minimum operating efficiency requirements and determine system eligibility, a Minimum Operating Efficiency Worksheet is available for download from the Program Administrators' websites.

Waste Heat Utilization

To meet minimum waste heat utilization, CHP systems must meet the requirements of Public Utilities Code 216.6, which are expressed in the following equations:⁷⁵

P.U. Code 216.6 (a) =>
$$T/(T + E) \ge 5\%$$

And,

P.U. Code 216.6 (b) =>
$$(E + 0.5 \times T) / F_{LHV} \ge 42.5\%$$

Where:

T ≡ the *annual* useful thermal output used for industrial or commercial process (net of any heat contained in condensate return and/or makeup water), heating applications (e.g., space heating, domestic hot water heating), used in a space cooling application (i.e., thermal energy used by an absorption chiller).

⁷⁵ PUC 216.6 - "Cogeneration" means the sequential use of energy for the production of electrical and useful thermal energy. The sequence can be thermal use followed by power production or the reverse, subject to the following standards: (a) At least 5 percent of the facility's total annual energy output shall be in the form of useful thermal energy; (b) Where useful thermal energy follows power production, the useful annual power output plus one-half the useful annual thermal energy output equals not less than 42.5 percent of any natural gas and oil energy input.

E ≡ the **annual** electric energy made available for use, produced by the generator, exclusive of any such energy used in the power production process.

F_{LHV} ≡ the generating system's *annual* Lower Heating Value (LHV) non-renewable fuel consumption.

Minimum Electric Efficiency⁷⁶

To meet the minimum electric efficiency criteria, the proposed generator's electrical efficiency must be equal or greater than 40%, which is expressed in the following equation:

Electrical Efficiency = E / F_{HHV} ≥ 40%

Where:

E ≡ the generating system's rated electric capacity as defined in *Section 6.1.5*, converted into equivalent Btu/hr using the factor 3,414 Btu/kWh.

F_{HHV} ≡ the generating system's Higher Heating Value (HHV) fuel consumption rate (Btu/hr) at rated capacity.

6.1.2 NOx Emission & Minimum System Efficiency Standards

Systems using blended fuel must not exceed a NOx emissions standard of 0.07 lbs/MW-hr and/or must meet the 60% minimum system efficiency requirement.⁷⁷

The minimum system efficiency shall be measured as useful energy output divided by fuel input in higher heating value. The calculated minimum system efficiency shall be based on 100 percent load. The following formula is to be used to determine the system efficiency:

CHP System Efficiency = (E + T) / F_{HHV} ≥ 60%

Electric only System Efficiency = E/ F_{HHV} ≥ 40%

Where:

E ≡ the generating system's rated electric capacity as defined in *Section 6.1.5*, converted into equivalent Btu/hr using the factor 3,414 Btu/kWh

T ≡ the generating system's useful waste heat recovery rate (Btu/hr) at rated capacity.

F_{HHV} ≡ the generating system's Higher Heating Value (HHV) fuel consumption rate (Btu/hr) at rated capacity.

⁷⁶ This requirement was included as an alternative requirement to meeting Public Utilities Code 216.6 in compliance with AB 2778.

⁷⁷ An emission credit for waste heat utilization may be used to adjust the final emissions determination of eligibility.

CHP systems using blended fuels that fail to meet the NOx emission standard but meet the 60% minimum system efficiency standard may be eligible to receive an emission credit for waste heat utilization.

CHP systems operating solely on waste gas are exempt from the NOx emission requirements if the local air quality management district or air pollution control district, in issuing a Permit to Operate for the project, provides in writing a determination that the operation of the project will produce an on-site net air emissions benefit compared to permitted on-site emissions if the project does not operate. Waste gas systems, though exempt from NOx emission requirements, still must meet the minimum operating efficiency requirement.

Meets Minimum Operating Efficiency?

YES

NO

Not qualified

Meets 60% System Efficiency?

YES

NO

Not qualified

Meets 0.07 lb. NOX/MWh w/o credit?

NO

YES

Qualifies

Meets 0.07lb. NOX/MWh w/ credit?

Figure 6.2.2: CHP System Efficiency and NOx Emissions Eligibility

NO

Not qualified

6.1.3 Greenhouse Gas Emission Standards

The GHG eligibility factor is established based on the year in which the application is accepted. GHG eligibility for electric only systems is based on the First-Year Average factor. GHG eligibility for CHP systems is based on the ten-year average factor. The ten-year average and first-year factors for years 2016 through 2020 are listed below.

YES

Qualifies

Application Year	2016	2017	2018	2019	2020
10-Year Average	350	347	344	340	337
First-Year Average	334	332	329	325	321

Table 6.1.3: SGIP GHG Eligibility Emissions Factors, kgCO₂/MWh

Greenhouse Gas Emission Standards for CHP Systems

CHP systems operating on blended fuel must emit GHG emissions at a rate equal or lower than the applicable ten-year average factor. The gross GHG output is calculated by multiplying the annual fuel consumption of the generator in MMBtus by an emission factor of 53.02 kg CO₂/MMBtu⁷⁸ for the conversion of natural gas to CO₂. The GHG savings from waste heat recovery are calculated by dividing the annual waste heat recovered in MMBtus by 80% which represents nominal boiler efficiency and then multiplying by the 53.02 kg CO₂/MMBtu emission factor. The net GHG output of the generator is calculated by subtracting the GHG savings due to waste heat recovery from the gross GHG output. The GHG emissions rate for the generator is found by dividing the net annual GHG emissions by the annual electrical output of the generator in MWh and averaged over the years in operation.

Greenhouse Gas Emissions Standards for Electric-Only Systems

Electric-only systems operating on blended fuel must demonstrate they will emit GHG emissions at a rate lower than the applicable annual average over ten years of operations, accounting for performance degradation, in order to receive SGIP incentives. For example, 2017 ten-year average (347 kg CO₂/MWh) is equivalent to a first-year emissions rate of 332 kg CO₂/MWh.

The ASME PTC 50-2002 will be used to determine the system's first year electrical efficiency and first year emission rate. The ten-year average can be verified through performance warranties, contractual requirements, or other supporting documentation. Alternatively, the ten-year cumulative average net power of the fuel cell coupled with the fuel input rate (HHV) can be used to calculate the annual power generation (MWh) and fuel consumption (MMBtu) based upon an assumed capacity factor of 80%. The GHG output is calculated by multiplying the annual fuel consumption of the fuel cell in MMBtus by the emission factor of 53.02 kg CO2/MMBtu for the conversion of natural gas to CO2. The GHG emissions rate for the generator is found by dividing the annual GHG emissions by the annual electrical output of the generator in MWh.

6.1.4 Reliability Criteria

CHP systems operating on blended fuel must meet both of the following reliability requirements:

- 1. The self-generating facility must be designed to operate in power factor mode such that the generator operates between 0.95 power factor lagging and 0.90 power factor leading. This design feature will be verified by reviewing the manufacturer's specifications at the time of application and as part of the field verification visit before incentive payment approval.
- System owners with facilities sized greater than 200 kW must coordinate the self-generation facility planned maintenance schedule with the electric utility. This allows the utility to more accurately schedule load and plan distribution system maintenance. The system owner will only schedule a

⁷⁸ Unspecified natural gas conversion emission factor of Section 95112 of the mandatory GHG reporting regulation. Title 17 of the California Code of Regulations.

facility's planned maintenance between October and March and, if necessary, during off-peak hours and/or weekends during the months of April to September.

6.1.5 Rating Criteria for System Output

The generating system rated capacity is the net continuous power output of the packaged prime mover/generator under the conditions defined below for each technology. In order to determine the net continuous power output, all ancillary loads must be subtracted from the gross output of the generator. Ancillary loads are defined as equipment loads, added as part of the SGIP generator project, necessary for the operation of the generator (e.g. fuel compressors, intercooler chillers, pumps associated with waste heat recovery, blowers used to transport renewable fuel, fuel clean-up equipment). System capacity ratings are established at the time of Conditional Reservation Notification in order to determine the SGIP reservation dollar amount. If system modifications (i.e. changes in equipment make/model) are made after the Conditional Reservation Letter is sent, the system capacity must be re-rated using currently available published component information for the changed equipment. If the number of components has increased or decreased and there is no change in the make/model of the equipment used, system components can be re-rated using the same published information used at the time of the Conditional Reservation.

- Steam turbine CHP rated capacity is the net continuous power output of the packaged prime mover/generator at the average pressure and temperature of the steam produced by a boiler operating on blended fuel.
- For on-site biogas projects, the generating system capacity is the operating capacity based on the average annual available renewable fuel flow rate, including allowable non-renewable fuel at ISO conditions.⁷⁹
- For directed biogas projects, the generating system rated capacity is the net continuous power output of the packaged prime mover/generator at ISO conditions operating on a non-renewable fuel.
- Pressure reduction turbine technologies rated capacity is based upon the average pressure drop across and flow through the turbine, when flow exists, as determined by historical flow and pressure data from the previous year, if available, or from an engineering estimate if new construction or expanded load. Eligible technology system rated capacity must be substantiated with documentation from the manufacturer.

6.2 Capacity Factors

Generation systems are expected to operate at the following capacity factors:

⁷⁹ Industry standard conditions to measure output – temperature at 59 degrees Fahrenheit and altitude at sea level (0 feet).

Table 6.2: Assumed Capacity Factors

Technology Type	Capacity Factor
Wind turbine	25%
All other generation	80%

6.3 Operational Eligibility Requirements for Renewable Technologies and Generation Projects Operating on 100% Renewable Fuel

Generation systems operating on 100% renewable fuel are not subject to the operational requirements above. However, they are subject to all rating and sizing criteria below.

6.3.1 Rating Criteria for System Output of Renewable Technologies

The generating system rated capacity is the net continuous power output of the packaged prime mover/generator under the conditions defined below for each technology. In order to determine the net continuous power output, all ancillary loads must be subtracted from the gross output of the generator. Ancillary loads are defined as equipment loads, added as part of the SGIP generator project, necessary for the operation of the generator (e.g. fuel compressors, intercooler chillers, pumps associated with waste heat recovery, blowers used to transport renewable fuel, fuel clean-up equipment)

- Wind turbine technologies, less than 30 kW in capacity, a minimum hub height of 80 feet is required. No height limitation is imposed for turbines equal to or larger than 30 kW. For wind turbines of all sizes the wind turbine's rated capacity is based upon the highest electrical output from the manufacturer's power output curve for wind speeds up to 30 mph including inverter losses.
- Waste heat to power technologies rated capacity is based on the average waste heat production
 rate and temperature, when waste heat is available, as determined by historical waste heat and
 temperature data from the previous year, if available, or from an engineering estimate if new
 construction or expanded load.
- Pressure reduction turbine technologies rated capacity is based upon the average pressure drop
 across and flow through the turbine, when flow exists, as determined by historical flow and pressure
 data from the previous year, if available, or from an engineering estimate if new construction or
 expanded load. Eligible technology system rated capacity must be substantiated with
 documentation from the manufacturer.

6.4 Sizing Requirements for all Generation Systems

The following system sizing requirements are applicable for systems that are rated above 5 kW.

6.4.1 System Sizing for Wind Turbines

Host customers with a previous 12-month annual peak demand that is less than 333 kW may size wind turbine projects up to 200% of the annual peak demand at the proposed site. If the Host customer's annual peak demand is greater than or equal to 333 kW, wind turbine projects may be sized up to 300% of the peak demand at the proposed site. Sites hosting existing generation, must also meet these sizing limits including both the capacity of the proposed wind turbine and the capacity of any existing generators (excluding any backup generators).

6.4.2 System Sizing for PRT, Waste Heat to Power, CHP and Fuel Cells

Pressure reduction turbine, waste heat to power, steam turbine, gas turbine, microturbine, internal combustion engine and fuel cell projects may be sized up to the host customer's previous 12-month annual peak demand at the proposed site.

If the site hosts existing generation, the combined capacity of the proposed and existing generators (excluding any backup generators) must be no more than the host customer's maximum site electric load.

In order to reduce GHG emissions and optimize system efficiency, CHP systems using blended fuel must not exceed the on-site thermal load with the recovered waste heat on an annual basis.

6.4.3 System Sizing for Projects Exporting Power to the Grid

Systems that will be exporting power to the grid will size their generators based upon 125% of the last twelve months of electrical consumption (kWhs) at the site. The incentivized capacity of the generator will be based upon 100% of the last twelve months of electrical consumption at the site. The incentivized capacity will be determined by dividing the annual electrical consumption at the site (in kWh) by 8760 hours and the expected capacity factor of the technology as stated in *Table 6.3*.

6.4.4 System Sizing for RES-BCT Customers

Any local governments participating in the RES-BCT tariff (AB 2466) or any customer participating in FC-NEM (fuel cell customers who have been determined by the CPUC to achieve reductions in emissions of greenhouse gases pursuant to subdivision (b) of PU Code § 2827.10, and meets the emissions requirements for eligibility for funding set forth in subdivision (c) of PU Code Section 379.6, and has commenced operation on or before January 1, 2015) may size their systems to the total annual electrical load at the site where the generating system is located and the benefiting site(s) combined. However, they are only eligible for incentives up to the total annual electrical load (kWh) at the site where the generating system is located. Local government sites participating in the RES-BCT tariff must comply with the 5 MW cap per site.

6.4.5 System Sizing Limitations - Ineligible Host Customer Loads

The following loads cannot be considered when sizing a system:

- Customers who have entered into contracts for distributed generation services (e.g. distributed generation installed as a distribution upgrade or replacement deferral) and who are receiving payment for those services. This does not include Power Purchase Agreements, which are allowed.
- Any portion of a Host Customer's load that is committed to electric utility interruptible, curtailable rate schedules, programs or any other state agency-sponsored interruptible, curtailable or demand-response programs. For electric utility customers who are on an interruptible rate, only the portion of their electric load designated as firm service is eligible for the SGIP. Customers must agree to maintain the firm service level at or above capacity of the proposed generating system for the duration of the applicable warranty period. Customers may submit a letter requesting an exemption to the firm service rule if they plan to terminate or reduce a portion of their interruptible load. Wind and energy storage projects need not abide by this portion.
- Publicly-owned or investor-owned gas, electricity distribution utilities or any electrical corporation (ref. Public Utility Code 218) that generates or purchases electricity or natural gas for wholesale or retail sales.

6.5 Eligible Fuel Requirements

Eligible fuels are those that produce useful energy when undergoing combustion or a chemical reaction and are classified as renewable, non-renewable, or blended.⁸⁰

6.5.1 Renewable Fuel Blending Requirements

All gas generation technologies are required to blend a minimum amount of renewable fuel beginning in 2017. The minimum percentage is determined by the year in which the application is accepted. The table below incorporates the minimum renewable fuel blending requirements by application year.

Table 6.5.1: Minimum Renewable Fuel Blending Requirement

Application Year	% Renewable Fuel
	Required
2016	0%
2017	10%
2018	25%
2019	50%
2020	100%

6.5.2 Directed Biogas Project Requirements

Directed biogas projects must meet the following eligibility requirements and conditions:

⁸⁰ For the purposes of SGIP, renewable resources such as wind, pressure, waste heat, and water are not categorized as a fuel.

Directed renewable fuel must be injected into a common carrier pipeline system that is either
within the Western Electricity Coordinating Council (WECC) region or interconnected to a
common carrier pipeline system located within the WECC region.

6.5.3 Directed Biogas Renewable Fuel Audits

Program Administrators or a third-party designee will conduct an annual audit of the renewable fuel invoices for ten years after the renewable fuel contract commences to verify renewable fuel consumption.

The audit and verification will use metered fuel consumption data for SGIP directed biogas projects; invoices from SGIP participants for directed renewable fuel purchases; documentation/verification on any deliveries of directed renewable fuel along the path from the SGIP participant back to the original directed renewable fuel supply source; and documentation to verify the amount and energy content of directed renewable fuel injected into the path. To complete the audit, the Program Administrator or a third-party designee will request all pertinent information from the system owner and/or renewable fuel provider at the completion of each year after renewable fuel contract start date.

If invoices show that nominated renewable fuel deliveries fell below the generator's committed renewable fuel demand during any 1-year period, the SGIP Program Administrators may request that the system owner refunds the appropriate renewable fuel SGIP incentive amount and reserve the right to request a refund of additional costs associated with the administrative and legal fees incurred by the Program Administrators.

6.5.4 Renewable Fuel Commitment Modifications

Customers are required to meet their renewable fuel commitment throughout the permanency period. In the event that modifications are made applicants are required to notify their Administrator.

- Modifications Pre-ICF applicants must notify their administrator as soon as practical but no later than ICF. Increase of renewable fuel percent prior to the completion of the project may not automatically increase incentive amount.
- Modifications Post-ICF applicants must notify their Administrator within 90 days of any changes made to the contract and amount of renewable fuel. Increase in amount of renewable fuel may not result in an incentive increase. Additionally, the host customer must provide to the Program Administrator with all pertinent documentation.

6.5.5 Pressure Reduction Turbine Requirements

Pressure reduction turbines operating in a steam system the boiler fuel used to generate the steam may be blended or 100% renewable, and as such would be eligible for the renewable fuel adder. The source of the renewable fuel may be on-site or directed. The temperature and flow rate of steam to the pressure reduction turbine will be used along with the boiler efficiency to determine the amount of fuel used to

generate steam supply to the turbine. The adder would apply only to the renewable fraction of the boiler fuel necessary to operate the pressure reduction turbine.

6.6 Incentive Calculation for Generation Projects

Incentives for a proposed generation system are calculated by multiplying the rated capacity (W) of the system⁸¹ by the incentive rate for the appropriate technology type and step.

Incentive = rated capacity * incentive rate

Incentives for 100% renewable and blended fuel projects are calculated by multiplying the rated capacity of the system by the technology incentive rate, plus the rated capacity of the system, multiplied by the percentage of renewable fuel above the minimum requirement, multiplied by the renewable fuel adder rate (\$.60/watt).

Incentive = (rated capacity * incentive rate) + (rated capacity * % above min RN Fuel * RN incentive)

6.6.1 Incentive Declines Based on Generation Capacity

SGIP incentives are paid for up to 3 MW of capacity with tiered incentive rates. For projects that are greater than 1 MW, the incentives identified in Table 3.1.1 decline according to the following schedule:

 Capacity
 Incentive Rate (Pct. of Base)

 0 - 1 MW
 100%

 >1 MW - 2 MW
 75%

 >2 MW - 3 MW
 50%

Table 6.6.1: Tiered Incentive Rates

6.7 Performance-Based Incentive Payment (PBI)

On-site biogas, wind, waste heat to power and pressure reduction turbine projects 30 kW and larger will be paid 50% of the full incentive amount upon project completion and Post-Installation Inspection. The remaining 50% of the incentive will be paid annually over five years. Annual payments will be structured so that based upon the expected capacity factor and renewable fuel commitment (if applicable) a project would receive the entire stream of performance payments in five years.

⁸¹ For more information on rating criteria for system output, see Section 6.1.5.

\$/kWh = remaining 50% of incentive / total anticipated kWh production

Total anticipated kWh production = rated capacity * anticipated capacity factor * hours per year * five years

For a 5-year period the PBI payment will be paid annually based on recorded kWh of electricity produced or offset over the previous 12 months.

PBI Payment = \$/kWh * actual annual kWh

Directed biogas projects 30 kW and larger will be paid 50% of the technology incentive upon project completion and Post-Installation inspection. The remaining 50% of the technology incentive will be paid annually over five years. Annual kilowatt hour-based technology payments will be structured so that under the expected capacity factor, a project would receive the entire stream of performance payments in five years. Additionally, the renewable fuel adder will be paid annually over five years. Annual renewable fuel adder payments will be structured so that, under the expected fuel consumption, a project would receive the entire renewable fuel adder for which it was approved.

\$/kWh = remaining 50% of technology incentive / total anticipated kWh production over 5 years

Total anticipated kWh production = rated capacity * anticipated capacity factor * hours per year * five years

For a 5-year period the PBI payment will be paid annually based on recorded kWh of electricity produced over the previous 12 months.

PLUS

Renewable Incentive Annual Payment = ((rated capacity* % above min RN fuel * RN adder rate) / 5) * (actual capacity factor / anticipated capacity factor)

Both on-site and directed biogas projects will receive a prorated PBI payment for the percentage of renewable fuel that is actually consumed.

6.7.1 PBI Payments for Export to the Grid Projects

SGIP projects that qualify for the feed-in tariff are allowed to export a percentage of their output to the grid. Once on-site electric load has been met, excess generation of electricity may be exported to the grid. The amount exported to the grid is not to exceed 25% of on-site consumption on an annual basis.

In cases where a customer is exporting electricity to the grid, the PBI payment will be calculated based on generated electricity consumed on-site as opposed to the generating system's output.

Based on this description and the \$/kWh calculated during the incentive claim step of the project, the calculation of a PBI payment is as follows:

PBI = \$/kWh * generated electricity consumed on-site

Program Administrators must be informed of arrangements made with the utility for sale of excess generation. For verification purposes, proof of export documentation may be required prior to payment.

6.8 Renewable Fuel Annual Payment Requirements

The following outlines the data and verification requirements for directed and on-site renewable fuel annual payments of systems 30kW and larger.

6.8.1 Directed Renewable Fuel Verification

The amount of directed renewable fuel procured will be verified prior to issuing annual fuel payments. For Directed Renewable Fuel users, applicant or Performance Data Provider (PDP) must provide monthly data of the amount of directed renewable fuel consumed on a monthly basis. This information must be provided through the SGIP database as part of the performance data requirements. Additionally, the applicant or PDP will be required to provide the source, path, and destination of the renewable fuel. Data reporting will commence one month after the initial payment. The following information and documentation must be provided on a monthly basis:

1. Transportation Path and Energy Accounting

The PDP must upload supporting documentation (i.e. invoices) reporting the amount of renewable fuel that was documented on the Directed Renewable Fuel Verification Form. Supporting documentation includes but not limited to:

- Documentation from the source showing the amount of directed biogas being moved onto the pipeline. Any non-renewable gas added at the source must be identified.
- o Documentation from the gas transmission system showing:
- Receipt of directed biogas (from source, storage, or other pipelines)
- Pipeline losses or fees paid in gas (not carried over)
- Positive or negative imbalances (carried over)
- Delivery of directed biogas to either another pipeline, storage facility, or California utility receipt point

2. Gas Fuel Consumption

The PDP must provide gas fuel consumption documentation from the gas utility matching the directed renewable fuel receipts reporting the metered total energy input to the generator. Utility gas fuel consumption receipts must be reported on a quarterly basis.

It is the responsibility of the PDP to supply the renewable fuel use documentation and to ensure that the renewable fuel is reported to the Program Administrator or their designee on a monthly and quarterly basis for five years.

6.8.2 On-site Renewable Fuel Verification

All on-site renewable fuel projects 30 kW and larger are required to install a fuel metering system that reports both renewable and non-renewable fuel. The PDP is required submit both renewable and non-renewable fuel data on a monthly basis.

The System Owner must provide make, model, specifications, and serial number of installed revenue grade gas meters

It is the responsibility of the system owner to contract with a performance data provider (PDP) for a minimum of five years and ensure that both renewable and non-renewable fuel data is provided to the Program Administrator or their designee monthly for five years.

6.9 Incentive Limitations for Projects using Renewable Fuel

The following sections outline the incentive limitations for projects using renewable fuel. Additionally, all generation projects are subject to the general incentive limitations as outlined in *Section 3.2*.

6.9.1 Limitations on PBI based on GHG Emissions Reductions

PBI payments will be reduced or eliminated in years that do not result in the required GHG emissions reductions. Because many factors may lead to a project performing below expected levels of efficiency, there is a 5% exceedance band before penalties are assessed.⁸² The following example describes how a PBI payment for a 2017 project would be affected:

- PBI payments will be reduced by half in years where a project's cumulative emission rate is equal
 to or greater than 364 kg CO₂/MWh (i.e., 5% higher than 347 kg CO₂/MWh) but less than 382 kg
 CO₂/MWh (i.e., 10% higher than 347 kg CO₂/MWh).
- Projects that equal or exceed a cumulative emissions rate of 382 kg CO₂/MWh will receive no PBI payments for that year.

$$emission\ rate < 364\ \frac{kg\ CO_2}{MWh}
ightarrow No\ penalty\ assessed\ on\ PBI\ payment$$

$$364\ \frac{kg\ CO_2}{MWh} \leq emission\ rate < 382\ \frac{kg\ CO_2}{MWh}
ightarrow PBI\ payment\ reduced\ by\ 50\%$$

$$emission\ rate \geq 382\ \frac{kg\ CO_2}{MWh}
ightarrow No\ PBI\ payment\ for\ that\ year$$

The table below illustrates the eligibility emission factors for blended fuel generation projects for the years 2017 through 2020:

Table 6.9.1: Eligibility Emission Factors for Blended Fuel Generation Projects

-

⁸² D.11-09-015, §4.3.2. pg. 32

Program Year	Eligibility Emission Factor	Year 1 to 5
	Baseline	347
2017	5% Exceedance	364
	10% Exceedance	382
	Baseline	344
2018	5% Exceedance	361
	10% Exceedance	378
	Baseline	340
2019	5% Exceedance	357
	10% Exceedance	374
	Baseline	337
2020	5% Exceedance	354
	10% Exceedance	371

6.9.2 Limitations on PBI Adjustments based on Renewable Fuel Verification

All gas generation technologies must meet minimum renewable fuel requirements to be eligible for incentives. However, the annual renewable fuel PBI payment will be adjusted according to the verified consumption. PBI projects whose annual consumption fails to meet the minimum renewable fuel requirement, will forfeit the annual renewable fuel adder payment.

6.9.3 Incentive Limit for the Renewable Fuel Adder

In the case of directed projects, the adder is compared to the cost of the renewable fuel contract and should not exceed the cost difference between the renewable fuel contract and a similar contract for standard natural gas. Projects utilizing 100% on-site renewable fuel will receive the full renewable fuel adder. Projects utilizing blended fuel, where renewable fuel is either on-site or directed, the incentive will be prorated to the percentage of fuel that is actually consumed based on audits which are conducted throughout the PBI period.

6.9.4 Blended Fuel Generating Systems Converted to 100% Renewable Fuel

Blended SGIP-funded generating systems can be converted to 100% renewable fuel and receive the additional renewable fuel adder if the conversion takes place no later than 1 year from the first SGIP incentive payment. However, these conversions are only eligible to receive the additional renewable fuel adder; and all project costs caps are still applicable.

For systems under 30kW the renewable fuel adder will be paid upon completion of conversion. For systems 30kW and larger, 50% will be paid upon completion, and the remaining 50% will be included in the annual PBI payments. The recalculated PBI incentive payments will be based on the following calculation:

PBI Rate (\$/kWh) = (remaining Incentive (\$) + ½ RN fuel adder x Rated Capacity (kW) x (1000W/kW)) / (rated capacity of the generator (kW) * 8760 (hrs/year) * capacity factor * number of years payments will be made)

6.9.4.1 Renewable Fuel Conversion Reservation Request

All renewable conversion reservation requests will follow a 2-step process and must include the following applicable documents (see Section 6.10.1 & Section 6.10.2 for document details):

- 1. Reservation Request Form Application Fee
- 2. Proof of Adequate Renewable Fuel

6.9.4.2 Renewable Fuel Conversion Incentive Claim

Once the conversion has been completed, applicants are required to submit an Incentive Claim Form with the following applicable documents (see Section 6.10.3 for document details):

- 1. Final Permits
- Substantiations:
 - a. Renewable fuel (on-site renewable fuel)
 - b. Fuel Cleanup Skid Cost Documentation (on-site renewable fuel only)
 - c. Renewable Fuel Documentation & Contract Commencement (directed renewable fuel only)
 - d. Renewable fuel metering specifications
- 3. Final Monitoring Schematic (for projects that are 30 kW or larger) to include the name of the Performance Data Provider (PDP).

6.10 Application Documentation Requirements for Generation Projects

Applications are required to meet all documentation requirements on time. Documents are based on the application timeline. The following sections describe the required documents based on the specific step.

Required Documentation for Reservation Request

Generation applications must provide a copy of the following:

Table 6.10.1. December Decuest Dequirements

	Required Materials
1.	Completed Reservation Request Form (All Projects)
2.	Application Fee (Non-Residential Projects only)
3.	Equipment Specifications (All Projects)
4.	Proof of Utility Service/ Load Documentation (All Projects)

Required Materials

- 5. Preliminary Monitoring Plan (All 3-Step Applications >=30 kW)
- 6. Minimum Operating Efficiency Worksheet w/Backup Documentation (Blended Projects Only)
- 7. Proof of Adequate Fuel or Waste Energy Resource (Renewable Fuel, Waste Energy, Waste Gas Projects Only)
- 8. Customer Resiliency Attestation (Resiliency Projects)
- 9. Proof of Coordination with Local Government and California Office of Emergency Services (Non-Residential Renewable Generation Resiliency Adder Applications only)

Two-Step Applications should also include all applicable Proof of Project Milestone Documents (as outlined in Section 6.10.2) as part of their Reservation Request

1. Online Reservation Request Form (All Projects)

All applicants are required to complete the online Reservation Request Form (RRF). This online form is used to provide project details, contact information, and signed declarations.

As part of the RRF the project Host Customer must attest that an Energy Efficiency Audit (EEA)⁸³ has been performed for the project site and they received and reviewed the findings of the audit.

The RRF must be printed and signed by the applicant, host customer and system owner (if not host customer). A copy of the signed document must be uploaded to the online application database and the information contained on the signed copy of the RRF must match the information provided on the online RRF.

2. Application Fee (Non-Residential Projects only)

The application fee is equal to 5% of the requested incentive amount, payable by check, cashier check or money order, and should reference the project by site address.⁸⁴ The application fee is the only required document that must be mailed directly to the Program Administrator.⁸⁵ After a project is assigned to an incentive step, the application fee check must be mailed within 7 calendar days. If the check is not mailed within 7 calendar days, the project may be cancelled. Application

⁸³ An Energy Efficiency Audit (EEA) report issued within the past five years. A Title 24 energy efficiency compliance report issued within the past three years may also be used in lieu of an Energy Efficiency Audit.

⁸⁴ Application fees are specific to an application, not a site. If the same site reapplies to the program, they will need to submit a new application fee.

⁸⁵ A single application fee check for each project is required.

fee checks returned by the financial institution without payment may result in cancellation of the application.

The application fee will be refunded upon completion and verification of the installed SGIP project. Prior to project completion, application fees are non-refundable once a Confirmed Reservation has been issued.⁸⁶ All forfeited application fees will be allocated to the Program Administrator's current incentive step.

3. Equipment Specifications (All Projects)

Manufacturer equipment specifications for all major components of the system are required, in addition to the nameplate capacity, rated capacity (kW) efficiency and, if necessary, fuel consumption and waste heat recovery rate of the system.

Proof of power factor eligibility is also required for microturbines, internal combustion engines, gas turbines and stream turbine CHP applications (where applicable) and must include self-generating facility design specifications and/or manufacturer's specifications which show that the system will be capable of operating between 0.95 PF lagging and 0.90 PF leading.

4. Proof of Utility Service & Load Documentation (All Projects)

Participation in the SGIP is restricted to customers who are located in PG&E, SCE, SoCalGas or SDG&E service territories and physically connected to the Electric Utility transmission and distribution system. All applications must include a copy of a recent electric utility bill indicating the account number, meter number, site address, and Host Customer name. Customers applying in PG&E and SoCalGas territory must also submit a copy of a recent gas utility bill. For new construction, the Host Customer must submit confirmation from the serving utility that their site is within the Program Administrator's service territory. For projects applying in CSE and SoCalGas territory, applications for technologies that discharge electricity to the onsite load must include a copy of the previous 12-months of electric consumption including maximum demand and kWh consumption to confirm that the participating generation system meets the program sizing requirements. Electric utility customers of SCE and PG&E are not required to submit their 12 month electric consumption and demand data. SDG&E customers are also required to submit an Authorization to Receive Customer Information form, signed by the utility customer of record that authorizes CSE to access utility account information for systems greater than 10 kW or any other instance under CSE discretion if needed.

If the generation system is being sized based on new or future load growth (i.e. new construction or load growth due to facility expansion or other load growth circumstances) applications must include an engineering estimate with appropriate substantiation of the site's annual peak demand

-

⁸⁶ Application Fees will not be altered due to project changes that may result in a different incentive.

forecast. Suggested methods of demonstrating load growth include Application for Service with corresponding equipment schedules and single line diagram; building simulation program reports such as eQUEST, EnergyPlus, EnergyPro, DOE-2, and VisualDOE; or detailed engineering calculations.

5. Preliminary Monitoring Plan (All 3 Step Applications >=30 kW)

The preliminary monitoring plan should demonstrate the following components:

Description of the proposed SGIP system:

Description of the system with an overview of the energy services to be provided (e.g., generation, waste heat recovery, storage, etc.) by the system to the host site; the major components making up the system; and the general operating schedule of the system (e.g., is it 24x7x365 or 10x6x365, etc.); Include photos of the system if available.

Break out subsystems such as waste heat recovery systems in order to provide context for thermal energy metering systems. Provide similar descriptions for other important subsystems such as energy storage when combined with wind systems.

A description of the existing load at the Site and identification of the sources of the fuel that would be displaced by operation of the SGIP system (i.e., electricity provided by XYZ utility or natural gas provided by ABC utility) and photos of the interface locations where the SGIP system would be located to displace the load.

Description of the metering system and metering approach:

An overview of the performance data to be collected (e.g., electrical, useful thermal energy, fuel consumption, etc.) and a simplified layout of the system showing major components (e.g. generator, waste heat recovery, storage etc.) and location of the proposed metering points and data to be collected at those points (i.e. electrical, flow, temp, fuel etc.) is required.

Two Step Applications will include (as part of their Reservation Request) a Proposed Monitoring Plan as outlined in *Section 6.10.2*.

6. <u>Minimum Operating Efficiency Worksheet w/Backup Documentation (Blended Projects Only)</u>

The Minimum Operating Efficiency Worksheet (MOEW) is used to evaluate a project's technical ability to meet the following requirements:

- a) Minimum operating efficiency requirement which can either be satisfied by meeting:
 - Waste Heat Utilization or
 - Minimum Electrical Efficiency Requirements
- b) Thermal Load Coincidence

- c) CHP System Efficiency and NOx Emission Qualification
- d) Greenhouse Gas Emission Standard
- e) Electrical Load Coincidence (Electrical Export Eligibility)

a) Minimum Operating Efficiency Calculations

The MOEW is a spreadsheet used to evaluate the project's estimated minimum operating efficiency over 10-years. Additionally, the MOEW verifies the first year minimum operational efficiency for Electric-only fuel cells. All applications proposing blended renewable-fueled technologies must provide backup documentation along with the MOEW. CHP systems must additionally include engineering calculations with documented assumptions regarding the site's thermal load. All assumptions, backup documentation, hand calculations, models (with inputs and outputs) and custom spreadsheets used to develop the forecasts must be included in the documentation. Forecasts based solely on "professional experience" or subjective observation will be rejected.

Specifically, the following applicable documentation must be provided:

• Generator & Thermal System Description

The application must include the performance and capacity specifications for the proposed CHP system and all thermal system equipment that the CHP system interacts with or serves. This includes but is not limited to the generator system, heat recovery system, heat exchangers, absorption chillers, boilers, furnaces, etc. In addition, a thermal process diagram must be provided as part of the documentation package that shows the configuration of the generator(s), heat recovery system, pumps, heat exchangers, thermal load equipment, and the working fluid flow and temperatures in/out of each piece of major equipment at design conditions.

Forecast of Generator Electric Output

The MOEW must include a forecast of the monthly generator electric output (kWh/month) for a twelve-month period. The generator electric output forecast must be based on the operating schedule of the generator, historical or site electric load forecast and maximum/minimum load ratings of the generating system; exclusive of any electric energy used in ancillary loads necessary for the power production process (i.e., intercooler, external fuel gas booster, etc.).

Forecast of Generator Thermal Output

The application must include a forecast of the monthly generator thermal output (Btu/month) for a twelve-month period. The generator thermal output forecast must be

based on the electric output forecast of the generating system and the waste heat recovery rate specifications of the system.

Forecast of Generator Fuel Consumption

The application must include a forecast of the generating systems monthly fuel consumption (Btu/month) for a twelve-month period. The generator's fuel consumption forecast must be based on the generating system electric output forecast and the systems fuel consumption specifications.

Forecast of Thermal Load Magnitude

The application must include a monthly Thermal Load forecast (Btu/month) for a twelvemonth period for the Thermal Load served by the CHP system. The forecast must be based on engineering calculations, thermal system modeling, historical fuel billing, measured data or a combination of these methods. The Thermal Load forecast must be independent of the generator operation forecast. If historical natural gas or other fossil fuel consumption records (e.g., billing records) are used, the combustion efficiency of the natural gas or fossil fuel fired equipment that is being displaced must be included. Historical fuel consumption must be discounted to account for equipment Thermal Load that will not be displaced by the prime mover's thermal energy.

Forecast of Useful Thermal Output

The useful thermal output of the CHP system will be the lesser of the Thermal Load forecast, or the prime mover's thermal output coincident with the Thermal Load. The useful thermal output is the value used in calculating the P.U. Code 216.6 requirements.

b) Thermal Load Coincidence

Thermal load coincidence is calculated in the worksheet by comparing the waste heat recovered to the thermal load on an annual basis. The backup documentation listed above for the forecast of generator thermal output and forecast of thermal load magnitude will be sufficient to meet this operating efficiency requirement.

c) CHP System Efficiency and Proof of NOx Emission Qualification

Applications must include documentation substantiating that the generating system meets or exceeds the 60% minimum system efficiency and NOx emissions are at or below the applicable emission standard. One of the following documents must be included to determine the NOx emissions (lb/MWh) of the proposed system:

- Manufacturer emission specifications based on factory testing using California Air Resources Board (CARB), EPA or local air district test methods⁸⁷ for the proposed generating system as configured for the Site.
- CARB distributed generation certification
- Emission engineering calculations for the proposed generating system as configured for the site.

Conversion of emissions concentration (ppm) to production based emissions rates (lb/MWh) shall use the method found in Appendix C of this Handbook. Units that do not pass the emission standard may use emission credits. If the application claims NOx emissions credits for their waste heat utilization emission, credit calculation documentation based on the amount of waste heat utilized over a twelve-month period must be provided.

d) Greenhouse Gas Emission Rate Testing Protocol (Electric-Only Fuel Cells)

Electric-only fuel cells operating on blended fuel must provide the ASME PTC 50-2002 test as backup documentation to the MOEW. Please see Section 6.1.3 for further information.

e) Electric Load Coincidence (Electrical Export Eligibility)

The application must include the monthly electrical load for the previous 12 months. This information will be used in the MOEW to determine electrical load coincidence with electrical generation on an annual basis. If the generator is eligible to export electricity to the grid, the electrical generation will be compared to 125% of electrical load on an annual basis.

7. Proof of Adequate Fuel or Waste Energy Resource (Renewable Fuel, Waste Gas & Waste Energy Projects)

<u>On-site Biogas Projects</u> must include an engineering survey or study confirming the renewable fuel (*i.e.*, adequate flow rate) and the generating system's average capacity during the term of the Project's required permanency period.

<u>Biogas Projects utilizing Directed Renewable Fuel</u> must include documentation of the forecasted fuel consumption of the generator over the life of project.

<u>Projects utilizing Waste Gas Fuel</u> (microturbines, internal combustion engines, gas turbines and steam turbine CHP waste gas fuel applications only) must include an engineering survey or study

93

⁸⁷ Acceptable test methods include but not limited to CARB Test Method 100 and USEPA Test Method 7.

confirming that there is adequate on-site waste gas fuel (i.e., adequate flow rate) for continuous operation of the self-generation unit for the term of the project's required permanency period.

<u>Proposed Pressure Reduction Turbine applications</u> must include an engineering survey or study confirming adequate temperature, pressure and flow within the piping system, and the generating system's rated capacity. The rated capacity must be based upon the average pressure drop across and flow through the turbine, when flow exists, as determined by historical flow and pressure data from the previous year if available, or from an engineering estimate if new construction or expanded load. Additionally, the survey or study must show that the capacity factor for the proposed project will be greater than or equal to 40% based upon conditions over the course of a full year, or from an engineering estimate for future conditions.

<u>Proposed Waste Heat to Power applications</u> must include an engineering survey or study confirming adequate waste heat production rate and temperature, and the generating system's rated capacity. The rated capacity must be based upon the average waste heat production rate and temperature, when waste heat is available, as determined by historical waste heat and temperature data from the previous year if available, or from an engineering estimate if new construction or expanded load. Additionally, the survey or study must show that the capacity factor for the proposed project will be greater than or equal to 40% based upon conditions over the course of a full year, or from an engineering estimate for future conditions.

<u>Proposed Wind projects</u> must include an engineering survey or study evaluating the annual average wind speed at the hub height of the wind turbine. The study must confirm that the average annual wind speed is equal to or greater than 10 mph (4.5 m/s). The wind resource can be verified using wind resource maps from NREL or the CEC and standard formulas for correcting for differences in tower heights or by gathering wind data on site at the turbine's proposed hub height for one year.

8. Customer Resiliency Attestation (Resiliency Projects)

For renewable generation projects applying for the resiliency adder incentives we require that renewable generation:

- A. Provide an estimate of how long a project's renewable generation system will provide electricity for the relevant facility average load during an outage;
- B. Indicate whether a project's critical loads can and will be isolated;
- C. Provide an estimate of how long a project's renewable generation system will provide electricity to critical uses during an outage;
- D. Provide an estimate of how long the renewable generation system can operate in less-than favorable circumstances.
- E. Summarize information given to the customer about how the customer may best ensure operation of the renewable generation system, in the case of a PSPS event announced in advance;
- F. Attest to the truth of the information provided;

- G. Provide an attestation from the customer indicating that he or she received this information prior to signing a contract; and
- H. Demonstrate that an Authority Having Jurisdiction has approved plans showing that the system can operate in islanded mode, has inspected the system after installation and has authorized operation.

9. Proof of Coordination with Local Government and California Office of Emergency Services

Non-Residential customers must demonstrate that they are coordinating with their local government and the California Office of Emergency Services if applying for Equity Resiliency or for a Resiliency Adder Incentive. Customers that cannot demonstrate this, will have their application deprioritized in favor of applicants that can provide proof.

Additional Requirements for Two Step Applications

All 2 step applications must include as part of their Reservation Request materials, all applicable requirements of the Proof of Project Milestone. See Section 6.10.2.

6.10.2 Required Documentation for Proof of Project Milestone

Generation applications must provide a copy of the following:

Table 6.10.2: Proof of Project Milestone Requirements

Required Materials 1. Completed Proof of Project Milestone Form (All 3-Step Projects) 2. Copy of RFP or equivalent after 90 days (Public Entity Projects Only) 3. Copy of Executed Contract or Agreement for Installation (All Projects) Includes Required Warranty Documentation 4 Proposed Monitoring Plan (All Projects >=30 kW) 5. Proof of Fuel Contract and Documentation (Renewable Fuel and Waste Gas Projects Only) Renewable Fuel Contract (Directed Renewable Fuel Only) Directed Renewable Fuel Attestation - System Owner & Fuel Supplier (Directed Renewable Renewable Fuel Affidavit (On-site Renewable Fuel Only)

1. Online Proof of Project Milestone Form (All 3-Step Projects)88

Fuel Clean-up (On-site Renewable Fuel Only) Waste Gas Fuel Affidavit (Waste Gas Fuel Only)

⁸⁸ Not required for 2-Step Applications as part of the Reservation Request Package.

The Online Proof of Project Milestone Form must be completed and signed by the applicant and representatives with signature authority for both the system owner and host customer (if not host customer). The online form must identify updated project information including the installation contractor's name, telephone number and contractor license number. All systems must be installed by an appropriately licensed California contractor in accordance with rules and regulations adopted by the State of California Contractors' State Licensing Board. Installation contractors must have an active A, B, or C-10 license.

2. Request for Proposals (RFP) Documentation (Public Entities Only)

Notice to Invite Bids, or similar solicitation issued for the installation, lease and/or purchase for systems proposed for the SGIP. The RFP must include sufficient project details such as the scope of work, schedule, terms, budget, and/or system components desired. For Public Entities not issuing an RFP, alternative documentation such as an executed letter of intent to engage with a contractor on the Host Customer letterhead, an executed contract/agreement for system installation/lease, an equipment purchase order, or alternate system ownership agreement must instead be submitted within 90 calendar days of the date the Conditional Reservation Letter. Proof of Project Milestone documentation must then be submitted within 240 days of the date the Conditional Reservation Letter.

3. Executed Contract and/or Agreement for System Installation (All Projects)

A copy of the executed contract for purchase and installation of the system, and/or alternative System Ownership Agreement (such as a Power Purchase Agreement) is required. The contract/agreement must be legally binding and clearly spell out the terms and scope of work. Purchase and/or installation agreements must also include system equipment and eligible system costs. All contracts/agreements must be signed by appropriate representatives (Host Customer, Installer, and/or System Owner) who are a party to the agreements of the SGIP reservation.

As part of the Executed Contract, all generation systems are required to include a minimum 10 year service warranty (with the exception of wind turbines which must have a minimum 20 year service warranty). A service warranty ensures proper maintenance and continued project performance. The service warranty must cover the system maintenance to include (but not limited to) system support, problem diagnosis, on-site repair and preventative maintenance. The warranty should also include language to guarantee the continued performance of the system over the warranty period. The system owner must provide proof of warranty and maintenance contract, and specify the warranty and maintenance contract start and end dates.

4. Proposed Monitoring Plan (All Projects that are 30 kW or larger)

The proposed monitoring plan should demonstrate the following components:

Description of the proposed SGIP system(s)

Description of the system(s) with an overview of the energy services to be provided (e.g., generation, waste heat recovery, storage, etc.) by the system(s) to the host site; the major components making up the system(s); and the general operating schedule of the system(s) (e.g., is it 24x7x365 or 10x6x365, etc.). Include photos of the system(s) if available.

Break out subsystems such as waste heat recovery systems in order to provide context for thermal energy metering systems. Provide similar descriptions for other important subsystems such as energy storage when combined with wind systems.

A description of the existing load at the Site and identification of the sources of the fuel that would be displaced by operation of the SGIP system(s) (i.e., electricity provided by XYZ utility or natural gas provided by ABC utility) and photos of the interface locations where the SGIP system would be located to displace the load.

Description of the metering system and metering approach

An overview of the performance data to be collected (e.g., electrical, useful thermal energy, fuel consumption, etc.) and a simplified layout of the system showing major components (e.g., generator, waste heat recovery, storage, etc.) and location of the proposed metering points and data to be collected at those points (i.e., electrical, flow, temp, fuel, etc.) is required.

<u>Description of the approach to be used for collecting, storing and transferring the necessary</u> performance data

- For example, if useful thermal energy data is to be collected, the reasoning behind the selected metering points
- Frequency with which the data is to be collected (e.g., 15 min intervals)
- Data storage capability and approach for transfer of data (e.g., cell modem) and frequency of reporting to PDP (e.g., daily, weekly) [this could also include frequency for reporting of data to Program Administrators, such as monthly]

Identification of the metering system components by performance data type (including manufacturer and model number)

- Electrical metering equipment (AC meters must be listed on the CEC's list of Eligible System Performance and Revenue Grade Meters to be found on http://www.gosolarcalifornia.ca.gov/equipment/index.html)
- Thermal energy metering equipment
- Fuel consumption metering equipment
- Data acquisition (i.e., logger) system

5. Proof of Fuel Contracts and Documentation (Renewable Fuel and Waste Gas Projects Only)

Copy of Executed Renewable Fuel Contract (Directed Renewable Fuel Only)

The Contract should at a minimum include term (minimum of 10 years), cost, amount of renewable fuel injected on a monthly basis for the length of the contract, address of renewable fuel facility, location of pipeline injection site, name of pipeline owner, and facility address of Host Customer.

The SGIP Program Administrator or designee has the right to audit and verify the generator's renewable fuel consumption upon request over the life of the contract.

The Host Customer will consume the contracted renewable fuel for the sole purpose of fueling the SGIP project and the contract should include a forecast for the system's anticipated fuel consumption.

The contract should include a quarterly true-up mechanism in which the customer and renewable fuel supplier agree to true-up based on actual deliveries of renewable fuel. Note that the fleet of SGIP systems will have its own revenue-grade, electric NGOM and gas meters that are accessible via internet by the Program Administrator or designee.

- If less on-site fuel is consumed than renewable fuel is nominated into the pipeline, then parties agree to a financial make-whole provision.
- If more on-site fuel is consumed than renewable fuel is nominated into the pipeline, then parties
 agree to a make whole provision, such that Customer Generator consumes renewable fuel, as
 measured annually.

Directed Renewable Fuel Attestation (Directed Renewable Fuel Only)

Attestation letter from the system owner of the intent to notionally procure renewable fuel and attestation from the fuel supplier that the fuel meets the applicable renewable portfolio standard eligibility requirements for renewable fuel injected into a natural gas pipeline.

Renewable Fuel Use Affidavit (On-site Renewable Fuel Only)

Application documentation must include a signed affidavit that projects will not switch to non-renewable fuel for a period of ten years for all technologies. The SGIP Program Administrator has the right to audit and verify the generator's renewable fuel consumption upon request over the life of the contract.

Fuel Cleanup Equipment Purchase Order (On-site Renewable Fuel Only)

When applicable, application documentation must include a purchase order for Renewable Fuel cleanup equipment that lists the fuel cleanup equipment as a separate invoice item.

Waste Gas Fuel Use Affidavit (Waste Gas Only)

When applicable, application documentation must include a signed affidavit that Projects will be fueled solely (100%) with Waste Gas for a period of ten years.

6.10.3 Required Documentation for Incentive Claim

Generation applications must provide a copy of the following:

Table 6.10.3: Incentive Claim Requirements

Required Materials

- 1. Completed Incentive Claim Form (All Projects)
- 2. **Proof of Authorization to Interconnect** (Projects that interconnect with the electrical grid)

3. Final Permits

- Building Permit Inspection Report (All Projects)
- Additional AHJ Grid Islanding Plan (Equity Resiliency and Equity)
- Air Permit Documentation (Non-Renewable Fuel Only)

4. Substantiations:

- New or Expanded Load (All Projects)
- Renewable or Waste Resource (On-site Renewable Fuel and Waste Energy Only)
- Fuel Cleanup Skid Cost (On-site Renewable Fuel Only)
- Renewable Fuel Documentation/Contract Commencement (Directed Renewable Fuel Only)
- Renewable Fuel Metering Specifications (Directed Renewable Fuel Only)
- 5. Planned Maintenance Coordination Letter (>=200 kW CHP Systems Only)
- 6. Final Monitoring Schematic (All Projects)
- 7. PBI Setup Sheet (All Projects >= 30kW)

1. Online Incentive Claim Form (All Projects)

The ICF information must be complete, accurate and represent the actual system, project costs, and/or fuel information as installed (including system size and type). It must also be signed by the applicant, host customer and system owner (if not the host customer).

2. **Proof of Authorization to Interconnect** (Projects that interconnect with the electrical grid)

Host customers and/or system owners will be required to execute certain documents such as, but not limited to, an "Application to Interconnect a Generating Facility" and a "Generating Facility Interconnection Agreement" with the local Electric Utility. A copy of the signed letter from their Electric Utility granting the host customer and/or system owner permission to interconnect and operate in parallel with the local grid should be submitted as proof of Authorization to Interconnect.

Applicants, host customers and system owners are solely responsible to submit interconnection applications to the appropriate electric utility interconnection department as soon as the information to do so is available to prevent any delays in system parallel operation.

3. Final Permits

Building Inspection Report (All Projects)

A copy of the final building inspection report (or proof of exemption) demonstrating that the Project has been inspected by the local Authorities having Jurisdiction (AHJ), meets all codes and standards of the permitting jurisdiction, and has been authorized for operation. Contact your local permitting jurisdiction to learn about permitting requirements.

<u>Air Permitting Documentation</u> (Non-Renewable Fuel Only)

For those Projects that require an air permit from the local air district, the application must include a copy of the final documentation indicating compliance with all applicable air pollution regulations (or proof of exemption).

4. Substantiations:

New Construction or Added Load (All Projects)

For Projects where host customer estimated the future load to justify system size, applications must include documentation demonstrating that the load forecast has materialized.

Renewable Fuel or Waste Energy Resource (On-site Renewable Fuel and Waste Energy Only)
For Projects where the host customer, applicant or system owner provided renewable fuel estimates or Waste Energy resource estimates, applications must include documentation demonstrating that the on-site Renewable Fuel or Waste Energy resource has materialized.

<u>Fuel Cleanup Skid Cost Documentation</u> (On-site Renewable Fuel Only)

On-site biogas projects must include documentation substantiating the fuel cleanup skid cost.

Renewable Fuel Documentation & Contract Commencement (Directed Renewable Fuel Only) Documentation from the supplier showing that the fuel is renewable and that it meets the quality standards to be injected into the local natural gas pipeline. Documentation should also be submitted showing that the contract has commenced and the supplier has begun nominating the renewable fuel into the pipeline (e.g. one month fuel invoice). The project will be given up to one year from the date the Incentive Claim was received by the SGIP Program Administrator for commencement of the contract. However, no incentive will be paid until the contract has commenced.

Renewable Fuel Metering Specifications (Directed Renewable Only)

Documentation should also be provided to include make, model, specifications and serial number of installed revenue grade electric NGOM and gas meters.

5. Planned Maintenance Coordination Letter (CHP Projects >= 200 kW Only)

When applicable, applications with microturbine, internal combustion engine, gas turbine and steam turbine CHP systems operating on blended fuel sized greater than 200 kW must include a maintenance coordination letter to the host customer's electric utility. The maintenance coordination letter shows the

system owner will schedule planned maintenance only between October and March and, if necessary, only during off-peak hours and/or weekends during the months of April to September.

6. Final Monitoring Schematic (All Projects)

The final monitoring schematic includes an electrical single line diagram (SLD) that includes the generator, the utility meter, and the load panel, and if applicable, the PBI meter. CHP projects must also include a Process and Instrumentation Diagram (P&ID) that shows the configuration of the generator(s), heat recovery system, pumps, heat exchangers, and thermal load equipment as well as the fuel and thermal metering points. Documentation must also be provided if there is a change in the make and model of the meters to be used (from what was submitted with the Proposed Monitoring Plan at the Proof of Project Milestone).

For PBI projects, the applicant must also provide the name of the Performance Data Provider (PDP) they are contracting with. A copy of the contract between the PDP and the applicant may be requested at the Program Administrator's discretion.

7. PBI Setup Sheet (for projects 30 kW or larger)

The PBI setup sheet must include information for all meter(s) installed for the purpose of monitoring system performance. The meter IDs listed on the PBI setup sheet must also match the meter IDs specified on the Final Metering Schematic.

6.11 Metering & Monitoring Requirements for Generation Projects

All SGIP technologies 30 kW or larger must install metering and monitoring equipment that measures net electrical output from the system(s). In addition to electrical output, fuel input metering into the generator(s) is required for all conventional CHP and fuel cell technologies, regardless of fuel type. CHP technologies operating on blended fuels must also install metering and monitoring equipment that measures and reports useful thermal energy delivered to the site from the CHP system.

System owners must install and maintain metering and monitoring equipment at their own cost and are responsible for the choice and installation of the metering hardware. All AC electric meters must be listed on the CEC's list of Eligible System Performance and Revenue Grade Meters to be found on http://www.gosolarcalifornia.ca.gov/equipment/index.html.

It is also the responsibility of the System Owners to contract with a Performance Data Provider (PDP) for a minimum of five years. PDPs will be tasked with recording performance data from the installed metering and monitoring equipment and submit it to the Program Administrators on a monthly basis.

All installed meters (electrical, thermal and fuel) must fulfill the following requirements:

- Protocols for the minimum required performance/output data must enable any PDP to communicate with the meter and obtain the minimum required performance/output data from the meter and/or logger.
- All meters must have the capability to report their data remotely.
- · Data reporting must occur on a daily basis.
- All meters must have the ability to retain collected data in the event of a power outage. Meters
 must have the capability to store 7 days' worth of data.
- All meters must provide the PDP provider or defined list of authorized users with the ability to
 access and retrieve the minimum required data from the meter using the Meter Communication /
 Data Transfer Protocols described in Section 7.1.
- The meters will be inspected as part of the project Post-Installation inspection process.

The following section details the minimum meter requirements per type of meter installed (electrical, thermal, fuel).

6.11.1 Minimum Electrical Meter Requirements

All systems 30 kW and larger must be installed with a meter or metering system which allows the System Owner and Program Administrator to determine the amount of net system energy production and allows the System Owner to support proper system operation and maintenance.

Electrical meters installed on the SGIP project provide data used to assess performance of the system, to analyze impact of the SGIP system on utility distribution systems, the peak system demand of the utility and net GHG emission impacts.

Meter Type

All systems are allowed to use on-board electrical meters, however, the meter must meet the minimum meter requirements of this section. For all systems receiving PBI payments, the installed meter(s) may be a separate Interval Data Recording (IDR) meter(s), or a complete onboard system that is functionally equivalent to an IDR meter, recording data no less frequently than every 15 minutes. Program Administrators may have additional meter functionality requirements for systems receiving PBI payments, as the Program Administrators will use these meters to process PBI payments, and system compatibility may be required. For example, meters and service panels must meet all local building codes and utility codes. The meter serial number must be visible after installation.

Acceptable Electrical Metering Points

The electrical metering system must meter delivered energy by having a meter at the output of the generator and after power delivery to all parasitic loads. When an on-board electrical metering system is used, the meter must have multiple channels in order to monitor parasitic energy consumption as well as generator output and report net generation output.

Meter Accuracy

All systems receiving a PBI incentive must install a meter accurate to within ± 2% of actual system output. This applies to on-board electrical meters as well as external IDR meters.

Meter Measurement and Time Granularity of Acquired Data

Electric meters must measure the net energy generated (kWh) and net real power delivered (kW). The PDP must log all required generator performance / output data points no less frequently than once every 15 minutes. The meter needs to generate an accurate time/date stamp.

Meter Testing and Certification

 \pm 2% meters required for PBI for AC systems must be tested according to all applicable ANSI C-12 testing protocols pertaining to the monitoring of power (kW) and energy (kWh). The accuracy rating of \pm 2% meters must be certified by an independent testing body (i.e., a NRTL such as UL or TUV). DC electrical meters must also have a \pm 2% accuracy pertaining to the monitoring of power (kW) and energy (kWh), by the meter manufacturer.

Meter Display

All meters must provide a display showing the meter's measured net generated energy output or offset and measured instantaneous power. This display must be easy to view and understand and must be physically located either on the meter or on a remote device. For PBI, if a remote device is the only visible access, the PA may ask for verification.

6.11.2 Minimum Thermal Metering Requirements

All CHP systems 30 kW and larger running on blended fuel must be installed with a metering system which allows the System Owner and Program Administrator to determine the amount of useful thermal energy production and allows the System Owner to support proper system operation and maintenance. Thermal energy metering systems installed on the SGIP project provide data used to assess thermal performance of the system; including its ability to meet on-site thermal energy demands (thereby offsetting consumption of fossil fuels) and meet thermal energy efficiency requirements prescribed by PUC 216.6. Thermal energy performance data will also be used to assess impact of the SGIP system on net GHG emission impacts⁸⁹ and minimum system operating efficiency requirements.

Meter Type

All CHP systems that are 300 kW and smaller will be allowed to use an on-board thermal metering system in order to minimize cost. The specific instrumentation required to measure useful thermal energy production will vary depending on the configuration and type of heat recovery system (e.g., liquid, steam, direct exhaust). Common flow measuring devices include insertion type or ultrasonic flow meters.

⁸⁹ Thermal energy metering systems may also provide SGIP System Owners with a potential means of verifying carbon emissions and carbon emission reductions.

Temperature measurement may be done with thermocouples. On-board thermal metering systems just as external thermal metering systems must measure useful thermal energy production. Proposed meter and sensor types shall be identified in a Monitoring Plan developed for each individual project. On-board meters must meet the same requirements as external meters which are outlined below.

Acceptable Thermal Metering Points

Proposed meter and sensor locations shall be identified in a Monitoring Plan developed for each individual project. It is recommended for direct exhaust combined cooling heating and power (CCHP) systems, that the chilled water output be measured, rather than measuring exhaust flows and temperatures as a way to calculate the useful thermal output.

Meter Accuracy

The accuracy of the metering system for useful thermal energy production must be within \pm 5% at design conditions. This requirement applies to on-board as well as external thermal metering systems. The Monitoring Plan shall include a section describing monitoring system maintenance plans that will be implemented to ensure compliance with the accuracy requirement throughout the PBI period.

Meter Measurement and Time Granularity of Acquired Data

The PDP must log all required useful heat recovery system performance / output data points no less frequently than once every 15 minutes. Calculated values of useful heat recovery must be reported in 15 minute intervals. The sum of four consecutive intervals would represent the industry standard rate of Useful Heat Recovery in units of MBtu/hr. The heat transfer fluid specific heat and density must be reported at the time of commissioning and then reported again to the PA if there is a change. The meter needs to generate an accurate time date stamp.

6.11.3 Minimum Fuel Metering Requirements

All CHP systems and electric-only Fuel Cells 30 kW and larger and regardless of fuel type or renewable fuel blending percentage must be installed with a fuel metering system which allows the System Owner and Program Administrator to determine the amount of fuel consumption and allows the System Owner to support proper system operation and maintenance. In the case of onsite blended fuel projects, the metering system must allow the Program Administrator to determine the amount of renewable and non-renewable fuel consumption. The recorded data will be used to calculate the minimum system operating efficiency and GHG emissions of the system. These calculated values will be used to monitor compliance with the Program's GHG emission limits and minimum system operating efficiency requirements.

Meter Type

All CHP systems and electric-only Fuel Cells that are 300 kW and smaller will be allowed to use an on-board fuel metering system to minimize cost. External fuel gas flow measurements are typically done in one of three ways:

1. Mass flow meter

- 2. Calculated based upon continuous differential pressure measurements across an orifice
- 3. Utility gas meter

On-board fuel metering systems, just as external fuel metering systems, must measure fuel consumption by the generator. The proposed meter type shall be identified in a Monitoring Plan developed for each individual project. On-board meters must meet the same requirements as external meters which are outlined below.

Acceptable Fuel Metering Points

For fuel metering that is external to the generator, an acceptable metering point is before fuel entry into the generator but downstream of any other loads (e.g., natural gas boiler, un-incentivized CHP system). For on-board metering systems, the fuel must be metered before any portion is consumed by the generator. For on-site blended fuel projects there must be separate meters for the renewable fuel and the non-renewable fuel. Proposed meter locations shall be identified in a Monitoring Plan developed for each individual project.

Meter Accuracy

Flow measurement must include temperature and pressure compensation and must measure standard cubic feet (at 60 °F and 1 atmosphere) to within ± 5% of maximum flow for the generator at design conditions. This requirement applies to on-board as well as external fuel metering systems.

• Meter Measurement and Time Granularity of Acquired Data

The PDP must log all required generator system fuel input data points no less frequently than once every hour. Calculated values must be reported in one-hour intervals. Data must be recorded in units of standard cubic feet per minute. The Btu content and basis (HHV/LHV) of the fuel must be reported during commissioning either through data provided by the gas company or determined by analysis. Btu content of the fuel will need to be re-analyzed and reported to the PA when there is a reason to believe it has changed. The meter needs to generate an accurate time date stamp.

7 Metering & Data Collection

This section describes the requirements for data reporting, PDP application process, data security, and measurement and evaluation activities.

7.1 Data Reporting and Transfer Rules – Contract for PDP Services

It is the responsibility of the System Owners to contract with a Performance Data Provider (PDP) for a minimum of five years and ensure that performance data is provided to the Program Administrator or their designee monthly for five years. A list of qualified PDPs can be found on the Program Administrators websites.

The following are the PDP's primary responsibilities:

- · Manage meter reading/data retrieval schedule
- Read and retrieve performance meter data
- Post data on appropriate Program Administrator server on a consistent and reliable schedule, per Program Administrator requirements.
- Validate performance data prior to providing to the PA using the approved validation rules outlined in this document
- Calculate annual production of generating system and net electrical energy discharged or offset for energy storage systems for incentive payment
- Format data using an approved SGIP format.
- Troubleshoot and resolve communications issues
- Store data in accordance with program requirements
- Make historical performance data available to Program Administrators as requested
- Provide technical support to Program Administrators as well as customer support
- Communicate meter/device changes to the Program Administrator
- Provide disaster recovery and data backup services as requested by respective Program Administrator
- Manage data on PDP server
- Ensure confidentiality of customer information and performance data
- Possess technical expertise and capability
- · Comply with all State and Federal laws

The purpose of the following section is to outline the data reporting requirements (format, delivery method) and schedule for submitting data reports to the Program Administrators.

7.1.1 Data Format

Meter data must conform to the SGIP program requirements as set forth in *Section 5.5* and *Section 6.11* (15-minute electrical and thermal data as well as hourly fuel data, when applicable).

The PDP is responsible for submitting the incoming meter data to the Program Administrators on a monthly basis in the following two formats: Meter Interval Report and Application Interval Report. The Application Interval Report will be used for calculating the Annual Performance Based Incentive payments. The Meter Interval Report will be referenced as needed to support the data submitted for payment processing in the Application Interval Report. Both file types may be used for reporting and/or auditing purposes.⁹⁰

The data file format for submission will be ".csv". The file formats are designed for bulk submission of data for any number of applications in a single calendar month. However, the file types (Meter Interval and Application Interval) must be submitted in separate files. If a Performance Data Provider is contracted to report data for more than one Program Administrator, they shall submit a separate file for each Program Administrator to maintain the confidentiality of the data.

7.1.2 Meter Reading and Data Submission Timeline

Meter data will be read remotely no less frequently than on a daily basis. In the event there is a communication problem between the PDP and the meter, and the 15-minute interval data is accumulated over a 24-hour period, it is acceptable to estimate the meter's 15-minute interval data. For more information, refer to The PDP Specification. Accumulated data for a period longer than 24 hours will not be accepted. Other than this exception, the Program Administrator is not responsible for, and will not pay any customer incentives based on missing, estimated or invalid performance data.

Data for a SGIP project must be submitted in full calendar months. Once a month of data has been collected, the PDP has up until the 1st of the following month to validate, format, and submit the Meter and Application Interval data for that project. Annual PBI Incentive Payment amounts will only be processed after a full year of data has been submitted for the project.

For new SGIP projects, data recording for PBI Payment purposes should typically commence on the 1st of the month following the initial 50% Incentive Payment. Customers may choose to submit data since interconnection, if interconnection has occurred within the past 6 months. Any request for an alternate data collection commencement date will require Program Administrator approval.

⁹⁰ For a detailed description of the Meter and Application Interval Reports and submittal processes, please refer to The PDP Specification.

7.1.3 Online Submission Process

All performance data will be submitted via the SGIP online application database PDP Upload Portal. The portal will be accessed through www.selfgenca.com. Files that are submitted via e-mail will not be accepted. 91

7.1.4 PDP Data Validation

The PDP must validate all data prior to submitting it to the PAs. The following data validation rules shall apply:

- Time Check of Meter Reading Device/System (all)
- Meter Identification Check (all)
- Time Check of Meter (all)
- Pulse Overflow Check (if applicable to metering system)
- Test Mode Check (if applicable to metering system)
- Sum Check
- Spike Check
- kVARh Check

Descriptions of these validation rules are included in The PDP Specification.

7.1.5 Data Audits & Payment Validation

The Program Administrators may, at their discretion, perform validations on incentive payments prior to issuing payments to customers participating in this program. The validations will compare actual yearly incentive payments with expected payments based on design specifications and expected performance data submitted with the approved incentive reservation documentation. If payments fall outside expected ranges for the year, the incentive payment will be withheld until the Program Administrator determines to its satisfaction the reason for the discrepancy.

The PDP will work with the System Owner to resolve any discrepancies identified by the Program Administrator, which may include testing and/or recalibrating the meter/devices if deemed necessary. The Program Administrators are not responsible for the costs associated with investigating and resolving any such discrepancies (i.e., testing, meter replacement hardware, installation labor). However, if the Program Administrator requests an investigation that finds that the metering system is accurate, the Program Administrator will pay all reasonable and necessary costs for the investigation.

⁹¹ Please refer to The PDP Specification for a detailed description of the data submittal process.

The Program Administrator may also perform random audits of PDP data to ensure accuracy and compliance with the requirements outlined in this document, or as part of the SGIP Measurement and Evaluation Program in accordance with the SGIP Handbook. Any PDP found to be in violation of any of these requirements will be subject to the penalties outlined later in this document. The Program Administrator, via the servicing local utility or its designated contractor may, at its discretion, inspect and test the performance meter or install separate metering in order to check meter accuracy, verify system performance, or confirm the veracity of monitoring and reporting services.

Any additional metering installed by or at the request of the Program Administrator will be paid for by the Program Administrator. However, in the event metering is installed during the course of an audit or investigation initiated by the Program Administrator where cheating or tampering is suspected and confirmed, the System Owner will be charged for these costs.

7.1.6 **PDP Performance Exemptions**

The PDP is responsible for meeting the above noted program requirements and for consistently posting performance data in accordance with the Program Administrator's scheduling and data posting requirements. At its discretion, the Program Administrator may grant reasonable allowances for occasional issues or technical problems, as well as for large catastrophic events such as earthquakes.

7.1.7 PDP Non-Performance

The Program Administrator will not issue incentive payments to customers based on estimated data from the PDP, nor will the Program Administrator estimate incentive payments under any circumstances. It is the PDP's responsibility to ensure timely and accurate posting of validated performance data so customer incentive payments can be made. Performance data also includes fuel consumption and useful thermal output data as this information will be used to verify compliance with program rules and impact PBI payments.

The following conditions may result in penalties, suspension of activity, or revocation of PDP approval from the Program Administrator:

- Data not posted by specified date
- Data not validated in accordance with program requirements
- Estimated data posted instead of actual data
- Meter change information not reported within 30 days of the meter change.
- If an audit or investigation shows a discrepancy of ± 5% between the PDP reported data and Program Administrator check meter production data for one data report period. This discrepancy will trigger an audit schedule set by the Program Administrator for the PDP.

The PDP will be given reasonable opportunity to correct problems identified by the Program Administrator. The Program Administrator will work with the PDP to correct any such problems and avoid unnecessary delays in issuing incentive payments to customers, to the extent feasible.

Upon receipt of a non-performance notice from the PA, the PDP must, as soon as reasonably practicable:

- 1. perform a root-cause analysis to identify the cause of such a failure;
- 2. provide the PA with a report detailing the cause of, and procedure for correcting such failure within 3 days of completion of such root-cause analysis;
- 3. Implement such procedure after obtaining the respective PA approval of such procedure.

PDP Providers that fail to submit data to the Program Administrators when requested by the PA or an authorized agent of the CPUC may be removed as an eligible PDP from the Program Administrators' approved list. It is the Host Customer and/or System Owner's responsibility to ensure the transfer of production data from the Performance Data Providers (PDP) to the Program Administrators. The System Owner is responsible for resolving any issues relative to PBI and PDP performance data.

Should the PDP disagree with a PA decision regarding a penalty, the PDP has the right to appeal to the SGIP Working Group for further consideration.

7.1.8 Data Retention

Monthly performance data must be retained in accordance with program requirements. The PDP must be prepared to post historical interval data at the Program Administrator's request. The Program Administrator audit will include raw interval data, which is to be maintained by the PDP for comparison with validated interval data transmitted to the Program Administrator. The PDP is also responsible for providing backup and disaster recovery services for 100% of the data.

7.1.9 **Technical and Customer Support**

The PDP must provide a technical support number to the Program Administrator for use during normal business hours (8am to 5pm Pacific time, Monday through Friday, except holidays) to help resolve any data availability, format or corruption issues, communication problems, server access problems, or other technical issues. Within those normal business hours, the PDP must respond to Program Administrator requests within two business days with a status report and plan for correcting the issues. The PDP must also provide a customer support number to respond to customer inquiries within two business days from the initial customer contact. Program Administrators will have the discretion to set deadlines for the resolution of data transfer problems/issues.

7.1.10 **Program Administrator Liability**

Apart from the requirements identified herein, the PAs are not liable for the performance or non-performance of a PDP that may result in a delay of or incorrect amount of a PBI payment. The Program Handbook defines the criteria required for PDPs to participate in the Program only.

7.2 **PDP Application Process**

Any entity may choose to become a Performance Data Provider for the SGIP. Providers interested in becoming a PDP for the SGIP must submit information detailing their qualifications to become a PDP for the program. All PDPs must meet the requirements established herein in addition to the requirements set forth in the other metering sections of the SGIP Handbook. Interested parties may apply to qualify as a PDP for the SGIP program at any point in time.

To apply to qualify as a PDP, the Applicant completes the SGIP PDP Application and provides all documentation in the attached questionnaire ⁹². Note that the PDP Applicant may submit one application for statewide PDP services to any of the Program Administrators. The Program Administrator will review the submitted documentation, determine if the PDP Applicant meets the program requirements and send the PDP Applicant a conditional approval letter if all requirements are met. Upon conditional PDP approval, the Applicant may contact the SGIP online application database provider to set up an account for the PDP Upload Portal.

7.2.1 Data Transfer Test

Once the prospective PDP has accumulated a month worth of data for the first SGIP project they must contact the SGIP online application database provider to schedule a data transfer test. The prospective PDP must create, format and validate Application Interval and Meter Interval Reports for all types of data services they are applying for (electrical, thermal and fuel, and/or energy storage) and submit the test files to the SGIP online application database via the PDP Upload Portal.

The Program Administrator will check the test files to ensure they comply with the SGIP guidelines. If the PDP Applicant fails the data transfer test and report approval, they will be given 2 weeks to resolve any technical or data format issues. Upon successful completion of the PDP data test procedures, the PDP will receive a final approval notice and will be qualified to provide PBI data to the Program Administrator for incentive payment.

7.3 **Data Privacy and Security**

Protecting the privacy of System Owners and Host Customer is of the highest order. As such, data shall be collected, processed, and reported by the PDP to the System Owner and the Program Administrator in accordance with this section. The PDP is responsible to ensure timely, consistent and accurate reporting of performance data. Data must be located in a secure facility, on a secure server and have firewall and equivalent protection. The PDP must protect the confidentiality of the customer information and performance data in accordance with all program guidelines. The PDP must also follow all applicable state and federal privacy and data security laws.

⁹² Please refer to The PDP Specification and the Program Administrators' websites for the PDP Application and detailed application instructions.

The PDP may provide data to third parties, including Contractors and Host Customers (if different than the System Owners), provided the System Owner has consented in writing to the release of such performance data. Electricity, thermal and fuel meters shall be kept secure from Denial of Service (DOS) Attacks, Port Scanning, Unauthorized Access and other security violations. To achieve this security, Communications Interfaces to all meters must be located in a physically secure location and include strong password protection with either a network firewall or encrypted connection to limit the meter's network access to the PDP and/or a defined list of authorized users. In addition, security measures may be implemented as needed to ensure data security including restriction of direct meter access for real time data to sequential access basis.

7.4 Measurement & Evaluation (M&E) Activities

As a condition of receiving incentive payments under the SGIP, System Owners and Host Customers agree to provide full access to Site and system equipment and participate in Measurement and Evaluation (M&E) activities as required by the CPUC for five-years. M&E activities will be performed by the Program Administrator (PA) or the PA's independent third-party consultant and include, but are not limited to, periodic telephone interviews, Site visits, development of a M&E Monitoring Plan, review of monitoring plans developed by the project Developer or host Site, installation of metering equipment or review/inspection of metering equipment installed by the project Developer or host Site, collection and transfer of data from installed system monitoring equipment, whether installed by Host Customer, System Owner, a third party, or the PA. This data will be used to show the performance of technologies by class (e.g. wind turbines) and may determine the performance of those technology classes as they see fit. Performance data from specific projects, however, will remain confidential.

7.4.1 M&E Field Visits

During the course of the Project, the PA or the PA's independent third-party consultant may require one or more visits to the Site for M&E purposes. These site visits can occur before, during or after startup of the system for the purposes of developing a monitoring plan, installing additional M&E instrumentation, performing equipment operations inspection and retrieving system data. These visits are separate and distinct from the Post-Installation Inspection visits by the PA or its consultants (see Section 2.5.3), which are used to determine eligibility of the installed system and occur during the Incentive Claim stage of the application process.

7.4.2 **M&E Metering Requirements**

All SGIP systems require installation of metering devices to measure and record electrical output or offset, waste heat, and fuel consumption for M&E purposes. The PAs may collect this information from the data submitted by the Performance Data Providers (PDP) for projects subject to PBI payments. For non-PBI projects, the PA or the PA's independent third-party consultant may install meters to collect M&E data at the Program's expense.

The Host Customer and System Owner agree to provide system monitoring data (15-minute interval data for electrical and thermal and hourly for fuel consumption) to the SGIP M&E consultant on a monthly basis for the duration of five years.

7.4.3 Disposition of SGIP Metering Equipment

Upon completion of the SGIP M&E metering activities at the Site, the Program Administrator will offer all M&E metering equipment installed by the Program to the System Owner for transference. The Program Administrator will provide an Equipment Transfer Agreement with a schedule of the SGIP M&E equipment located at the Site. The Equipment Transfer Agreement must be signed by both the System Owner and the Program Administrator. If the System Owner does not wish to accept the M&E metering equipment, the Program Administrator or its independent third-party consultant will remove the M&E metering equipment. The Program Administrator shall pay the costs for meter removal. Dispute Resolution and Infractions

8 Dispute Resolution

All participants shall attempt in good faith to resolve any dispute arising out of, or relating to, this transaction promptly by negotiations between the Program Administrator or his or her designated representative and the Host Customer, System Owner and/or Applicant or their designees. Either party must give the other party, or parties, written notice of any dispute. Within thirty (30) calendar days after delivery of the notice, the parties shall meet, and attempt to resolve the dispute. If the matter has not been resolved within thirty (30) calendar days of the first meeting, any party may pursue other remedies including mediation. All negotiations and any mediation conducted pursuant to this clause are confidential and shall be treated as compromise and settlement negotiations, to which Section 1152.5 of the California Evidence Code shall apply. Notwithstanding the foregoing provisions, a party may seek a preliminary injunction or other provisional judicial remedy if in its judgment such action is necessary to avoid irreparable damage or to preserve the status quo. Each party is required to continue to perform its obligations under this Contract pending final resolution of any dispute arising out of, or relating to, this Contract.

9 Participant Performance and Infractions

9.1 Participant Performance

All participants⁹³ are expected to follow program rules and eligibility requirements. Failure to do so will result in warnings and/or infractions. Please see *Section 9.2* for additional information regarding warnings and infractions. Program Administrators will exercise their judgment in issuing warnings and assessing infractions. In an effort to ensure participant performance, all participants will be required to maintain a high level of performance in each of the following categories:

- Application
- Inspection
- Attrition and Extensions
- Data Reporting
- SGIP Online Application Database Operation
- Developer

9.1.1 **Application**

Applications must be submitted with complete and accurate documentation and must meet all deadlines. Applications with incomplete, falsified, or inaccurate documentation⁹⁴ or that do not meet required due dates may receive warnings and/or result in an infraction.

9.1.2 *Inspection*

All projects may be inspected at any time during project's permanency period and are expected to meet document and operational requirements of the program. Failure to do so will count as a failed inspection. Participants with a high statewide inspection failure rate may forfeit future participation and/or all active applications/incentive.

9.1.3 Attrition and Extensions

Participants are expected to submit committed projects. Cancelled withdrawn applications will be counted towards attrition rate. Participants with a high attrition rate may forfeit future participation and/or all active applications/incentives.

Participants are also expected to meet project milestone due dates as originally assigned. Excessive extension requests among a participant's applications may receive warnings and/or result in an infraction.

⁹³ For the purpose of this section, participant is defined by an entity or group of entities submitting applications, data, or developing and/or installing SGIP projects.

⁹⁴ Including criteria that would grant higher priority in the event of a lottery.

9.1.4 Data Reporting

Participants are required to submit all necessary operational and performance data. Participants submitting data past their designated due date may receive warnings. Participants that do not submit data at all for their projects may be subject to an infraction. Any falsified or blank submissions will result in an automatic infraction.

9.1.5 SGIP Online Application Database Operation

Participants are required to comply with the Terms of Use of the SGIP database for all accounts, applications, and PBI data submissions. Non-compliance with the Terms of Use or attempting to circumvent the SGIP database application policies or procedures will result in an automatic infraction.

9.1.6 **Developer**

An infraction may be issued if the Developer of a project does not fully and/or accurately disclose Developer and ownership information as listed in *Section 4.1.5*.

9.1.7 Residential GHG Emissions Compliance

Developers of non-IOU residential customers have a continuing obligation to produce documentation regarding participating customers' systems, rates and charge and discharge patterns upon request. Failure to do so shall be considered an infraction. Developers must also provide the PAs and SGIP evaluator with documentation on participating customers' systems upon request. Failure to do so shall be considered an infraction.

Additionally, the PAs will annually verify that new residential fleets are reducing GHG emissions using annual statistical sampling within the SGIP impact evaluation process and, if not, to take appropriate enforcement actions. The Commission adopts four additional requirements for new residential fleet verification:

- PAs may issue infractions for any new residential developer that does not provide the information requested by the PAs or the SGIP Impact Evaluator in the timeframe requested and to suspend any developer with two infractions for this reason for applying for new incentives for a period of six months.
- PAs will suspend any developer with a new residential fleet verified as having increased GHG
 emissions for two successive six-month periods from applying for new SGIP incentives for six
 months, or until the developer's fleet is verified to be reducing GHG emissions, whichever is later.
- 3. PAs will require any developer verified as having increased GHG emissions to biannually submit GHG emissions data for all projects in the developer's new residential fleet using the existing SGIP online data upload portal until the fleet is verified to reduce GHG emissions
- 4. PAs may suspend any new residential developer verified as having increased GHG emissions for submitting data for less than 90% of the meters of the developer's fleet from applying for any additional incentives for a period of one year.

9.2 Infractions

Infractions are any actions that circumvent program policy or requirements, or have the intent to do so, in addition to low performance levels. Infractions can be issued to any participant, as defined in *Section 4.1*. The Program Administrators or Energy Division will evaluate program infractions, which may include gross negligence or intentional submission of inaccurate project information. Program infractions may be determined at any stage of the SGIP process and are applicable statewide. If it is determined that a program infraction has been committed, a reasonable sanction shall be imposed. The following sanctions may be applied:

- Suspension or expulsion from future program participation
- Cancellation of existing projects
- Application fee forfeiture
- Fiscal or programmatic audit

9.2.1 Issuance of Warnings and Infractions

If a Program Administrator determines that an infraction may be warranted, a notice will be sent to the violating participant. Participants may be issued one or more warnings before being issued an infraction; however, serious violations may result in an immediate infraction. Participants may receive no more than three warnings before an infraction is issued. Infractions will be reviewed by all SGIP Program Administrators and will be communicated to the participant. Notwithstanding the above, Energy Division may initiate an investigation of potential infractions, which may result in a penalty for any infraction determined to have occurred.

Definitions and Glossary

Amp-hour Capacity or Nominal Capacity (Ah for a specific C-rate): The total Amp-hours available when

the battery is discharged at a continuous current over a specified period of time (specified as a C-rate) from

100 percent state-of-charge to a specified cut-off voltage.

Applicant: The entity, either the Host Customer, System Owner, or third party designated by the Host

Customer responsible for the development and submission of the SGIP application materials. Functions as

the main point of communication between the SGIP Program Administrator for a specific SGIP Application.

Application Interval Report: Monthly meter data report provided by the Performance Data Provider

(PDP). The Application Interval Report format will be the same for all PDPs and all Applications as specified

by the Program Administrators. The data in this file will be will be validated upon upload and used for the

calculation of Annual PBI payments. An Entry in this file shall represent the combined data readings of all

meters associated with a particular SGIP Application Code.

Backup Generators: Operate as short-term temporary replacement for electrical power during periods of

Electric Utility power outages. In addition to emergency operation they ordinarily only operate for testing

and maintenance. Backup generators do not produce power to be sold or otherwise supplied to the grid or

provide power to loads that are simultaneously serviced by the Electric Utility grid. Backup generators only

service customer loads that are isolated from the grid either by design or by manual or automatic transfer

switch.

Blended Fuel: A blended fuel is a combination of any renewable fuel with a non-renewable fuel and where

the amount of renewable fuel is less than 100%. Projects using less than 100% of an eligible renewable

fuel will be identified as a blended project equivalent to their renewable fuel percentage.

California Manufacturer: A California Manufacturer operates a manufacturing facility in California, is

licensed to conduct business in California and is registered with a primary or secondary manufacturing

NAICS code. Equipment is deemed to be manufactured in California if at least 50% of the value of the

capital equipment has been made in a dedicated production line by an approved California Manufacturer.

Commercial: Commercial entities are defined as non-manufacturing business establishments, including

hotels, motels, restaurants, wholesale businesses, retail stores, and health, social, and educational

institutions. For the purpose of SGIP, commercial sectors include agricultural and industrial customers.

CSE: Center for Sustainable Energy®

CEC: California Energy Commission

CPUC: California Public Utilities Commission

Developer: For applications received in Steps 1 and 2, a Developer is the corporate entity that holds the

contract for purchase and installation of the system, and/or alternative System Ownership Agreement (such

118

as a Power Purchase Agreement) with the host customer and handles the project's development activities. For applications received in Step 3 and later, a Developer is, if not individual homeowners applying for SGIP incentives for systems located on their own property, the corporate entity registered and in good standing with the Secretary of State of California that handles a substantial amount of the project's development activities.

Developer Fleet: Ten or more projects comprise a developer fleet. For compliance purposes, a developer's legacy commercial fleet includes all legacy commercial projects within their ten-year permanency requirement, whose SGIP agreements list the same developer. A developer's legacy residential fleet includes all legacy residential projects whose SGIP agreements list the same developer. A developer's new commercial fleet includes all new commercial projects within their ten-year permanency requirement, whose SGIP agreements list the same developer. A developer's new residential fleet includes all new residential projects within their ten-year permanency requirement whose SGIP agreements list the same developer.

Directed: A directed renewable fuel is produced and captured at a different location than the project site of the electrical generation facility. The renewable fuel is delivered to the facility through a common carrier pipeline, which must be demonstrated through a procurement contract. A directed renewable fuel must be injected into a common carrier pipeline system that is either within the Western Electricity Coordinating Council (WECC) region or interconnected to a common carrier pipeline system located within the WECC region.

Disadvantaged Community: Any census tract that ranks in the statewide top 25% most affected census tracts in the most recently released version of the environmental health screening tool, CalEnviroScreen, plus those census tracts that score within the highest 5% of CalEnviroScreen's pollution burden, but do not receive an overall CalEnviroScreen score.

Discrete PSPS Event: For public safety, electric utilities may deem it necessary to turn off electricity when winds and dry conditions, combined with a heightened fire risk, are forecasted. The utility may need to deenergize the grid for some number of hours or even days until the weather event has passed and it is safe to restore power. This is a Public Safety Power Shutoff (PSPS).

For PG&E and SCE Customers: For the purposes of SGIP, if the utility de-energizes a customer for safety and then restores power after the weather event has passed, this would count as one PSPS event – whether that PSPS event endured for the customer for only a few hours or some number of days. If power is restored for the customer and another weather event subsequently requires that the utility de-energize the same customer again – whether this occurred days, weeks or months later – this would count as the customer's second PSPS event. PSPS events are counted per calendar year.

In non-IOU territories, the applicable electric provider's definition of a discrete PSPS event will be used for SGIP eligibility purposes.

For SDG&E Customers: A discrete PSPS event is defined as an event with 72 hours or more between concurrent weather events, similar to a storm series.

Education Institution: Any institution that would otherwise be eligible for funding through the California Clean Energy Jobs Act (Proposition 39), or a college or university accredited to operate in California.

Electric Utility: The Host Customer's local electric transmission and distribution service provider for their site.

Energy Capacity or Nominal Energy for DC/AC systems (Wh (for a specific C-rate)): The total watthours available when a storage system is discharged at a continuous current over a specified period of time (specified as a C-rate) from 100 percent state-of-charge to the cut-off voltage.

Energy Storage: Technologies able to store energy that can be discharged as useful energy at another time in order to directly supply electricity or offset electricity consumption. Unless specified otherwise, "energy storage" in the SGIP Handbook applies to all eligible storage technologies, including mechanical, electrochemical, or thermal energy storage.

Energy Storage Paired with and Charging from an On-site Renewable Generator: Energy storage system that is paired with an on-site generator and charges at a minimum 75% from the generator.

ESCO: Energy Service Company (ESCO), a business entity that designs, builds, develops, owns, operates or any combination thereof self-generation Projects for the sake of providing energy or energy services to a Host Customer.

Fuel Cell: Power plants that produce electricity through an electrochemical reaction with a fuel source.

Gas Service: The gas line from the Utility's distribution main to the serving gas meter.

GHG signal: A digitally accessible data feed of current marginal greenhouse gas emissions rates (in units of kg/kWh) that updates at regular intervals (e.g. every five minutes) combined with additional data feeds that deliver regularly updated forecasts of grid conditions for use in the optimization of dispatch.

HVAC-integrated S-TES: Small thermal energy storage systems integrated to offset peak energy consumption of direct expansion refrigerant based air conditioning units less than or equal to 20 tons.

Host Customer: An entity that meets all of the following criteria: 1) has legal rights to occupy the site, 2) receives retail level electric or gas distribution service from PG&E, SCE, SoCal Gas or SDG&E, 3) is the utility customer of record at the site 4) is connected to the electric grid, and 5) is the recipient of the net electricity generated from the self-generation equipment. Multifamily buildings that are installing a system on behalf of tenants and are enrolled in a VNEM tariff also qualify.

Inspection: An inspection can be conducted as an on-site Post-Installation Inspection by the PAs or their third-party consultant or as a virtual post-installation inspection executed by the Developer, System Owner or Host Customer. For a description of the post-installation inspection protocol, please refer to the Energy Storage Post-Installation Inspection and Discharge Testing Protocol and SGIP Inspection Sampling Protocol at www.selfgenca.com. In addition, general inspections can be conducted at any point during the permanency period.

Investor Owned Utility: For purposes of the SGIP, this refers to Pacific Gas & Electric Company, San Diego Gas & Electric Company, Southern California Edison Company and Southern California Gas Company.

ISO: International Standards Organization

ITC: Investment Tax Credit

Legacy Projects: Any project for which a completed SGIP reservation request was submitted before the GHG rules outlined in D 19-08-001 became effective.

Local Governmental Agency: Any entity described by Public Contracts Code section 22161 (f).

Low-Income Community: Census tracts with median household incomes at or below 80% of the statewide median income or with median household incomes at or below the threshold designated as low-income by the Department of Housing and Community Development's list of state income limits adopted pursuant to Section 50093.

Low-Income Residential Customer: An individual customer living in a low-income residence, as described in subparagraph (C) of paragraph (3) of subdivision (a) of section 2852 of the Public Utilities Code.

Low-Income Residential Housing: Multifamily residential building of at least five rental housing units that is operated to provide deed-restricted low-income residential housing, as described in clause (i) of subparagraph (A) of paragraph (3) of subdivision (a) of section 2852 of the Public Utilities Code, and is either: 1) in a disadvantaged community; or 2) a building where at least 80% of the households have incomes at or below 60% of the area median income, as defined in subdivision (f) of section 50052.5 of the Health and Safety Code.

Meter Interval Report: Monthly meter data report provided by the Performance Data Provider (PDP). The Meter Interval Report format will vary by PDP. The PDPs shall format their report so that it includes all meter and sensor data recorded over the measured interval at a minimum 15-minute frequency, with the exception of Fuel Consumption, which shall be measured and reported hourly. This report should be formatted as specified by SGIP program requirements.

New Projects: Any project submitting a complete SGIP application containing all required information on or after April 1, 2020 and any project submitting an SGIP application prior to April 1, 2020 that is not

complete and for which all required information is not provided until or on April 1, 2020 that receives SGIP incentives. GHG emissions reductions requirements apply throughout a project's ten-year permanency period. The phrase "complete application" refers to the "application submitted" date not the "incentive claim form submitted" date.

Non-Residential: Any project for which the customer is not residential. This includes eligible agriculture, commercial, and industry customers.

Non-Profit Organization: An organization registered and in good standing with the California Secretary of State as a domestic non-profit entity.

Non-Renewable Fuel: A non-renewable fuel includes fossil fuels and synthetic fuels. Synthetic fuels are fuels derived from materials that are not renewable or fossil fuels. SGIP eligible fossil fuels are gasoline, natural gas and propane. SGIP eligible synthetic fuels include, but are not limited to, the direct use or synthesis of fuels sewage sludge, industrial waste, medical waste or hazardous waste.

On-Site: An on-site renewable fuel is produced and captured at the same location as the site of the electrical generation facility. Additionally, the renewable fuel is delivered from the source to the generating system via a dedicated pipeline. A dedicated pipeline is defined as only physically capable of delivering gas to the generating facility.

Paired: Two or more technologies located on the same electrical circuit and behind the same utility electrical meter.

Parallel Operation: The simultaneous operation of a self-generator with power delivered to or received by the Electrical Utility while interconnected to the grid. Parallel Operation includes only those generators that are interconnected with the Electric Utility distribution system for more than 60 cycles.

PDP: Performance Data Provider. A company that contracts with the SGIP Participant to read and communicate their metering data to the Program Administrators.

PG&E: Pacific Gas and Electric Company

Power Purchase Agreements: An agreement for the sale of electricity from one party to another, where the electricity is generated and consumed on the Host Customer site. Agreements that entail the export and sale of electricity from the Host Customer site do not constitute Host Customer's use of the generated electricity and therefore are ineligible for the SGIP.

Pressure Reduction Turbine: If a facility with a high-pressure fluid (e.g., steam, water, natural gas, etc.) distribution network and pressure reduction valve(s), installs a turbine to replace or operate in parallel with these pressure reduction valve(s), then the application is considered a Pressure Reduction Turbine.

Program Year: A project's program year is the year its incentive application was accepted by the Program Administrator.

Project: For purposes of the SGIP, the "Project" is the installation and operation of the proposed eligible self-generation technology(ies) at a specific site, as described by the submitted Reservation Request documentation.

Public Entity: Includes the United States, the state and any county, city, public corporation, or public district of the state, and any department, entity, agency, or authority of any thereof.⁹⁵

Rated energy capacity (kWh): The rated energy capacity (kWh) for DC/AC energy storage technologies as the nominal voltage multiplied by the amp-hour capacity multiplied by the applicable efficiency (VDC x Amp-Hours x Applicable Efficiency).

Refrigeration TES: Thermal energy storage systems integrated to offset peak energy consumption of direct expansion refrigerant-based refrigeration systems.

Renewable Fuel: A renewable fuel is a non-fossil fuel categorized as one of the following: biodiesel or gas derived from digester gas, landfill gas or biomass. SGIP projects can use one or more eligible renewable energy sources, as identified by the Renewable Portfolio Standard (RPS).

Reservation Expiration Date: The Reservation Expiration Date is the date the Incentive Reservation expires and all required documentation must be provided by.

Round Trip Efficiency (RTE): The total kWh discharge of the system divided by the total kWh charge over some period of time or number of cycles.

Single-cycle roundtrip efficiency (SCRTE): The total kWh discharge of the system divided by the total kWh charge after one complete cycle. SCRTE is often verified in the factory and specified on a device's technical specifications sheet.

SCE: Southern California Edison

SDG&E: San Diego Gas and Electric

Single Business Enterprise: For purposes of defining a site, a Single Business Enterprise is a business that has a unique taxpayer or employer identification number. Two or more businesses with the same taxpayer or employer identification number, as a group, are a Single Business Enterprise.

Site: A Single Business Enterprise or home located on an integral parcel or parcels of land undivided by a public road or thoroughfare regardless of the number of meters serving that site; or if divided by a public road or thoroughfare, served by a single Electric Utility meter. Separate business enterprises or homes on a single parcel of land undivided by a highway, public road, thoroughfare or railroad would be considered for purposes of the SGIP as separate sites.

⁹⁵ Source: CALIFORNIA CODES - PUBLIC CONTRACT CODE, SECTION 21611

Small Business: A business or manufacturer, including affiliates, with average annual gross receipts of \$15 million or less, over the last three tax years.

SoCalGas: Southern California Gas Company

Stand-alone Energy Storage: An energy storage system located on a Host Customer project site that does not also host a generating technology.

State Governmental Agency: Any entity described by Government Code section 11000.

Steam Turbine CHP: If a facility with a steam distribution network installs a steam turbine that is placed between the boiler and the steam distribution network, and steam pressure is increased, then the application is considered a Steam Turbine CHP. Pressure can be increased either through the purchase of new boilers or by increasing the pressure on existing boilers. In either case, boiler fuel usage per unit of steam production increases with boiler pressure.

System Owner: The owner of the SGIP system at the time the incentive is paid. For example, in the case when a vendor sells a turnkey system to a Host Customer, the Host Customer is the System Owner. In the case of a leased system, the lessor is the System Owner.

TES: Thermal Energy Storage

Thermal Energy Storage: Technologies able to store energy and discharge it at a later time as thermal energy to offset electricity consumption.

Thermal Load: Host Customer heating process(es) including but not limited to industrial process heating, space heating, domestic hot water heating and/or heat input to an absorption chiller used for space cooling or refrigeration.

Thermal Load Equipment: Thermal end-use equipment such as but not limited to absorption chillers (indirect or direct fired), boilers, water heaters, space heaters, furnaces, dryers, secondary heat exchangers, thermal storage tanks or vessels including pumps, cooling towers, and piping or any other ancillary equipment.

Waste Gas: Waste Gas fuels used for conventional CHP technologies and Fuel Cells are strictly defined as natural gas that is generated as a byproduct of petroleum production operations and is not eligible for delivery to the utility pipeline system.

Waste Heat to Power: If a facility discarding heat as a result of commercial or industrial processes installs a turbine driven by the discarded heat, then the application is considered a Waste Heat to Power system. These systems typically involve a heat recovery system, which transfers the waste heat to a working fluid which drives the turbine.

Legislation and Regulatory Background

Date	Bill Number	Description
9/6/2000	AB 970	Assembly Bill required the CPUC to initiate load control
		and distributed generation activities.
3/27/2001	D 01-03- 073	CPUC Decision complying with Assembly Bill 970 and establishing the Self Generation Incentive Program. Implementation of PU Code Section 399.15(b), Paragraph 4-7; Load Control and Distributed Generation Initiatives.
06/01/2001	D. 01-06-035	CPUC Decision establishing waste heat recovery standards for SGIP. Requires Energy Branch to develop reliability criteria.
01/18/2002	Letter on Reliability Criteria	CPUC Energy Branch Letter establishing reliability criteria requirements for level 3 technology applications received after January 1, 2002
02/07/2002	D. 02-02-26	CPUC Decision addressing eligibility of customers served by electric municipalities, maximum size and annual program budget.
04/04/2002	D. 02-04-004	CPUC Decision clarifying Applicant's ability to receive incentive funding from multiple sources. Addressing SCAQMD's PTM of Decision 01-03-073
09/19/2002	D. 02-09-051	CPUC Decision adding technology level 3-R, which establishes a new level of incentives. Contains specific requirements for projects using renewable fuels for level 3-R. Addressing Capstone's PTM
10/12/2003	AB 1685	 Extended the SGIP through 2007 Required that projects commencing January 1, 2005 meet a NOx emission standard Required that projects commencing January 1, 2007 meet a more stringent NOx emission standard and a minimum system efficiency standard. Established a NOx emission credit that can be used by combined heat and power (CHP) units to meet minimum system efficiency standard
9/22/2004	AB 1684	Exempts certain projects from NOx emission standards set forth in AB 1685 that meet waste gas fuel and permitting requirements.

Date	Bill Number	Description
12/16/2004	Decision 04-12-045	 Modified SGIP to incorporate provisions of AB 1685: Eliminates maximum percentage payment limits Reduces incentive payments for several technologies Expands opportunities for public input regarding developing a declining incentive schedule, developing an exit strategy and adopting a data release format Required an application fee for all projects received after 1/1/2005 in order to deter against "phantom projects". This requirement was removed beginning in 2007 except in the case of new technologies that are in the process of certification.
1/12/2006	Decision 06-01-047	Established the California Solar Initiative (CSI) and ordered changes in the 2006 SGIP to accommodate the transition of solar program elements to the CSI beginning January 1, 2007.
9/29/2006	AB 2778	 Extended SGIP until January 1, 2012 Limited eligible technologies beginning January 1, 2008 to fuel cells and wind systems that meet emissions standards required under the distributed generation certification program adopted by the State Air Resources Board Requires that eligibility of non-renewable fuel cell projects be determined either by calculating electrical and process heat efficiency according to PU Code 216.6 or by calculating overall electrical efficiency
4/24/2008	Decision 08-04-049	Removed the 1 MW cap on incentives for 2008 and 2009 allowing projects to receive lower incentives on a tiered structure for the portion of a system over 1 MW.
9/28/2008	AB 2267	Requires an additional 20% incentive for the installation of eligible distributed generation resources from a California Supplier. This additional incentive is applied only to the technology portion of the incentive; the additional incentive for renewable fuels is not included in calculating the 20%.
11/21/2008	Decision 08-11-044	 Determined that Advanced Energy Storage systems coupled with eligible SGIP technologies will receive an incentive of \$2/watt of installed capacity. Revises the process for the review of SGIP program modification requests
9/09/2009	Decision 09-09-048	Grants a petition to modify SGIP policies expanding eligibility for Level 2 incentives to include "directed biogas" projects where renewable fuel is nominated via contract.

Date	Bill Number	Description
2/25/2010	Decision 10-02-017	 Revises Decision 08-11-044 so that Advanced Energy Storage systems coupled with fuel cells must meet the site specific requirements for on-site peak demand reduction and be capable of discharging fully at least once per day in order to be eligible for the \$2/watt incentive from the self-generation incentive program. Determines that Advanced Energy Storage systems coupled with eligible technologies under the SGIP must install metering equipment capable of measuring and recording interval data on generation output and Advanced Energy Storage system charging and discharging.

00/00/00/4	00110 0 11 00 015	
09/08/2011	CPUC D.11-09-015	 Adds eligibility requirements based upon greenhouse gas reductions. Establishes an on-site emission rate that projects
		must beat to be eligible for SGIP participation of 379 kg CO2/MWh.
		 Adds Waste Heat to Power, Pressure Reduction Turbine, Internal Combustion Engine – CHP,
		Microturbine – CHP, Gas Turbine – CHP, Stand-
		Alone AES technologies to the list of eligible technologies.
		Revises the incentive rates for all technologies and adds a \$2.00/Watt biogas adder.
		Directs that Directed Biogas can only be procured
		from in-state suppliers. • Eliminates maximum size restrictions given a project
		meets on-site load. Sets a 30 kW minimum for wind and renewable fueled fuel cell projects.
		Adopts a hybrid payment structure with 50% upfront,
		50% PBI based on kWh generation of on-site load for projects 30 kW and larger. Projects under 30 kW
		will receive the entire incentive upfront.Adopts the following assumed capacity factors to be
		used in PBI calculations: 10% for AES, 25% for wind,
		and 80% for all other distributed energy resources.Implements incentive decline in the following manner
		10% per year for emerging technologies and 5% per year for all other technologies, beginning 1/1/2013.
		Adopts a supplier concentration limit where no more than 40% of the annual statewide budget available
		on the first of a given year may be allocated to any
		single manufacturer's technology during that year.Establishes a maximum project incentive of \$5
		million. • Establishes that the minimum customer investment
		in a project must be 40% of eligible project costs.
		• Establishes an SGIP incentive budget allocation of 75% for renewable and emerging technologies, and
		25% for non-renewable technologies.Determines that the Program Administration Budget
		will be reduced to 7%. • Establishes that projects exporting to the grid are
		eligible for SGIP incentives as long as they do not
		export more than 25% on an annual net basis.Makes an energy efficiency audit mandatory for
		participation in SGIP unless an extensive audit has been conducted within five years of the date of the
		reservation request.
		Establishes an application fee that is 1% of the amount of incentive requested

Date	Bill Number	Description
		Limits all projects to one six month extension. Request for a second extension may be made to the Working Group. Extends the warranty period to 10 years
	ADVICE LETTER 4410-G	ADVICE LETTER COMPLYING WITH RESOLUTION E- 4519Proposed Amendments to the Self-Generation Incentive Program Handbook to Conform to Resolution E-4519. Changes to the RTE for AES technologies and elimination of certain data formatting requirements for PDP providers
	ADVICE LETTER No. 3253-G/3940 –E	Proposed Revisions to the Self-Generation Incentive Program Handbook to Implement Decision (D.) 11- 09-015: Implementation of the Hybrid-Performance- Based Incentive Payment Structure; Metering and Monitoring Protocols; Other Amendments.
	ADVICE LETTER No 3253-G-A/3940-E-A	Supplemental Filing: Proposed Revisions to the Self-Generation Incentive Program Handbook to Implement Decision (D.) 11-09-015: Implementation of the Hybrid-Performance-Based Incentive Payment Structure; Metering and Monitoring Protocols; Other Amendments
5/24/2012	Decision 12-05-037	Orders that all technologies previously eligible for the Emerging Renewables Program should be immediately eligible for the SGIP Determines that consolidating the ERP and SGIP programs now is preferable to perpetuating two competing programs that serve the same types of technologies and policy purposes

Date	Bill Number	Description
5/20/2014	SB 861	 Extended SGIP funding through 2019 and extended SGIP administration until January 1, 2021 Directed the Commission to update the factor for avoided greenhouse gas emissions based on the most recent data available to the State Air Resources Board Established eligibility requirements for distributed technologies that: reduce demand from the grid by offsetting some or all of the customer's onsite load, are commercially available, safely utilize the existing T&D system, and improve air quality by reducing criteria air pollutants Specified that SGIP incentive recipients are subject to data collection and site inspections upon request Directed the Commission to develop a capacity factor for each technology in the SGIP Directed the Commission to consider the cost of greenhouse gas emissions reductions, peak demand reductions, system reliability benefits, and other measurable factors when allocating program funds between eligible technologies Change the California supplier requirement to "manufactured in California" Specified that the SGIP will be evaluated on the following performance measures: reductions of GHGs, reductions of air pollutants, amount of energy reductions measured in energy value, reductions of customer peak demand, capacity factor, value to T&D system measured in avoided cost of upgrades and replacement, ability to improve onsite electricity reliability
09/27/2014	AB 1478	 Clarified that eligible technologies can shift onsite energy use to off-peak times
12/18/2014	Decision 14-12-033	Decision authorizing Pacific Gas and Electric Company, Southern California Edison Company, San Diego Gas & Electric Company, and Southern California Gas Company to continue to collect funds for the Self-Generation Incentive Program pursuant to Public Utilities Code Section 379.6 as amended by Senate Bill 861
	ADVICE LETTER 47-A	 Advice Letter to propose modifications to the SGIP Handbook to include revised capacity rating methodologies for Pressure Reduction Turbine (PRT) and Waste Heat to Power (WHP) technologies Included conventional topping cycle Steam Turbines in the program.

Date	Bill Number	Description
6/11/2015	Resolution E-4717	Approval of Advice Letter No. PGE 3552-G/4563-E, CSE 55, SCE 3165-E, SCG 4741, filed January 20, 2015, to incorporate Residential AES Operational Requirements Affidavit into the SGIP Handbook.
6/11/2015	D.15-06-002	 Approval of Program Administrator's Petition for Modification to allow a maximum of three six-month extensions, filed November 13, 2014.
7/1/2015	Advice Letter No. CSE 60, PG&E 4663-E, SCE 3242-E, SCG 4828.	Compliance Advice Letter to incorporate third six- month extension into the SGIP Handbook.
7/17/2015	Disposition Letter for Advice Letter CSE 56	Approval of CSE Advice Letter 56, filed on February 17, 2015, to incorporate kW, kWh offset methodologies, and other associated changes for HVAC-Integrated Small TES systems.
11/19/2015	D.15-11-027	 Decision Revising the GHG Emissions Factor to Determine Eligibility to Participate in the SGIP. GHG Emissions Factor revised to 350 kg CO2₂/MWh averaged over 10 years for non-renewable generation technologies. For technologies that are subject to a 1% annual degradation rate, the first-year GHG Emissions Factor for Electric-Only Fuel Cells is set at 334 kg CO2₂/MWh., assuming a 1% annual degradation rate. Storage devices should demonstrate an average round trip efficiency of at least 66.5% over ten years to qualify for SGIP, which is equivalent to a first-year round trip efficiency of 69.6%.
12/16/2015	Advice Letter No. CSE 66, PG&E 4759-E, SCE 3327- E, SCG 4904	 Advice Letter to incorporate kW, kWh offset methodology and other associated changes for refrigeration TES. Advice Letter became effective January 15, 2016.
12/17/2016	D.15-12-027	 Decision Partially Suspending Disbursement of 2016 Program Year Funds and Acceptance of New Applications for the Self-Generation Incentive Program. Half of 2016's program year funds are available to fund new applications at the beginning of the year. The PAs shall not disburse any additional funds authorized for program year 2016 until further ordered by the Commission.
12/21/2015	Advice Letter No. PG&E 3663-G/4763- E, CSE 67, SCE 3331-E, SCG 4907	 Compliance Advice Letter incorporating the GHG Emissions Factor from D.15-11-027 into the SGIP Handbook. Advice Letter became effective January 1, 2016.
6/23/2016	D. 16-06-055	Decision Revising the SGIP Pursuant to SB 861, AB 1478, and Implementing Other Changes

Date	Bill Number	Description
		 Divided SGIP budget 75% storage, 25% generation Created renewable and small residential storage carve-outs.
		 Revised incentive rates for all SGIP technologies. Made several substantial changes to the SGIP. Resolved several petitions for modifications regarding Distributed Wind Energy Association, pressure reduction turbines and biogas incentive, PowerTree's request to extent deadlines, and Maas Energy's request to investigate the February 23, 2016 program opening.
4/6/2017	D.17-04-017	 Decision revising the SGIP pursuant to AB 1637 and CALSEIA's Petition for Modification Authorized doubling authorized SGIP collections for years 2017-2019
10/12/2017	D.17-10-004	 Decision establishing Equity Budget for Self-Generation Incentive Program Beginning in Step 3, 25% of the funds collected for energy storage projects shall be reserved for the SGIP Equity Budget, which will be awarded only to projects that meet specific criteria.
08/01/2019	D.19-08-001	 Decision establishing a Greenhouse Gas signal and compliance rules for legacy and new projects. Established PBI for all non-residential projects.
9/19/2019	D.19-09-027	Decision establishing new rates for Equity projects, establishing:
1/16/2020	D. 20-01-021	Decision revising SGIP pursuant to SB 700.

Appendix A - System Calculation Example

Efficiency Calculations

Example #1: Efficiency Calculations for 255 kW IC Engine CHP System

Three 85 kW internal combustion engines operating on natural gas are proposed to provide electricity and heat to a hospital. The internal combustion engines are sized such that they will operate at close to full load most of the year. Their output will be reduced in July and August so that the recovered waste heat does not exceed the thermal load. The hospital's Thermal Load consists primarily of domestic hot water and space heating. The Minimum Operating Efficiency Worksheet used for this application is similar to the residential version, but the Thermal Load and Electrical Load per Month must be calculated and justified separately and entered manually for each month. The internal combustion engines exceed the PU Code 216.6. (a) and (b) requirements, therefore they meet the minimum operating efficiency requirement for the program. They also pass the NOx emissions eligibility with CHP credits and pass the GHG emissions eligibility. Their thermal coincidence factor is less than 1.0 for every month of the year indicating that they are utilizing waste heat recovery effectively and since they are qualified for the feed-in-tariff the export factor indicates that they are exporting less than the program export limit which is 25% more than the site electrical load.

Appendix B – Combustion Emission Credit Calculation

Micro-turbine, internal combustion engine, gas Turbine and steam turbine CHP Projects that do not meet the applicable NOx emission standard (.07 lb/MWh) may receive emission credits for waste heat utilization. Credit shall be at the rate of one MWh for each 3.4 million British thermal units (Btu) of heat recovered.

The following formula is used to modify the emissions rating for a generating system by giving credit for waste heat utilization: ⁹⁶

Lb/MWh_{w/credit} = Lb/hr_{EmissionRate} / (MW_{Rated} + MW_{ProcessHeat}) ≡ System emissions with thermal credit Where:

 $Lb/hr_{EmissionRate} = Lb/MWh_{w/o_credit} X MW_{Rated} \equiv NOx$ emission rate at the system's rated capacity

Lb/MWh_{w/o credit} ≡ System's verified emissions without thermal credits

MW_{Rated} ≡ System's Rated Capacity as defined in Section 6.1.5.

MW_{ProcessHeat} = (MMBtu/yr_{UtilizedWasteHeat} / 3.4 MMBtu/MWh) / EFLH/yr ≡ Capacity credit for useful thermal energy

MMBtu/yr_{UtilizedWasteHeat} ≡ Annual utilized waste heat

3.4 MMBtu/MWh ≡ Heat recovered conversion factor

EFLH/yr ≡ System's annual equivalent full load hours of operation

All assumptions, backup documentation, hand calculations, models (with inputs and outputs) and custom spreadsheets used to develop the forecasts must be included in the documentation. Forecasts based solely on "professional experience" or subjective observation will be rejected. Applications must include a completed Waste Heat/AB1685 spreadsheet, available from the Program Administrators' websites, that calculates the waste heat utilization, minimum system efficiency and emissions requirements.

Example #1: Emissions Credit for 360 kW IC Engine Generator

A 360 kW IC engine generator set is proposed to supply electric power and heat to a furniture manufacturing facility. The system utilizes an intercooler chiller that is rated at 10 kW. Its full load fuel consumption is 4.4 MMBtu/hr LHV (4.8 MMBtu/hr HHV⁹⁷) and its full load waste heat recovery rate is 2.6 MMBtu/hr. Source testing documentation for the same generating system make/model and configuration, but from another site, indicate that the NOx emissions from this unit are 0.16 lb/MWh. The generator is fueled with a Non-Renewable fuel and is not a fuel cell. The generator electric output follows the load of the Host Customers facility, but shuts down when the load falls below 40 kW, the minimum load of the generator. The Host Customer annual peak demand is approximately 400 kW. Waste heat from the generating system is used to deliver hot water for manufacturing process, equipment cleanup and space heating. Detailed analysis of the system and Host Customer load reveals that the system will be generating 1,715,000 kWh/yr at a

⁹⁶ Emissions credit calculation is based on the California Air Resources Board, Guidance for the Permitting of Electric Generation Technologies, Appendix D: Quantifying CHP Benefits, July 2002.

⁹⁷ For natural gas, LHV ≈ HHV x 0.9

capacity factor of 56%. The system will produce 12,730 MMBtu/yr of recovered waste heat to serve 12,400 MMBtu/yr of thermal load, however only 8,256 MMBtu/yr of waste heat is actual useful thermal output because of non-coincident monthly load. The system consumes 21,521 MMBtu/yr LHV and 23,673 MMBtu/yr HHV of fuel. Thus -

Minimum Operating Efficiency Requirement

P.U. Code 216.6 (a)

 $8,255,800,000 \text{ [Btu/yr]} / \{(1,715,000 \text{ [kWh/yr]} \times 3,413 \text{ [Btu/kWh]}) + 8,255,800,000 \text{ [Btu/yr]}\} = 58.5\% \ge 5\%$ Passes

P.U. Code 216.6 (b)

 $\{(1,715,000 \text{ [kWh/yr] } \times 3,413 \text{ [Btu/kWh]}) + 0.5 \times 8,255,800,000 \text{ Btu/yr}\} / 21,520,800,000 \text{[Btu/yr]} = 46.4\% \ge 42.5\%$ **Passes**

AB 2778 Minimum Electric Efficiency

 $(360 \text{ [kW] x } 3,414 \text{ [Btu/kWh]}) / 4,831,200 \text{ Btu/hr} = 25.4 \ge 40\% \text{ Fails}$

Air Emissions Requirement

AB 1685 Minimum System Efficiency

 $\{(360 \text{ [kW] x } 3,414 \text{ [Btu/kWh]}) + 2,598,000 \text{ [Btu/hr]}\} / 4,831,200 \text{ Btu/hr} = 79.2 \ge 60\%$ **Passes**

AB 1685 NOx Emissions w/o Waste Heat Credit

0.16 [lb/MWh] ≤ 0.07 lb/MWh NOx *Fails*

AB 1685 NOx Emissions w/ Waste Heat Credit

 $\{0.16 \text{ [lb/MWh] x .} 360 \text{ [MW]} \} / \{.360 \text{ [MW]} + (8,256 \text{ [MMBtu/yr]} / 3.4 \text{ [MMBtu/MWh]}) / 4,900 \text{ EFLH/yr} \} = 0.067 \text{ lb/MWh} \le 0.07 \text{ lb/MWh NOx}$

The Minimum Operating Efficiency worksheet is designed to perform this calculation. Applications must include in their application a completed Minimum Operating Efficiency worksheet, which is available from the Program Administrators' websites.

Appendix C - Conversion of Emissions PPM to Lb/MWH

Procedure for Converting Emission Data to lb/MW-hr

Engines

Engine emission standards are typically expressed in terms of ppmv or in grams/brake horsepower-hour. Given below are factors to convert from ppm to grams/brake horsepower-hour and from grams/brake horsepower-hour to pound/megawatt hour.

The resulting answers will be approximate values since various default assumptions were used to develop natural gas default factors. The efficiency of the engine has the greatest effect on the concentration (ppmvd) to mass emission rate conversion (g/bhp-hr), which can vary from 20 to 40 percent. IN the calculations below, the efficiency is proportional to the engine brake specific fuel consumption.

PPM TO GM/Bhp-hr

Concentration in exhaust by volume (dry)(ppmvd) = volume of pollutant (Vp) x 10 volume of exhaust (Ve)

Vp = emission factor (g/bhp-hr) x horsepower x (1/molecular weight) x molar volume x conversion factors

Ve = F-factor for exhaust volume x excess air correction x engine brake specific fuel consumption x horsepower x conversion factors

These factors can be reduced to: ppmvd = (gm/Bhp-hr) * factor

Reciprocating Engines, natural gas fueled98

Pollutant	Factor
NOx	57-59
VOC	163-170
СО	93-97

Lean-burn Engines, natural gas fueled99

Pollutant	Factor
NOx	80
VOC	212
СО	123

⁹⁸ Values taken from California Air Pollution Control Officers Association (CAPCOA) report: <u>Portable Equipment Rule Piston IC Engine</u> <u>Technical Reference Document, 1995.</u>

⁹⁹ Factors provided from Waukesha

GM/Bhp-hr to Lb/MW-hr

 $Gm/Bhp-hr \times 3.07 = Ib/MW-hr$

- Includes 95 % factor for generator efficiency
- Conversion factors for grams to pounds and brake horsepower to watts

Gas Turbines

lb/MW-hr = (emission rate [lb/MMBtu]) x (3.413 [MMBtu/MWh]) / (efficiency)

2.5 ppmvd = 0.0093 lb/MMBtu for NOx

2 ppmvd = 0.0027 lb/MMBtu for VOC

5 ppmvd = 0.013 lb/MMBtu for CO

Efficiency for central station power plant is 50%

Source: California Air Resources Board, Guidance for the Permitting of Electric Generation Technologies, Appendix C: Procedure for Converting Emission Data to lb/MW-hr, July 2002.

Appendix D – Conversion Tables for HVAC-Integrated S-TES

Before 1984	kW Offset Table with Source Multiplier of 1.															
8 SEER	Equipment Nominal Tonnage (In Tons)															
Climate Zone	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
1	1.0	1.6	2.3	3.5	4.2	4.8	6.1	7.3	8.6	9.2	10.5	11.1	12.4	15.5	18.7	25.0
2	1.4	2.3	3.1	4.8	5.6	6.5	8.2	9.8	11.5	12.4	14.1	14.9	16.6	20.8	25.0	33.4
3	1.3	2.1	2.9	4.5	5.3	6.0	7.6	9.2	10.8	11.5	13.1	13.9	15.5	19.4	23.3	31.2
4	1.4	2.2	3.0	4.6	5.4	6.2	7.8	9.5	11.1	11.9	13.5	14.3	15.9	20.0	24.0	32.1
5	1.4	2.2	3.0	4.6	5.4	6.2	7.8	9.4	11.0	11.8	13.4	14.2	15.8	19.8	23.8	31.8
6	1.4	2.2	3.0	4.7	5.5	6.3	7.9	9.6	11.2	12.0	13.7	14.5	16.1	20.2	24.3	32.5
7	1.4	2.2	3.1	4.7	5.5	6.4	8.0	9.7	11.3	12.1	13.8	14.6	16.3	20.4	24.5	32.8
8	1.4	2.2	3.1	4.7	5.5	6.4	8.0	9.7	11.3	12.1	13.8	14.6	16.3	20.4	24.5	32.8
9	1.4	2.2	3.0	4.7	5.5	6.3	7.9	9.6	11.2	12.0	13.7	14.5	16.1	20.2	24.3	32.5
10	1.5	2.3	3.2	4.9	5.8	6.7	8.4	10.1	11.9	12.7	14.5	15.3	17.1	21.4	25.7	34.4
11	1.5	2.4	3.3	5.1	6.0	6.9	8.6	10.4	12.2	13.1	14.9	15.7	17.5	22.0	26.4	35.3
12	1.4	2.3	3.1	4.8	5.7	6.5	8.2	9.9	11.6	12.5	14.2	15.0	16.7	21.0	25.2	33.7
13	1.4	2.2	3.1	4.7	5.5	6.4	8.0	9.7	11.3	12.1	13.8	14.6	16.3	20.4	24.5	32.8
14	1.5	2.3	3.2	4.9	5.8	6.7	8.4	10.1	11.9	12.7	14.5	15.3	17.1	21.4	25.7	34.4
15	1.6	2.5	3.4	5.2	6.1	7.0	8.9	10.7	12.5	13.4	15.3	16.2	18.0	22.5	27.1	36.2
16	1.4	2.2	3.0	4.6	5.4	6.2	7.8	9.5	11.1	11.9	13.5	14.3	15.9	20.0	24.0	32.1
1984 - 1991	kW Offset Table with Source Multiplier of 1.															
8.9 SEER	Equipment Nominal Tonnage (In Tons)															
Climate	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
Zone 1	0.9	1.5	2.0	3.2	3.8	4.3	5.5	6.6	7.8	8.3	9.5	10.1	11.2	14.1	16.9	22.7
2	1.3	2.0	2.8	4.3	5.1	5.9	7.4	8.9	10.4	11.2	12.7	13.5	15.0	18.9	22.7	30.3
3	1.2	1.9	2.6	4.0	4.7	5.5	6.9	8.3	9.7	10.5	11.9	12.6	14.0	17.6	21.2	28.3
4	1.2	2.0	2.7	4.2	4.9	5.6	7.1	8.6	10.0	10.8	12.2	13.0	14.5	18.1	21.8	29.2
5	1.2	1.9	2.7	4.1	4.8	5.6	7.0	8.5	9.9	10.7	12.1	12.9	14.3	18.0	21.6	28.9
6	1.2	2.0	2.7	4.2	4.9	5.7	7.2	8.7	10.1	10.9	12.4	13.1	14.6	18.3	22.0	29.4
7	1.2	2.0	2.7	4.2	5.0	5.7	7.2	8.7	10.2	11.0	12.5	13.2	14.7	18.5	22.2	29.7
8	1.2	2.0	2.7	4.2	5.0	5.7	7.2	8.7	10.2	11.0	12.5	13.2	14.7	18.5	22.2	29.7
9	1.2	2.0	2.7	4.2	4.9	5.7	7.2	8.7	10.1	10.9	12.4	13.1	14.6	18.3	22.0	29.4
10	1.3	2.1	2.9	4.5	5.2	6.0	7.6	9.2	10.7	11.5	13.1	13.9	15.5	19.4	23.3	31.2
11	1.4	2.2	3.0	4.6	5.4	6.2	7.8	9.4	11.0	11.9	13.5	14.3	15.9	19.9	24.0	32.0
12	1.3	2.1	2.8	4.4	5.1	5.9	7.5	9.0	10.5	11.3	12.9	13.6	15.2	19.0	22.9	30.6
13	1.2	2.0	2.7	4.2	5.0	5.7	7.2	8.7	10.2	11.0	12.5	13.2	14.7	18.5	22.2	29.7
14	1.3	2.1	2.9	4.5	5.2	6.0	7.6	9.2	10.7	11.5	13.1	13.9	15.5	19.4	23.3	31.2
15	1.4	2.2	3.1	4.7	5.5	6.4	8.0	9.7	11.3	12.2	13.8	14.7	16.3	20.4	24.6	32.9
16	1.2	2.0	2.7	4.2	4.9	5.6	7.1	8.6	10.0	10.8	12.2	13.0	14.5	18.1	21.8	29.2
4000						1.6"	4 7	145	0-	- 80 21	!!					
1992 - 2005					ΚV		et Tabl				<u> </u>					
9.7 SEER Climate							ment									
Zone	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
1	0.8	1.3	1.9	2.9	3.5	4.0	5.1	6.1	7.2	7.7	8.8	9.3	10.4	13.0	15.7	21.0
2	1.2	1.9	2.6	4.0	4.7	5.4	6.8	8.2	9.7	10.4	11.8	12.5	13.9	17.4	21.0	28.1
3	1.1	1.7	2.4	3.7	4.4	5.0	6.4	7.7	9.0	9.7	11.0	11.7	13.0	16.3	19.6	26.2
4	1.1	1.8	2.5	3.8	4.5	5.2	6.6	7.9	9.3	10.0	11.3	12.0	13.4	16.8	20.2	27.0
5	1.1	1.8	2.4	3.8	4.5	5.1	6.5	7.8	9.2	9.9	11.2	11.9	13.2	16.6	20.0	26.7
6	1.1	1.8	2.5	3.9	4.6	5.3	6.6	8.0	9.4	10.1	11.4	12.1	13.5	16.9	20.4	27.3
7	1.1	1.8	2.5	3.9	4.6	5.3	6.7	8.1	9.5	10.2	11.6	12.2	13.6	17.1	20.6	27.5
8	1.1	1.8	2.5	3.9	4.6	5.3	6.7	8.1	9.5	10.2	11.6	12.2	13.6	17.1	20.6	27.5
9	1.1	1.8	2.5	3.9	4.6	5.3	6.6	8.0	9.4	10.1	11.4	12.1	13.5	16.9	20.4	27.3
10	1.2	1.9	2.7	4.1	4.8	5.6	7.0	8.5	9.9	10.7	12.1	12.8	14.3	17.9	21.6	28.9
11	1.2	2.0	2.7	4.2	5.0	5.7	7.2	8.7	10.2	11.0	12.5	13.2	14.7	18.4	22.2	29.6
12	1.2	1.9	2.6	4.0	4.8	5.5	6.9 38 7	8.3	9.8	10.5	11.9	12.6	14.0	17.6	21.2	28.3
13	1.1	1.8	2.5	3.9	4.6	5.3		8.1	9.5	10.2	11.6	12.2	13.6	17.1	20.6	27.5
14	1.2	1.9	2.7	4.1	4.8	5.6 5.9	7.0 7.4	8.5	9.9	10.7	12.1	12.8	14.3	17.9	21.6	28.9
15	1.3	2.1 1.8	2.8	3.8	5.1 4.5	5.9	6.6	9.0 7.9	10.5 9.3	11.3	12.8	13.6	15.1	18.9	22.8	30.4
16	1.1	1.0	2.5	ა.გ	4.5	3.2	0.0	7.9	ყ.პ	10.0	11.3	12.0	13.4	16.8	20.2	27.0

2006-2009 12 SEER			kW Offset Table with Source Multiplier of 1. Equipment Nominal Tonnage (In Tons)													
Climate Zone	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
1	0.6	1.1	1.5	2.4	2.8	3.3	4.2	5.1	5.9	6.4	7.3	7.7	8.6	10.8	13.0	17.4
2	0.9	1.5	2.1	3.3	3.9	4.5	5.6	6.8	8.0	8.6	9.8	10.4	11.5	14.5	17.4	23.3
3	0.9	1.4	2.0	3.1	3.6	4.2	5.3	6.4	7.5	8.0	9.1	9.7	10.8	13.5	16.3	21.8
4	0.9	1.5	2.0	3.2	3.7	4.3	5.4	6.6	7.7	8.3	9.4	10.0	11.1	13.9	16.8	22.5
5	0.9	1.4	2.0	3.1	3.7	4.2	5.4	6.5	7.6	8.2	9.3	9.9	11.0	13.8	16.6	22.2
6	0.9	1.5	2.0	3.2	3.8	4.3	5.5	6.6	7.8	8.3	9.5	10.1	11.2	14.1	16.9	22.7
7	0.9	1.5	2.1	3.2	3.8	4.4	5.5	6.7	7.9	8.4	9.6	10.2	11.3	14.2	17.1	22.9
8	0.9	1.5	2.1	3.2	3.8	4.4	5.5	6.7	7.9	8.4	9.6	10.2	11.3	14.2	17.1	22.9
9	0.9	1.5	2.0	3.2	3.8 4.0	4.3	5.5	6.6	7.8 8.2	8.3	9.5	10.1	11.2	14.1	16.9	22.7
11	1.0	1.6	2.2	3.5	4.0	4.0	5.8 6.0	7.0	8.5	8.8 9.1	10.1	10.7	11.9 12.2	14.9 15.3	17.9 18.4	24.0
12	0.9	1.5	2.1	3.3	3.9	4.7	5.7	6.9	8.1	8.7	9.9	10.5	11.7	14.6	17.6	23.6
13	0.9	1.5	2.1	3.2	3.8	4.4	5.5	6.7	7.9	8.4	9.6	10.2	11.3	14.2	17.1	22.9
14	1.0	1.6	2.2	3.4	4.0	4.6	5.8	7.0	8.2	8.8	10.1	10.7	11.9	14.9	17.9	24.0
15	1.0	1.7	2.3	3.6	4.2	4.9	6.1	7.4	8.7	9.3	10.6	11.3	12.5	15.7	18.9	25.3
16	0.9	1.5	2.0	3.2	3.7	4.3	5.4	6.6	7.7	8.3	9.4	10.0	11.1	13.9	16.8	22.5
2010-2014		kW Offset Table with Source Multiplier of 1.														
13 SEER						Equip	ment	Nomin	al Ton	nage (In Ton	s)				
Climate Zone	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
1	0.4	0.8	1.1	1.8	2.1	2.4	3.1	3.8	4.4	4.8	5.5	5.8	6.5	8.1	9.8	13.2
2	0.6	1.1	1.5	2.4	2.9	3.3	4.2	5.1	6.0	6.5	7.4	7.8	8.7	10.9	13.2	17.6
3	0.6	1.0	1.4	2.3	2.7	3.1	3.9	4.8	5.6	6.0	6.9	7.3	8.1	10.2	12.3	16.5
4	0.6	1.0	1.5	2.3	2.8	3.2	4.1	4.9	5.8	6.2	7.1	7.5	8.4	10.5	12.7	17.0
5 6	0.6	1.0	1.5	2.3	2.7	3.2	4.0	4.9 5.0	5.7 5.8	6.1	7.0 7.1	7.4 7.6	8.3	10.4 10.6	12.5 12.8	16.8 17.1
7	0.6	1.1	1.5	2.4	2.8	3.3	4.1	5.0	5.9	6.3	7.1	7.7	8.5	10.7	12.9	17.3
8	0.6	1.1	1.5	2.4	2.8	3.3	4.1	5.0	5.9	6.3	7.2	7.7	8.5	10.7	12.9	17.3
9	0.6	1.1	1.5	2.4	2.8	3.2	4.1	5.0	5.8	6.3	7.1	7.6	8.4	10.6	12.8	17.1
10	0.7	1.1	1.6	2.5	3.0	3.4	4.3	5.3	6.2	6.6	7.6	8.0	8.9	11.2	13.5	18.1
11	0.7	1.2	1.6	2.6	3.1	3.5	4.5	5.4	6.4	6.8	7.8	8.3	9.2	11.6	13.9	18.6
12	0.7	1.1	1.6	2.5	2.9	3.4	4.3	5.2	6.1	6.5	7.4	7.9	8.8	11.0	13.3	17.8
13	0.6	1.1	1.5	2.4	2.8	3.3	4.1	5.0	5.9	6.3	7.2	7.7	8.5	10.7	12.9	17.3
14	0.7	1.1	1.6	2.5	3.0	3.4	4.3	5.3	6.2	6.6	7.6	8.0	8.9	11.2	13.5	18.1
15	0.7	1.2	1.7	2.7	3.1	3.6	4.6	5.6	6.5	7.0	8.0	8.5	9.4	11.9	14.3	19.1
16	0.6	1.0	1.5	2.3	2.8	3.2	4.1	4.9	5.8	6.2	7.1	7.5	8.4	10.5	12.7	17.0
6/2014 After					k۷	N Offse	et Tabl	e with	Source	e Mult	iplier o	of 1.				
14 SEER						Equip	ment	Nomin	al Ton	nage (In Ton	s)				
Climate Zone	1	1.5	2	3	3.5	4	5	6	7	7.5	8.5	9	10	12.5	15	20
1	0.4	0.8	1.1	1.8	2.1	2.4	3.1	3.8	4.4	4.8	5.5	5.8	6.5	8.1	9.8	13.2
2	0.6	1.1	1.5	2.4	2.9	3.3	4.2	5.1	6.0	6.5	7.4	7.8	8.7	10.9	13.2	17.6
2	0.6	1.0	1.4	2.3	2.7	3.1	3.9	4.8	5.6	6.0	6.9	7.3	8.1	10.2	12.3	16.5
3					2.8	3.2	4.1	4.9	5.8	6.2	7.1	7.5	8.4	10.5	12.7	17.0
4	0.6	1.0	1.5	2.3	2.0	_										
4 5	0.6	1.0	1.5	2.3	2.7	3.2	4.0	4.9	5.7	6.1	7.0	7.4	8.3	10.4	12.5	16.8
4 5 6	0.6	1.0	1.5 1.5	2.3	2.7	3.2	4.1	5.0	5.8	6.3	7.1	7.6	8.4	10.6	12.8	17.1
4 5 6 7	0.6 0.6 0.6	1.0 1.1 1.1	1.5 1.5 1.5	2.3 2.4 2.4	2.7 2.8 2.8	3.2	4.1 4.1	5.0 5.0	5.8 5.9	6.3 6.3	7.1 7.2	7.6 7.7	8.4 8.5	10.6 10.7	12.8 12.9	17.1 17.3
4 5 6 7 8	0.6 0.6 0.6	1.0 1.1 1.1 1.1	1.5 1.5 1.5	2.3 2.4 2.4 2.4	2.7 2.8 2.8 2.8	3.2 3.3 3.3	4.1 4.1 4.1	5.0 5.0 5.0	5.8 5.9 5.9	6.3 6.3	7.1 7.2 7.2	7.6 7.7 7.7	8.4 8.5 8.5	10.6 10.7 10.7	12.8 12.9 12.9	17.1 17.3 17.3
4 5 6 7 8 9	0.6 0.6 0.6 0.6	1.0 1.1 1.1 1.1 1.1	1.5 1.5 1.5 1.5	2.3 2.4 2.4 2.4 2.4	2.7 2.8 2.8 2.8 2.8	3.2 3.3 3.3 3.2	4.1 4.1 4.1	5.0 5.0 5.0	5.8 5.9 5.9 5.8	6.3 6.3 6.3	7.1 7.2 7.2 7.1	7.6 7.7 7.7 7.6	8.4 8.5 8.5 8.4	10.6 10.7 10.7 10.6	12.8 12.9 12.9 12.8	17.1 17.3 17.3 17.1
4 5 6 7 8 9	0.6 0.6 0.6 0.6 0.6 0.7	1.0 1.1 1.1 1.1 1.1	1.5 1.5 1.5 1.5 1.5	2.3 2.4 2.4 2.4 2.4 2.5	2.7 2.8 2.8 2.8 2.8 3.0	3.2 3.3 3.3 3.2 3.4	4.1 4.1 4.1 4.1 4.3	5.0 5.0 5.0 5.0 5.3	5.8 5.9 5.9 5.8 6.2	6.3 6.3 6.3 6.6	7.1 7.2 7.2 7.1 7.6	7.6 7.7 7.7 7.6 8.0	8.4 8.5 8.5 8.4 8.9	10.6 10.7 10.7 10.6 11.2	12.8 12.9 12.9 12.8 13.5	17.1 17.3 17.3 17.1 18.1
4 5 6 7 8 9 10	0.6 0.6 0.6 0.6 0.7 0.7	1.0 1.1 1.1 1.1 1.1 1.1	1.5 1.5 1.5 1.5 1.5 1.6 1.6	2.3 2.4 2.4 2.4 2.4 2.5 2.6	2.7 2.8 2.8 2.8 2.8 3.0 3.1	3.2 3.3 3.3 3.2 3.4 3.5	4.1 4.1 4.1 4.1 4.3 4.5	5.0 5.0 5.0 5.0 5.3 5.4	5.8 5.9 5.9 5.8 6.2 6.4	6.3 6.3 6.3 6.6 6.8	7.1 7.2 7.2 7.1 7.6 7.8	7.6 7.7 7.7 7.6 8.0 8.3	8.4 8.5 8.5 8.4 8.9 9.2	10.6 10.7 10.7 10.6 11.2 11.6	12.8 12.9 12.9 12.8 13.5 13.9	17.1 17.3 17.3 17.1 18.1 18.6
4 5 6 7 8 9	0.6 0.6 0.6 0.6 0.6 0.7	1.0 1.1 1.1 1.1 1.1	1.5 1.5 1.5 1.5 1.5	2.3 2.4 2.4 2.4 2.4 2.5	2.7 2.8 2.8 2.8 2.8 3.0	3.2 3.3 3.3 3.2 3.4	4.1 4.1 4.1 4.3 4.5 4.3	5.0 5.0 5.0 5.0 5.3	5.8 5.9 5.9 5.8 6.2	6.3 6.3 6.3 6.6	7.1 7.2 7.2 7.1 7.6	7.6 7.7 7.7 7.6 8.0	8.4 8.5 8.5 8.4 8.9	10.6 10.7 10.7 10.6 11.2	12.8 12.9 12.9 12.8 13.5	17.1 17.3 17.3 17.1 18.1
4 5 6 7 8 9 10 11	0.6 0.6 0.6 0.6 0.7 0.7	1.0 1.1 1.1 1.1 1.1 1.1 1.2	1.5 1.5 1.5 1.5 1.5 1.6 1.6	2.3 2.4 2.4 2.4 2.5 2.6 2.5	2.7 2.8 2.8 2.8 2.8 3.0 3.1 2.9	3.2 3.3 3.3 3.2 3.4 3.5 3.4	4.1 4.1 4.1 4.1 4.3 4.5	5.0 5.0 5.0 5.0 5.3 5.4 5.2	5.8 5.9 5.9 5.8 6.2 6.4 6.1	6.3 6.3 6.3 6.3 6.6 6.8 6.5	7.1 7.2 7.2 7.1 7.6 7.8 7.4	7.6 7.7 7.7 7.6 8.0 8.3 7.9	8.4 8.5 8.5 8.4 8.9 9.2 8.8	10.6 10.7 10.7 10.6 11.2 11.6	12.8 12.9 12.9 12.8 13.5 13.9	17.1 17.3 17.3 17.1 18.1 18.6 17.8
4 5 6 7 8 9 10 11 12	0.6 0.6 0.6 0.6 0.7 0.7 0.7	1.0 1.1 1.1 1.1 1.1 1.2 1.1	1.5 1.5 1.5 1.5 1.5 1.6 1.6 1.6	2.3 2.4 2.4 2.4 2.5 2.6 2.5 2.4	2.7 2.8 2.8 2.8 2.8 3.0 3.1 2.9 2.8	3.2 3.3 3.2 3.4 3.5 3.4 3.3	4.1 4.1 4.1 4.3 4.5 4.3	5.0 5.0 5.0 5.0 5.3 5.4 5.2	5.8 5.9 5.9 5.8 6.2 6.4 6.1 5.9	6.3 6.3 6.3 6.6 6.8 6.5 6.3	7.1 7.2 7.2 7.1 7.6 7.8 7.4 7.2	7.6 7.7 7.7 7.6 8.0 8.3 7.9	8.4 8.5 8.5 8.4 8.9 9.2 8.8 8.5	10.6 10.7 10.7 10.6 11.2 11.6 11.0	12.8 12.9 12.9 12.8 13.5 13.9 13.3 12.9	17.1 17.3 17.3 17.1 18.1 18.6 17.8

Appendix E – Updates to the GHG Emissions Factor Section 379.6(b)(2) as Amended by Senate Bill 861 (Legacy Projects Only)

SGIP GHG Emissions Eligibility Factor – The Equation

We find that to calculate the GHG emissions eligibility factor, it is reasonable to use the following equation:

```
GHG EF = (0.5(EROLF * (1 – WFP) + EROP* WFP) + 0.5 * (1-RPS% * (1 – LLF)) * (ERBLF* (1 – WFP) + ERBP * WFP))/(1 – LLF)
```

Where:

GHG EF = greenhouse gas emission factor

EROLF = operating margin emission rate of load-following plants = 382 kgCO2/MWh

WFP = weighting factor for peaker plants = 10%

EROP = operating margin emission rate of peaking

plants = 544 kgCO2/MWh

RPS% = average RPS portfolio requirement for the program year (i.e., project

years 6 - 10)

ERBLF = build margin emission rate of load-following plants = 368 kgCO2/MWh

ERBP = build margin emission rate of peaking

plants = 524 kgCO2/MWh

LLF = line loss factor = 8.4%

Substituting the adopted values for Program Year 2016 into this equation yields:

GHG EF = (0.5 (382 kgCO2/MWh * (1 - 0.10) + 544 kgCO2/MWh * 0.10) + 0.5 (1-0.40 * (1 - 0.084)) * (368 kgCO2/MWh * (1 - 0.10) + 524 kgCO2/MWh * 0.10))/(1 - 0.084)

GHG EF = 350 kgCO2/MWh

Share of Avoided Renewables in Calculating SGIP GHG Emissions Eligibility Threshold

Assumed RPS Targets 2020 – 2030, with and without Line Loss Adjustments

	2020	2021	2022	2023	2024	2025	2026	2027	2028	2029	2030
Nominal	33.0	34.8	36.5	38.3	40.0	41.7	43.3	45.0	46.7	48.3	50.0
RPS	%	%	%	%	%	%	%	%	%	%	%
Adjuste	30.2	31.8	33.4	35.0	36.6	38.2	39.7	41.2	42.8	44.2	45.8
d RPS	%	%	%	%	%	%	%	%	%	%	%

Note: The adjusted RPS is calculated as the product of the nominal percentage and (1 – the line loss factor)

Average Share of Avoided Renewable Energy in Build Margin by Program Year

Program Year	2016	2017	2018	2019	2020
Build Margin RPS,	40.0%	41.7%	43.3%	45.0%	46.7%
Nominal					
Build Margin RPS,	36.6%	38.2%	39.7%	41.2%	42.7%
Adjusted for line					
losses					

Note: The build margin for each program year is the simple average of the RPS percentages for years 6 – 10 after the program year. For example, the program year 2016 average share of renewable energy avoided equals the average of the RPS targets for 2022 through 2026.

SGIP GHG Eligibility Emissions Factors, kgCO2/MWh

Program Year	2016	2017	2018	2019	2020
10-Year Average	350	347	344	340	337
First-Year Average	334	332	329	325	321

Calculation of Minimum Roundtrip Efficiency

Line Loss On	10.3%		
Peak			
Line Loss Off	5.3%		
Peak			
Degradation Rate	1.0%		
First Year RTE	69.6%		
Ten-Year Avg	66.5%		
RTE			
Sum of Ann'l	0		
GHGs			

Year	Off-peak ER	On-peak ER	GHG emitted	GHG avoided	Net GHG per MWh
1	382	544	580	606	-27
2	382	544	585	606	-21
3	382	544	591	606	-15
4	382	544	597	606	-9
5	382	544	603	606	-3
6	368	524	587	584	3
7	368	524	593	584	9
8	368	524	599	584	15
9	368	524	605	584	21
10	368	524	611	584	27

Appendix F – Incentive rate for large-scale storage projects claiming the ITC at all incentive levels

	\$/Wh								
Non- ITC	0.50	0.45	0.40	0.35	0.30	0.25	0.20	0.15	0.10
ITC	0.36	0.32	0.29	0.25	0.22	0.18	0.14	0.11	0.07

SELF-GENERATION INCENTIVE PROGRAM

Energy Storage Proposed System Information Customer Attestation

The following information (Responses to Questions 1-5) is required for all Equity and Non-equity Resiliency projects and all other projects with a discharge duration greater than two hours.

1)	Provide an estimate of how long the project's fully charged battery will provide electricity for the relevant facility average load during an outage.						
2)	Indicate whether the project's critical loads can and will be isolated.						
3)	Provide an estimate of how long the project's fully charged battery will provide electricity to critical uses during an outage.						
4)	Provide an estimate of how long the project can operate in less-than favorable circumstances, such as if an outage occurs when the battery has been discharged or during the winter (if paired with solar).						
5)	Summarize information given to the customer about how the customer may best prepare the storage system to provide backup power, in the case of a Public Safety Power Shutoff (PSPS) event announced in advance (provide an attachment with more information if necessary).						

ATTESTATION

DEVELOPER SECTION]	
(print name of Developer), hereby attest that each of the statemen rovided in this document are true and correct.	ts
gnature:	
ame Printed:	
tle:	
ate:	
CUSTOMER SECTION]	
(print name of Host Customer), hereby attest I have received the formation provided in this document prior to signing a contract with the developer.	
gnature:	
ame Printed:	
ate:	

BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

Order Instituting Rulemaking Regarding Policies, Procedures and Rules for the California Solar Initiative, the Self-Generation Incentive Program and Other Distributed Generation Issues. RULEMAKING 12-11-005 (Filed November 8, 2012)

CERTIFICATE OF SERVICE

I hereby certify that I have this day served a copy of this *CSE Advice Letter 110-E-A / SCE Advice Letter 4192-E-A / SoCalGas Advice Letter 5619-A / PG&E Advice Letter 4237-G-A/5808-E-A* on all known parties of record in this proceeding by delivering a copy via email to the current service list for R.12-11-005. Pursuant to the temporary changes to the California Public Utilities Commission's Rules of Practice and Procedure, service is limited to electronic service until further notice. Accordingly, all requirements for delivery or service personally or by mail for filings are suspended until further notice.

Executed on June 5, 2020, in San Diego, CA.

Ben Airth

Senior Distributed Generation Policy Manager

Center for Sustainable Energy®

3980 Sherman St., Suite 170

San Diego, CA 92110

Dayounier Chith

Tel: (858) 244-1194

Benjamin.airth@energycenter.org



CALIFORNIA PUBLIC UTILITIES COMMISSION Service Lists

PROCEEDING: R1211005 - CPUC - OIR REGARDING

FILER: CPUC LIST NAME: LIST

LAST CHANGED: MAY 28, 2020

Download the Comma-delimited File

About Comma-delimited Files

Back to Service Lists Index

Parties

CARMEN L. GIBBS STAFF ATTORNEY

CA. DEPT. OF COMMUNITY SVCS. & DEV'T.

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: DEPARTMENT OF COMMUNITY SERVICES

AND DEVELOPMENT (DCSD)

CURTIS SEYMOUR

 ${\sf SR.\ MANAGER\ -\ GOVERNMENT\ AFFAIRS}$

.....

SUNEDISON LLC EMAIL ONLY

EMAIL ON LY, CA 00000 FOR: SUNEDISON LLC

DANIEL W. DOUGLASS DOUGLASS & LIDDELL

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: DIRECT ACCESS CUSTOMER COALITION

(DACC)

DARYL R. MAAS

CE0

MAAS ENERGY WORKS INC

EMAIL ONLY

EMAIL ONLY, CA 00000 FOR: MAAS ENERGY WORKS INC.

DAVID RUNSTEN

POLICY DIRECTOR

CALIF. CLIMATE AND AGRICULTURE NETWORK

EMAIL ONLY

EMAIL ONLY, CA 00000 FOR: CALIFORNIA CLLIMATE AND

AGRICULTURE NETWORK

EDRIC F. GUISE

NATIONAL ENERGY SOLUTIOINS, LLC

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: NATIONAL ENERGY SOLUTIONS, LLC

FRANCO CASTALDINI PRESIDENT/CO-FOUNDER LEVA ENERGY, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000 FOR: LEVA ENERGY, INC.

HOWARD CHOY GENERAL MGR.

COUNTY OF LOS ANGELES

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: SOUTHERN CALIFORNIA REGIONAL

ENERGY NETWORK (SCRENØ

JAMES M. QUAN

ELLI CIRCLE RENELIARLES ILC

FULL CIRCLE RENEWABLES, LLC

EMAIL ONLY

EMAIL ONLY, CA 00000

JOSEPH WEIDMAN

DIR - REGULATORY / LEGISLATIVE AFFAIRS

PENINSULA CLEAN ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: FULL CIRCLE RENEWABLES, LLC FOR: PENINSULA CLEAN ENERGY

JOSHUA NELSON
ASSOCIATE
BEST BEST & KRIEGER LLP
EMAIL ONLY
EMAIL ONLY, CA 00000

FOR: WESTERN MUNICIPAL WATER DISTRICT;
TERRA VERDE RENEWABLES PARTNERS, LLC;
LEMON GROVE SCHOOL DISTRICT; CITY OF
BENICIA; VALLEY CENTER MUNICIPAL WATER
DISTRICT; SAN DIEGO UNIFIED SCHOLL
DISTRICT; RANCHO CALIFORNIA WATER
DISTRICT; PADRE DAM MUNICIPAL WATER
DISTRICT;

KENNETH SAHM WHITE CLEAN COALITION EMAIL ONLY EMAIL ONLY, CA 00000

FOR: CLEAN COALITION

LISA ROSEN SECRETARY

ENERGY EFFICIENCY SOLAR
EMAIL ONLY
EMAIL ONLY, CA 00000
FOR: ENERGY EFFICIENCY SOLAR

MATTHEW KOBER
PYRAMID SOLAR INC.
EMAILL ONLY
EMAIL ONLY, CA 00000

EMAIL ONLY, CA 00000 FOR: PYRAMID SOLAR

NICK GOODWIN SELF GOODWIN SELF ECO CONSULTING, LLC EMAIL ONLY

EMAIL ONLY, CA 00000 FOR: GOODWIN SELF ECO CONSULTING, LLC

S. GARY VARGA
ATTORNEY AT LAW
LAW OFFICES OF S. GARY VARGA
EMAIL ONLY
EMAIL ONLY, CA 00000
FOR: ALISAL WATER CORPORATION D/B/A
ALCO WATER SERVICE / CALIFORNIA

TAM HUNT
HUNT CONSULTING
EMAIL ONLY
EMAIL ONLY, CA 00000
FOR: COMMUNITY ENVIRONMENTAL
COUNCIL/COMMUNITY RENEWABLE SOLUTIONS /
RECYCLE

DENNIS HAINES ENVIRONMENTAL POWER CORP. EMAIL ONLY

UTILITIES SERVICE, INC.

EMAIL ONLY, NY 00000-0000

FOR: MICROGY, INC.

MARTIN HOMEC REDWOOD RENEWABLES/CARE EMAIL ONLY EMAIL ONLY, CA 00000-0000 FOR: WOMEN'S ENERGY MATTERS LYNN M. HAUG ELLISON, SCHNEIDER & HARRIS, L.L.P. EMAIL ONLY EMAIL ONLY, CA 00000

MATTHEW SWINDLE CEO & FOUNDER NLINE ENERGY, INC. EMAIL ONLY

FOR: FUELCELL ENERGY

EMAIL ONLY, CA 00000 FOR: NLINE ENERGY, INC.

PETER MILLER
NATURAL RESOURCES DEFENSE COUNCIL

EMAIL ONLY EMAIL ONLY, CA 00000

FOR: NATURAL RESOURCES DEFENSE COUNCIL

(NRDC)

SHIN TANIMOTO
SANYO NORTH AMERICA CORPORATION

SANTO NORTH AMERICA CORPORATION

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: SANYO NORTH AMERICA CORPORATION

TIMOTHY LIPMAN CO-DIRECTOR

US DOE PACIFIC REGION CLEAN ENERGY CTR.

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: U.S. DOE PACIFIC REGION CLEAN

ENERGY APPLICATION CENTER

JOSEPH OLDHAM PROGRAM DIRECTOR

LOCAL GOVERNMENT COMMISSION

EMAIL ONLY

EMAIL ONLY, CA 00000-0000 FOR: LOCAL GOVERNMENT COMMISSION

MARK F. BUCKLEY

VP - ENVIRONMENTAL AFFAIRS

STAPLES, INC.
500 STAPLES DR.
FRAMINGTON, MA 01702
FOR: STAPLES, INC.

KENT MCCORD

DIR - MARKETING STRATEGY

DOOSAN FUEL CELL AMERICA, INC.

195 GOVERNORS HIGHWAY

SOUTH WINDSOR, CT 06074

FOR: DOOSAN FUEL CELL AMERICA, INC.

SHARALYN S. CONNER
ASSOCIATE GENERAL COUNSEL
PLUG POWER INC.
968 ALBANY SHAKER ROAD
LATHAM, NY 12110
FOR: PLUG POWER INC.

ANCA FAUR
DEVELOPMENT PROGRAM MGR.
JOHNSON MATTHEY FUEL CELLS INC.
1397 KING ROAD
WEST CHESTER, PA 19380
FOR: JOHNSON MATTHEY FUEL CELLS INC.

ROSS ALLEN BUNTROCK
ARNALL GOLDEN GREGORY, LLP
1775 PENNSYLVANIA AVE., NW, SUITE 1000
WASHINGTON, DC 20006
FOR: MULTIFAMILY AFFORDABLE SOLAR HOMES
COALITION (MASH)

JENNIFER A. WARREN
VP - TECH. POLICIES & REGULATION
LOCKHEED MARTIN CORPORATION
2121 CRYSTAL DR., STE. 100
ARLINGTON, VA 22202
FOR: LOCKHEED MARTIN CORPORATION

TRACY WOODARD
NISSAN NORTH AMERICA, INC.
ONE NISSAN WAY
FRANKLIN, TN 37067
FOR: NISSAN NORTH AMERICA, INC.

RAY SIADA
CORPORATE ENERGY MANAGER
GUARDIAN INDUSTRIES CORP
2300 HARMON ROAD
AUBURN HILLS, MI 48326
FOR: GUARDIAN INDUSTRIES CORP.

N. ROSS BUCKENHAM CEO CALIFORNIA BIOENERGY LLC 2828 ROUTH STREET, STE. 500 DALLAS, TX 75201 FOR: CALIFORNIA BIOENERGY LLC (CALBIO)

KELSEY M. WALKER
TAS ENERGY
6110 CULLEN BLVD.
HOUSTON, TX 77021
FOR: HEAT IS POWER

DENISE GRAB
WESTERN REGIONAL DIRECTOR
INSTITUTE FOR POLICY INTEGRITY
139 MACDOUGAL STREET, 3RD FLOOR
NEW YORK, NY 10012
FOR: INSTITUTE FOR POLICY INTEGRITY

ETHAN BROWN
BALLARD POWER SYSTEMS
2 INDUSTRIAL AVE.
LOWELL, MA 18510
FOR: BALLARD POWER SYSTEMS

JOSHUA C. GREENE
VP - GOV'T & INDUSTRY AFFAIRS
A.O. SMITH CORPORATION
400 NORTH CAPITOL ST., NW STE. 585
WASHINGTON, DC 20001
FOR: A. O. SMITH CORPORATION

BUD DEFLAVIIS
DIRECTOR OF GOVERNMENT AFAIRS
FUEL CELL AND HYDROGEN ENERGY ASSN.
1211 CONNECTICUT AVE., NW, SUITE 650
WASHINGTON, DC 20036
FOR: FUEL CELL AND HYDROGEN ENERGY
ASSOCIATION

KAREN B. MEYERS
VP - GOVN'T AFFAIRS
RHEEM MANUFACTURING COMPANY
1100 ABERNATHY ROAD, STE. 1700
ATLANTA, GA 30328
FOR: RHEEM MANUFACTURING COMPANY

ARTHUR H. SIEGAL ATTORNEY AT LAW JAFFE, RAITT, HEUER & WEISS 27777 FRANKLIN ROAD, SUITE 2500 SOUTHFIELD, MI 48034 FOR: GREEN CHARGE NETWORKS

ROBERT WOLFER
MGR - GOVERNMENT RELATIONS
BRADFORD WHITE CORPORATION
200 LAFAYETTE STREET
MIDDLEVILLE, MI 49333
FOR: BRADFORD WHITE CORPORATION

GARY HILBERG
EXECUTIVE VICE PRESIDENT
TAS
6110 CULLEN BLVD.
HOUSTON, TX 77021
FOR: TAS / POWER GENERATION /
AUGMENTATION. COOLING & HEATING SYSTEMS
/ INDUSTRIAL PROCESS SYSTEMS

KELLY MURPHY
BUSINESS DEVELOPMENT
STEFFES, LLC
BOULDER COLORADO OFFICE
5160 ILLINI WAY
BOULDER, CO 80303

FOR: KELLY MURPHY (MOVANT)

LORI A. GLOVER **PRESIDENT** S.O.L.I.D. USA, INC.

10645 N. TATUM BLVD., SUITE 200-306

PHOENIX, AZ 85028 FOR: S.O.L.I.D. SOLAR

HOWARD CHOY GENERAL MGR. COUNTY OF LOS ANGELES OFFICE OF SUSTAINABILITY 1100 NORTH EASTERN AVENUE LOS ANGELES, CA 90063

FOR: LOCAL GOVERNMENT SUSTAINABLE

ENERGY COALITION (LGSEC)

ROBERT RASMUSSEN ASSOCIATE, REGULATORY AFFAIRS SWELL ENERGY, LLC 1515 7TH ST., NO. 049 SANTA MONICA, CA 90401 FOR: SWELL ENERGY, LLC

ENID JOFFE PRESIDENT CLEAN FUEL CONNECTION, INC. 11800 CLARK STREET ARCADIA, CA 91006 FOR: CLEAN FUEL CONNECTION, INC. (CLEAN

FUEL)

MILISSA MARONA DIR - UTILITY & GOV'T RELATIONS SONNEN, INC. 10800 BURBANK BLVD., STE. C NORTH HOLLYWOOD, CA 91601 FOR: SONNEN, INC.

KENNETH J. FRISBIE MANAGING DIRECTOR BIOFUELS ENERGY, LLC 2211 ENCINITAS BLVD ENCINITAS, CA 92024 FOR: BIOFUELS ENERGY, LLC

DONALD C. LIDDELL **ATTORNEY** DOUGLASS & LIDDELL 2928 2ND AVENUE SAN DIEGO, CA 92103 FOR: POWERTREE SERVICES, INC./ ICE ENERGY, INC.

SEPHRA A. NINOW, J.D. DIR - REGULATORY AFFAIRS CENTER FOR SUSTAINABLE ENERGY 3980 SHERMAN ST., STE. 170 SAN DIEGO, CA 92110 FOR: CENTER FOR SUSTAINABLE ENERGY (FORMERLY CALIFORNIA CENTER FOR SUSTAINABLE ENERGY)

JOHN W. LESLIE, ESQ.

ATTORNEY

EDWARD L. HSU SR COUNSEL SOUTHERN CALIFORNIA GAS COMPANY 555 WEST 5TH STREET, GT14E7 LOS ANGELES, CA 90013 FOR: SOUTHERN CALIFORNIA GAS COMPANY

KEVIN R. MCSPADDEN **ATTORNEY** SHOREBREAK ENERGY DEVELOPERS, LLC 2801 BELDEN DRIVE LOS ANGELES, CA 90068 FOR: SHOREBREAK ENERGY DEVELOPERS, LLC

JESSALYN ISHIGO ENVIRONMENTAL BUSINESS DEVELOPMENT AMERICAN HONDA MOTOR CO., INC. 1919 TORRANCE BLVD. TORRANCE, CA 90501 FOR: AMERICAN HONDA MOTOR CO., INC.

JUSTIN RATHKE CAPSTONE TURBINE CORPORATION 21211 NORDHOFF STREET CHATSWORTH, CA 91311 FOR: CAPSTONE TURBINE CORPORATION

MARIO E. DOMINGUEZ SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE / PO BOX 800 ROSEMEAD, CA 91770 FOR: SOUTHERN CALIFORNIA EDISON COMPANY

SCOTT DEBENHAM DEBENHAM ENERGY LLC 11317 VALLE VISTA ROAD LAKESIDE, CA 92040 FOR: DEBENHAM ENERGY, LLC

DONALD C. LIDDELL ATTORNEY AT LAW DOUGLASS & LIDDELL 2928 2ND AVENUE SAN DIEGO, CA 92103 FOR: TRANE US, INC.

MICHAEL SHAMES, ESQ. SAN DIEGO CONSUMERS' ACTION NETWORK 6975 CAMINO AMERO SAN DIEGO, CA 92111 FOR: SAN DIEGO CONSUMERS' ACTION NETWORK

JONATHAN NEWLANDER SR. COUNSEL

DENTONS US LLP 4655 EXECUTIVE DRIVE, STE. 700 SAN DIEGO, CA 92121

FOR: SHELL ENERGY NORTH AMERICA

SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTURY PARK COURT, CP32D SAN DIEGO, CA 92123

FOR: SAN DIEGO GAS & ELECTRIC COMPANY

DAVID DAVIS 61736 ONAGA TRAIL JOSHUA TREE, CA 92252 FOR: DAVID DAVIS

JOSEPH PERRY **FLEXENERGY** 9400 TOLEDO WAY IRVINE, CA 92618 FOR: FLEXENERGY LLC

DR. JACK BROUWER

DIR

NATIONAL FUEL CELL RESEARCH CENTER UNIVERSITYOF CALIFORNIA, IRVINE 221 ENGINEERING LABORATORY FACILITY B323 IRVINE, CA 92697-3550 FOR: ADVANCED POWER AND ENERGY PROGRAM

SCOTT SAMUELSEN DIRECTOR NATIONAL FUEL CELL RESEARCH CENTER UNIVERISTY OF CALIFORNIA IRVINE, CA 92697-3550 FOR: NATIONAL FUEL CELL RESEARCH CTR / CALIFORNIA STATIONARY FUEL CELL

JAMES K. STEVENS MEMBERS

RENEWABLE ENERGY PARTNERS, LLC 13217 JAMBOREE ROAD, NO. 545

TUSTIN, CA 92782

FOR: RENEWABLE ENERGY PARTNERS, LLC

CATHERINE E. VON BURG PRESIDENT & CEO SIMPLIPHI POWER, INC / OES BATTERIES 420 BRYANT CIRCLE, SUITE B

COLLABORATIVE

OJAI, CA 93023 FOR: SIMPLIPHI POWER, INC.

LEIGH ESTUS CLEARPOWER SYSTEMS, INC. 948 ETTIN AVENUE SIMI VALLEY, CA 93065 FOR: CLEARPOWER SYSTEMS, INC. CHARLES HEWITT 770 LA BUENA TIERRA SANTA BARBARA, CA 93111 FOR: CHARLES HEWITT

STEVE EASTHAM, PE **ELECTRICAL ENGINEER** GUARDIAN INDUSTRIES CORP - KINGSBURG 11535 E. MOUNTAIN VIEW KINGSBURG, CA 93631 FOR: GUARDIAN INDUSTRIES CORP.

JAMES R. RILEY CO-CHAIR CCCCWE PO BOX 7223 SPRECKLES, CA 93962 FOR: CENTRAL COAST COALITION OF COMMUNITIES FOR WASTEWATER EQUITY

ADAM SIMPSON **FOUNDER** ETAGEN, INC. 186 CONSTITUTION DRIVE MENLO PARK, CA 94025 FOR: ETAGEN, INC.

MATTHEW B. WILSON FOUNDATION WINDPOWER 431 BURGESS DR., STE. 200 MENLO PARK, CA 94025-3478 FOR: FOUNDATION WINDPOWER, LLC

TED KO DIRECTOR OF POLICY STEM, INC. 100 ROLLINS ROAD MILLBRAE, CA 94030 FOR: STEM INC.

SUZANNE H. EMERSON EMERSON ENVIRONMENTAL, LLC 160 PLYMOUTH AVENUE SAN CARLOS, CA 94070-1621 FOR: EMERSON ENVIRONMENTAL, LLC

MARC D. JOSEPH ADAMS BROADWELL JOSEPH & CARDOZO 601 GATEWAY BLVD. STE 1000 SOUTH SAN FRANCISCO, CA 94080 FOR: COALITION OF CALIFORNIA UTILITY EMPLOYEES (CCUE)

SKY C. STANFIELD SHUTE, MIHALY AND WEINBERGER, LLP 396 HAYES STREET SAN FRANCISCO, CA 94102 FOR: INTERSTATE RENEWABLE ENERGY COUNCIL (IREC)

THERESA L. MUELLER CHIEF ENERGY & TELECOM DEPUTY CITY AND COUNTY OF SAN FRANCISCO CITY HALL, ROOM 234

ADAM LORIMER ALPHABET ENERGY INC. 239 8TH ST., S239 8TH ST., STE. 12 SAN FRANCISCO, CA 94103

1 DR. CARLTON B. GOODLETT PLACE SAN FRANCISCO, CA 94102-4682

FOR: CITY AND COUNTY OF SAN FRANCISCO

EDDIE AHN EXEC. DIRECTOR

BRIGHTLINE DEFENSE PROJECT 1028A HOWARD ST.

SAN FRANCISCO, CA 94103

FOR: BRIGHTLINE DEFENSE PROJECT

SERJ BERELSON

MGR - REGULATORY STRATEGY SUNVERGE ENERGY, INC. 950 MINNA STREET SAN FRANCISCO, CA 94103 FOR: SUNVERGE ENERGY, INC.

CHUCK RAMES

GROWING ENERGY LABS, INC. 657 MISSION STREET, STE. 500 SAN FRANCISCO, CA 94105 FOR: GROWING ENERGY LABS,INC.

MELICIA CHARLES SUNRUN INC.

595 MARKET STREET, 29TH FL. SAN FRANCISCO, CA 94105

FOR: SUNRUN INC.

NORA SHERIFF, ESQ. $\,$

ATTORNEY
BUCHALTER, A PROFESSIONAL CORPORATION
55 SECOND STREET, SUITE 1700
SAN FRANCISCO, CA 94105-3493

FOR: BP ENERGY COMPANY

JOE KARP ATTORNEY

WINSTON & STRAWN LLP 101 CALIFORNIA STREET SAN FRANCISCO, CA 94111 FOR: GRID ALTERNATIVES

VIDHYA PRABHAKARAN

ATTORNEY

DAVIS WRIGHT TREMAINE LLP 505 MONTGOMERY ST., STE. 800 SAN FRANCISCO, CA 94111

FOR: LIBERTY UTILITIES (FORMERLY

CALPECO) LLC

MICHAEL S. HINDUS PILLSBURY WINTHROP SHAW PITTMAN LLP 4 EMBARCADERO CENTER, FLR 22

SAN FRANCISCO, CA 94111-5998 FOR: CHEVRON ENERGY SOLUTIONS COMPANY

JULIA A. SOUDER RANDALL J. LITTENEKER
JAS ENERGIES ATTORNEY AT LAW

825 OAK STREET PAC

MARCEL HAWIGER

ENERGY ATTY
THE UTILITY REFORM NETWORK

785 MARKET ST., STE. 1400 SAN FRANCISCO, CA 94103

FOR: ALPHABET ENERGY INC.

FOR: TURN

WAYNE WAITE

POLICY DIR

CALIF. HOUSING PARTNERSHIP CORP. 369 PINE STREET, STE. 300 SAN FRANCISCO, CA 94104

FOR: CALIFORNIA HOUSING PARTNERSHIP

CORPORATION

JAMES FINE

ENVIRONMENTAL DEFENSE FUND 123 MISSION STREET, 28TH FLOOR SAN FRANCISCO, CA 94105

FOR: ENVIRONMENTAL DEFENSE FUND

MICHAEL ALCANTAR ATTORNEY AT LAW

BUCHALTER, A PROFESSIONAL CORPORATION

55 SECOND STREET, SUITE 1700 SAN FRANCISCO, CA 94105

FOR: COGENERATION ASSOCIATION OF

CALIFORNIA

JEANNE B. ARMSTRONG

ATTORNEY

GOODIN MACBRIDE SQUERI & DAY LLP 505 SANSOME STREET, SUITE 900 SAN FRANCISCO, CA 94111 FOR: SOLAR ENERGY INDUSTRIES

ASSOCIATION (SEIA)

MATTHEW VESPA STAFF ATTORNEY

EARTHJUSTICE

50 CALIFORNIA STREET, STE. 500 SAN FRANCISCO, CA 94111

FOR: SIERRA CLUB

MICHAEL B. DAY ATTORNEY

GOODIN MACBRIDE SQUERI DAY & LAMPREY LLP

505 SANSOME STREET, SUITE 900 SAN FRANCISCO, CA 94111-3133

FOR: COMMERCIAL ENERGY

NADIA MARQUEZ

MGR - POLICY & MARKETS ADVANCED MICROGRID SOLUTIONS 25 STILLMAN STREET, STE. 200 SAN FRANCISCO, CA 94114

FOR: ADVANCED MICROGRID SOLUTIONS (AMS)

ATTORNEY AT LAW
PACIFIC GAS AND ELECTRIC COMPANY

SAN FRANCISCO, CA 94117-2405 FOR: BALANCE ENERGY, A BAE SYSTEMS

INITIATIVE

PO BOX 7442, B30A SAN FRANCISCO, CA 94120 FOR: PACIFIC GAS AND ELECTRIC COMPANY

SARA STECK MYERS ATTORNEY AT LAW 122 28TH AVENUE SAN FRANCISCO, CA 94121

FOR: CENTER FOR ENERGY EFFICIENCY AND

RENEWABLE TECHNOLOGIES (CEERT)

BETH VAUGHAN
EXECUTIVE DIRECTOR
CALIFORNIA COGENERATION COUNCIL
4391 N. MARSH ELDER COURT
CONCORD, CA 94521

FOR: CALIFORNIA COGENERATION COUNCIL

BILL BROBECK BROBECK SOLAR ENERGY LLC 1125 BOLLINGER CANYON ROAD MORAGA, CA 94556

FOR: BROBECK SOLAR ENERGY LLC

GOPAL SHANKER
RECOLTE ENERGY
410 LAFATA STREET, SUITE 102
ST. HELENA, CA 94574
FOR: RECOLTE ENERGY

CHARLES SILVERMAN
FOUNDER & PRINCIPAL
SILVERMAN & LIGHT, INC.
1201 PARK AVE., STE. 100
EMERYVILLE, CA 94608
FOR: SILVERMAN & LIGHT, INC.

SHERIDAN PAUKER
PARTNER
KEYES & FOX, LLP
436 14TH STREET, SUITE 1305
OAKLAND, CA 94612
FOR: BOSCH BUILDING GRID TECHNOLOGIES
F/K/A ROBERT BOSCH, LLC

TIM LINDL COUNSEL KEYES & FOX LLP 436 14TH STREET, STE. 1305 OAKLAND, CA 94612

FOR: THE ALLIANCE FOR SOLAR CHOICE

ZEYNEP ERDAL
CAL. WASTEWATER CLIMATE CHANGE GROUP
155 GRAND AVE., STE. 800
OAKLAND, CA 94612
FOR: CALIFORNIA WASTEWATER CLIMATE
CHANGE GROUP

GREGORY MORRIS DIRECTOR GREEN POWER INSTITUTE 2039 SHATTUCK AVENUE, STE 402 TESLA, INC.
3500 DEER CREEK ROAD
PALO ALTO, CA 94304
FOR: TESLA, INC.

ENERGY POLICY ADVISOR

SARAH VAN CLEVE

STEPHEN RUMBAUGH
MEGAWATT ENERGY DEVELOPMENT
6360 CHATTSWOOD DR.
MARTINEZ, CA 94553
FOR: MEGAWATT ENERGY DEVELOPMENT

BOB TORRES
PRINCIPAL / SVP OF OPER
TRINCHERO FAMILY ESTATES
100 SOUTH MAIN STREET
ST. HELENA, CA 94574
FOR: SUTTER HOME WINERY

MATT EVANS
MANAGING DIR.
WATT TIME
1111 BROADWAY
OAKLAND, CA 94607
FOR: WATTTIME

BRIAN BALLEK
PRODUCT MGR.
AVALON BATTERY CORPORATION
426 17TH ST., STE. 700
OAKLAND, CA 94612
FOR: AVALON BATTERY CORPORATION

SUSANNAH CHURCHILL
CALIFORNIA DIR.
VOTE SOLAR
360 22ND STREET, SUITE 730
OAKLAND, CA 94612
FOR: VOTE SOLAR

TIM LINDT
KEYES & FOX LLP
436 14TH STREET, STE. 1305
OAKLAND, CA 94612
FOR: SUNRUN, INC.

ALEX J. MORRIS SR. DIR., POLICY & REGULATORY AFFAIRS CALIFORNIA ENERGY STORAGE ALLIANCE 2150 ALLSTON WAY, SUITE 400 BERKELEY, CA 94704 FOR: CALIFORNIA ENERGY STORAGE ALLIANCE (CESA)

REKHA RAO NEXTILITY 2015 SHATTUCK AVE., 5TH FLOOR BERKELEY, CA 94704

BERKELEY, CA 94704 FOR: GREEN POWER INSTITUTE FOR: NEXTILITY

STEPHANIE C. CHEN

DIR - ENERGY / TELECOMM POLICY THE GREENLINING INSTITUTE 1918 UNIVERSITY AVE., 2ND FL.

BERKELEY, CA 94704

FOR: THE GREENLINING INSTITUTE

CLYDE MURLEY

COMMUNITY COLLEGE LEAGUE OF CALIFORNIA

1031 ORDWAY STREET ALBANY, CA 94706

FOR: COMMUNITY COLLEGE LEAGUE OF

CALIFORNIA

JULIA A. LEVIN EXECUTIVE DIR.

BIOENERGY ASSOCIATION OF CALIFORNIA

PO BOX 6184 ALBANY, CA 94706

FOR: BIOENERGY ASSOCIATION F CALIFORNIA

JEAN WOO FOUNDER, CFO

CUSTOM POWER SOLAR, INC. 1442A WALNUT STREET, NO. 368

BERKELEY, CA 94709 FOR: CUSTOM POWER SOLAR

JOHN PROCTOR

PROCTOR ENGINEERING GROUP, LTD

418 MISSION AVENUE SAN RAFAEL, CA 94901

FOR: PROCTOR ENGINEERING GROUP LTD.

ARAM SHUMAVON

DISTRIBUTED ENERGY CONSUMER ADVOCATE

516 WHITEWOOD DR.

SAN RAFAEL, CA 94903-2455

FOR: DISTRIBUTED ENERGY CONSUMER

ADVOCATES

SUSAN E. BROWN ATTORNEY AT LAW

A WORLD INSTITUTE FOR SUSTAINABLE HUMANI

PO BOX 428

MILL VALLEY, CA 94942

FOR: A WISH

JENNIFER A. CHAMBERLIN

DIR. REG AFFAIRS - INT. DEMAND RESOURCES

JOHNSON CONTROLS, INC. 901 CAMPISI WAY, SUITE 260 CAMPBELL, CA 95008-2348 FOR: JOHNSON CONTROLS, INC.

CHRIS ROSS

DIRECTOR, GLOBAL SITE OPERATIONS

ADOBE SYSTEMS INC. 345 PARK AVENUE SAN JOSE, CA 95110

FOR: ADOBE SYSTEMS, INC.

HEIDI SICKLER

SR. ASSOCIATE, ENERGY & ENVIRONMENT SILICON VALLEY LEADERSHIP GROUP 2001 GATEWAY PLACE SUITE 101E

SAN JOSE, CA 95110

FOR: SILICON VALLEY LEADERSHIP GROUP

KELLY DESY SOLFOCUS, INC. 1841 ZANKER RD.

SAN JOSE, CA 95112-4213

FOR: SOLFOCUS, INC.

KARI SMITH

DIV. MGR. - REG. & LEGISLATIVE POLICY

SAN JOSE CLEAN ENERGY

200 EAST SANTA CLARA ST., 14TH FL.

SAN JOSE, CA 95113-1905 FOR: SAN JOSE CLEAN ENERGY

JULIE BLUNDEN SUNPOWER CORPORATION

77 RIO ROBLES SAN JOSE, CA 95134-1859

FOR: SUNPOWER CORPORATION

RYAN MCGRAW GENERAL COUNSEL

POWERWORKS WIND TURBINES LLC 15850P JESS RANCH ROAD

TRACY, CA 95377

FOR: POWERWORKS WIND TURBINES LLC

WOODY HASTINGS ENERGY PROGRAM MGR

CENTER FOR CLIMATE PROTECTION

PO BOX 3785

SANTA ROSA, CA 95402 FOR: THE CLIMATE CENTER NEAL M. REARDON

DIR - REGULATORY AFFAIRS SONOMA CLEAN POWER AUTHORITY 50 SANTA ROSA AVE. 5TH FL SANTA ROSA, CA 95404

FOR: SONOMA CLEAN POWER AUTHORITY

MICHAEL KYES 7423 SHAUN CT.

SEBASTOPOL, CA 95472 FOR: KGA ASSOCIATES LLC

JORDAN PINJUV SR. COUNSEL

CALIFORNIA INDEPENDENT SYSTEM OPERATOR

250 OUTCROPPING WAY FOLSOM, CA 95630 FOR: CALIFORONIA ISO MICHAEL SCHEIBLE
DEPUTY EXECUTIVE OFFICER
CALIFORNIA AIR RESOURCES BOARD
1001 I STREET
SACRAMENTO, CA 95677

FOR: CALIFORNIA AIR RESOURCES BOARD

ANDREA DEVEAU
EXECUTIVE DIR., CALIFORNIA
THE TECHNOLOGY NETWORK
1001 K STREET, 6TH FLOOR
SCARAMENTO, CA 95814
FOR: TECHNOLOGY NETWORK (TECHNET)

DAVID PEFFER
ATTORNEY
BRAUN BLAISING SMITH WYNNE, P.C.
555 CAPITOL MALL, SUITE 570
SACRAMENTO, CA 95814

FOR: MARIN CLEAN ENERGY (MCE)

KERRIANN SHEPPARD
CALIF PUBLIC UTILITIES COMMISSION
LEGAL DIVISION
300 Capitol Mall
Sacramento, CA 95814
FOR: CAL PA - PUBLIC ADVOCATES OFFICE
(FORMERLY ORA)

SIVA GUNDA
CALIFORNIA ENERGY COMMISSION
ENERGY ASSESSMENTS DIVISION
1516 9TH STREET, MS: 21
SACRAMENTO, CA 95814
FOR: CALIFORNIA ENERGY COMMISSION

JEDEDIAH J. GIBSON ATTORNEY ELLISON SCHNEIDER HARRIS & DONLAN LLP 2600 CAPITOL AVENUE, SUITE 400 SACRAMENTO, CA 95816-5931 FOR: BLOOM ENERGY, INC.

KAREN NORENE MILLS
ATTORNEY
CALIFORNIA FARM BUREAU FEDERATION
2600 RIVER PLAZA DRIVE
SACRAMENTO, CA 95833
FOR: CALIFORNIA FARM BUREAU FEDERATION

AMRUTA KHANOLKAR
PROJECT MANAGER
NEW BUILDINGS INSTITUTE
623 SW OAK ST
PORTLAND, OR 97205
FOR: NEW BUILDINGS INSTITUTE (NBI)

PHILIP UNDERCUFFLER
DIR. - PRODUCT MANAGEMENT & STRATEGIES
OUTBACK POWER TECHNOLOGIES
17825 9TH AVE., NE, STE. B
ARLINGTON, WA 98223

LON W. HOUSE, PH.D ASSN. OF CALIFORNIA WATER AGENCIES 4901 FLYING C RD. CAMERON PARK, CA 95682 FOR: ASSOCIATION OF CALIFORNIA WATER AGENCIES (ACWA)

BRAD HEAVNER
POLICY DIRECTOR
CALIFORNIA SOLAR & STORAGE ASSOCIATION
1107 9TH ST., NO. 820
SACRAMENTO, CA 95814
FOR: CALIFORNIA SOLAR & STORAGE
ASSOCIATION (CALSSA) F/K/A CALIFORNIA
SOLAR ENERGY INDUSTRIES ASSOCIATION
(CALSEIA)

JOHN KENNEDY
LEGISLATIVE ADVOCATE
RURAL COUNTY REPRESENTATIVES OF CALIF.
1215 K STREET, SUITE 1650
SACRAMENTO, CA 95814
FOR: RURAL COUNTY REPRESENTATIVES OF
CALIFORNIA

MICHAEL BOCCADORO
PRESIDENT
WEST COAST ADVISORS
925 L STREET, SUITE 800
SACRAMENTO, CA 95814
FOR: AGRICULTURAL ENERGY CONSUMERS
ASSOCIATION

ANDREW B. BROWN
ATTORNEY AT LAW
ELLISON SCHNEIDER HARRIS & DONLAN LLP
2600 CAPITOL AVENUE, SUITE 400
SACRAMENTO, CA 95816-5931
FOR: CONSTELLATION NEWENERGY, INC.

JAMES L. HODGES
ASSOC. OF CALIF. COMMUNITY & ENERGY SRV
1069 45TH STREET
SACRAMENTO, CA 95819
FOR: ASSOCIATION OF CALIFORNIA
COMMUNITY AND ENERGY SERVICES (ACCES)

ANN L. TROWBRIDGE
ATTORNEY
DAY CARTER & MURPHY LLP
3620 AMERICAN RIVER DRIVE, SUITE 205
SACRAMENTO, CA 95864
FOR: CALIFORNIA CLEAN DG COALITION

WILLIAM R. MARTINI
TECOGEN INC. - WESTERN OFFICE
2245 NW 111TH AVENUE
PORTLAND, OR 97229
FOR: TECOGEN INC.

FOR: OUTBACK POWER TECHNOLOGIES

Information Only

AIMEE BEASLEY

ENERGY SOLUTIONS EMAIL ONLY

EMAIL ONLY, CA 00000

ALEJANDRA BELALCAZAR

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

ALIA SCHOEN
PUBLIC POLICY MGR.
BLOOM ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

AMBER RIESENHUBER ENERGY ANALYST

INDEPENDENT ENERGY PRODUCERS ASSOC.

EMAIL ONLY

EMAIL ONLY, CA 00000

ANASTASSIA ZYGMUNT

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

ANDREA WOODALL

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

ANDREW ACE

MANAGER - CUSTOMER CARE PROGRAM PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

ANDREW ADAMS

VIRIDITY ENERGY, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

ANDREW TANNER

 ${\sf CHROMASUN\ INC.}$

EMAIL ONLY

EMAIL ONLY, CA 00000

ANDY BLAUVELT

MULTIFAMILY AFFORDABLE SOLAR HOMES

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: THE MULTIFAMILY AFFORDABLE SOLAR

HOMES COALITION

ANDY BLAUVELT

EAH HOUSING

EMAIL ONLY

EMAIL ONLY, CA 00000

BARBARA R. BARKOVICH

CONSULTANT

BARKOVICH & YAP, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

BEN FORD

OLD REDWOOD COMMONS ASSOCIATION

EMAIL ONLY

COTATI, CA 00000

BENJAMIN DAWSON

CALIFORNIA COMMUNITY CHOICE ASSOCIATION

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: CALCCA

BOB BLACKSTOCK SUNEDISON

EMAIL ONLY EMAIL ONLY, MD 00000 BOB HIGBIE EMAIL ONLY

EMAIL ONLY, CA 00000

BOB RAMIREZ

ITRON, INC. (CONSULTING & ANALYSIS DIV.)

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: ITRON, INC.

BOB WICHERT P. E.

DIR - TECH SERVICES

EMAIL ONLY

EMAIL ONLY, CA 00000

BRIAN JONES

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

BRIAN THEAKER

DIR - REGULATORY AFFAIRS NRG ENERGY, INC. - WEST

EMAIL ONLY

EMAIL ONLY, CA 00000

BURT Y. CHAO INSTITUTIONAL RESEARCH

SIMMONS & COMPANY INTERNATIONAL

EMAIL ONLY

EMAIL ONLY, TX 00000

REGULATORY ANALYST MARIN CLEAN ENERGY EMAIL ONLY

EMAIL ONLY, CA 00000

CARMELITA L. MILLER LEGAL COUNSEL

THE GREENLINING INSTITUTE

EMAIL ONLY

EMAIL ONLY, CA 00000

CAROLINE MCCORMACK

CALIFORNA HOUSING PARTNERSHIP

EMAIL ONLY

C.C. SONG

EMAIL ONLY, CA 00000

CATHERINE BUCKLEY

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

CHAD WORTH **ENERGY SOLUTIONS** EMAIL ONLY

EMAIL ONLY, CA 00000

CHUCK HORNBROOK

ITRON, INC. EMAIL ONLY

EMAIL ONLY, CA 00000

COLLIN TATEISHI

SUSTAINABLE HOUSING POLICY ANALYST CALIFORNIA HOUSING PARTNERSHIP

EMAIL ONLY

EMAIL ONLY, CA 00000

DAMON FRANZ

DIR - POLICY & ELECTRICITY MARKETS

TESLA, INC. EMAIL ONLY

EMAIL ONLY, CA 00000

DAN KAPLAN

SENIOR BUSINESS ANALYST

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

DARA SALOUR AESC

EMAIL ONLY

EMAIL ONLY, CA 00000

DAVID GIORDANO

DOOSAN FUEL CELL AMERICA

EMAIL ONLY

EMAIL ONLY

EMAIL ONLY, CA 00000

DAVID HUANG LEGAL FELLOW

THE GREENLINING INSTITUTE

EMAIL ONLY

EMAIL ONLY, CA 00000

DAVID SCHNEIDER SAFEPLUG

EMAIL ONLY, CA 00000

DAVID WIESNER

EMAIL ONLY

EMAIL ONLY, CA 00000

EBCE REGULATORY

EAST BAY COMMUNITY ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

EBONY ARENDT

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

EDWARD VINE

LAWRENCE BERKELEY NATIONAL LABORATORY

EMAIL ONLY

EMAIL ONLY, CA 00000

EILEEN COTRONEO

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

ELI HARLAND

CALIFORNIA ENERGY COMMISSION ENERGY RESEARCH & DEVELOPMENT DIV.

EMAIL ONLY

EMAIL ONLY, CA 00000

ELLIANA SPIEGEL

PACIFIC GAS & ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

ERIC J. PAUL

ALTA TERRA RESEARCH

EMAIL ONLY

EMAIL ONLY, CA 00000

ERIC VEIUM

PRINCIPAL, ENERGY ENGINEER

STOCKMAN'S ENERGY, INC.

ERIN GRIZARD BLOOM ENERGY, INC.

EMAIL ONLY

https://ia.cpuc.ca.gov/servicelists/R1211005_80602.htm

EMAIL ONLY EMAIL ONLY, CA 00000 EMAIL ONLY, CA 00000

FRANCIS RAQUEL

FOCAL POINT ENERGY, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

GREG MILLER ICE ENERGY EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: CENTER FOR SUSTAINABLE ENERGY SGIP

GREG RYBKA NEXANT

EMAIL ONLY

EMAIL ONLY, CA 00000

GREGORY CHIN

CALIFORNIA ENERGY COMMISSION

EMAIL ONLY

EMAIL ONLY, CA 00000

HEATHER E. RHOADS-WEAVER EFORMATIVE OPTIONS, LLC

EMAIL ON LY

EMAIL ONLY, WA 00000

HEIDI OCHSNER

THE CADMUS GROUP, INC.

EMAIL ONLY

EMAIL ONLY, OR 00000

HEWSON SHEPHERD

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

HOWARD E. SUSMAN

STOEL RIVES, LLP

EMAIL ONLY

EMAIL ONLY, CA 00000

JAN PEPPER

ELECTRIC DIV., MGR. SILICON VALLEY POWER

EMAIL ONLY

EMAIL ONLY, CA 00000

JANA KOPYCIOK ENERGY SOLUTIONS

EMAIL ONLY

EMAIL ONLY, CA 00000

JANA KOPYCIOK-LANDE SR. POLICY ANALYST

MARIN CLEAN ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

JANICE LIN
MANAGING PARTNER

STRATEGEN CONSULTING LLC

EMAIL ONLY

EMAIL ONLY, CA 00000

JAROD BISHOP

BUSINESS DEVELOPMENT MANAGER

EWT AMERICAS

EMAIL ONLY EMAIL ONLY, CA 00000 JASON D. LEGNER

EMAIL ONLY

EMAIL ONLY, CA 00000

JASON M. ACKERMAN BEST BEST & KRIEGER

EMAIL ONLY

EMAIL ONLY, CA 00000

JASON PERKINS

EMAIL ONLY

EMAIL ONLY, CA 00000

JAY CORRALES

TURNER REAL ESTATE

EMAIL ONLY

EMAIL ONLY, CA 00000

JEANINE COTTER

LUMINALT ENERGY CORPORATION

EMAIL ONLY

EMAIL ONLY, CA 00000

JEREMY WAEN

SR. REGULATORY ANALYST MARIN CLEAN ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

JESSIE DENVER PROGRAM MANAGER

EAST BAY COMMUNITY ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

JIM METROPULOS

OFFICE OF SENATOR LOIS WOLK

EMAIL ONLY

EMAIL ONLY, CA 00000

JOEL GAMORAN C3 ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

JOEY ANDERSON JONATHAN HART

DIRECT ENERGY CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY EMAIL ONLY

EMAIL ONLY, CA 00000 EMAIL ONLY, CA 00000

JORDAN NEWMAN

TIOGA ENERGY BLOOM ENERGY CORPORATION

EMAIL ONLY

EMAIL ONLY, CA 00000 EMAIL ONLY, CA 00000

JOSHUA WEINER
PRESIDENT
SEPISOLAR
EMAIL ONLY

EMAIL ONLY, CA 00000

JOY CALEBAUGH ACCOUNT MANAGER YOUR SOLAR MATE EMAIL ONLY

JOSH RICHMAN

EMAIL ONLY

EMAIL ONLY, CA 00000

JULIA JAZYNKA

ENERGY FREEDOM COALITION OF AMERICA

EMAIL ONLY EMAIL ONLY, CA 00000 KAREN HERTER

HERTER ENERGY RESEARCH SOLUTIONS

EMAIL ONLY

EMAIL ONLY, CA 00000

KAVYA BALARAMAN

STAFF WRITER / REPORTER CALIFORNIA ENERGY MARKETS

EMAIL ONLY

EMAIL ONLY, CA 00000

KELCEY GAVAR EMAIL ONLY

EMAIL ONLY, CA 00000

KEN WICKER

DIRECTOR, BUSINESS DEVELOPMENT FUELCELL ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

KEVIN D. BEST REAL ENERGY, LLC

EMAIL ONLY

EMAIL ONLY, CA 00000

KRISTIN GALLAGHER

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

LAURA CRUMP EMAIL ONLY

EMAIL ONLY, CA 00000

LES NELSON

IAPMO

EMAIL ONLY

EMAIL ONLY, CA 00000

LINDA GRAY

ENERGY STORAGE POLICY ADVISOR

CALIFORNIA SOLAR & STORAGE ASSOCIATION

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: CALIFORNIA SOLAR & STORAGE ASSOCIATION (CALSSA) F/K/A CALIFORNIA SOLAR ENERGY INDUSTRIES ASSOCIATION

LISA A. HERNANDEZ

SR. EXE MGR.

RHA INC.

EMAIL ONLY

LUPE KNOX

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

MAGGIE CHAN

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

AND ELECTRIC COMPANY

EMAIL ONLY, CA 00000

EMAIL ONLY, CA 00000

MARK WIEDEMAN EMAIL ONLY

EMAIL ONLY, CA 00000

MATT MEYER

MATTHEW CARLSON

CENTER FOR SUSTAINABLE ENERGY
EMAIL ONLY

CEO AOUAN

EMAIL ONLY, CA 00000

AQUANTA INC EMAIL ONLY

EMAIL UNLY, CA 00000

FOR: CENTER FOR SUSTAINABLE ENERGY (CSE) EMAIL ONLY, CA 00000

MCE REGULATORY MARIN CLEAN ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

MEGAN KIRKEBY

CALIFORNIA HOUSING PARTNERSHIP CORP.

EMAIL ONLY

EMAIL ONLY, CA 00000

MELISSA P. MARTIN SENIOR REGULATORY COUNSEL STATESIDE ASSOCIATES

EMAIL ONLY

EMAIL ONLY, VA 00000

MICHAEL KERKORIAN REAL POWER LLC EMAIL ONLY

EMAIL ONLY, CA 00000

MIKE GALLO

BUSINESS DEVELOPMENT MGR.

NEXTERA ENERGY EMAIL ONLY

EMAIL ONLY, CA 00000

MIKE LANDAU

CSI THERMAL PROGRAM MGR

SOUTHERN CALIFORNIA GAS COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

NANCY QUYNN EMAIL ONLY

EMAIL ONLY, CA 00000

NATALIE BUFFO

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

NEIL BLACK

CALIFORNIA BIOENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

NELLIE TONG SENIOR ANALYST KEMA, INC. EMAIL ONLY

EMAIL ONLY, CA 00000

NICK PAPPAS

OFFICE OF ASSEMBLYMAN NATHAN FLETCHER

EMAIL ONLY

EMAIL ONLY, CA 00000

NORA HAWKINS REGULATORY ANALYST

CPUC - ENERGY DIV.

EMAIL ONLY

EMAIL ONLY, CA 00000

PARRY BUCK NORCALWIND

EMAIL ONLY

EMAIL ONLY, CA 00000

PAUL NELSON CONSULTANT

BARKOVICH & YAP, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: CALIFORNIA LARGE ENERGY CONSUMERS

ASSOCIATION

RACHEL MCMAHON

EMAIL ONLY

EMAIL ONLY, CA 00000

RANDALL FISH SHIFTED ENERGY, INC.

EMAIL ONLY

EMAIL ONLY, HI 00000

REBECCA FEUERLICHT

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

RICHARD S. FLOOD

JOHNSON CONTROLS, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

ROBERT SPITZKA

WATER & ENERGY MANAGEMENT CO., INC.

EMAIL ONLY

EMAIL ONLY, CA 00000

ROBIN J. WALTHER

EMAIL ONLY

EMAIL ONLY, CA 00000

RYAN BERNARDO

BRAUN BLAISING MCLAUGHLIN, P.C.

EMAIL ONLY

EMAIL ONLY, CA 00000

SACHU CONSTANTINE

DIR - POLICY

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

FOR: CENTER FOR SUSTAINABLE ENERGY (FORMERLY CALIFORNIA CENTER FOR

SUSTAINABLE ENERGY)

SAMANTHA WEAVER PRINCIPAL REGULATORY ANALYST EAST BAY COMMUNITY ENERGY

EMAIL ONLY

EMAIL ONLY

EMAIL ONLY, CA 00000

EMAIL ONLY, CA 00000

SAMI MARDINI

DIR - PRODUCT MARKETING **ENERVAULT CORPORATION** EMAIL ONLY

EMAIL ONLY, CA 00000

SANJAY RANCHOD SARA BIRMINGHAM TESLA, INC.

DIRECTOR - WESTERN POLICY

SOLAR ENERGY INDUSTRIES ASSOCIATION

EMAIL ONLY

EMAIL ONLY, CA 00000

SCOTT BLAISING

BRAUN BLAISING MCLAUGHLIN P.C.

EMAIL ONLY

EMAIL ONLY, CA 00000

SHALINI SWAROOP REGULATORY COUNSEL MARIN CLEAN ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

SIERRA MARTINEZ

LEGAL DIR - CALIF. ENERGY PROJECT

NATURAL RESOURCES DEFENSE COUNCIL

EMAIL ONLY

EMAIL ONLY, CA 00000

SKIP FRALICK

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

SNULLER PRICE

ENERGY AND ENVIRONMENTAL ECONOMICS

EMAIL ONLY

EMAIL ONLY, CA 00000

STEVEN RYMSHA

DIR - GRID SOLUTIONS, PUBLIC POLICY

SUNRUN INC. EMAIL ONLY

EMAIL ONLY, AA 00000

SUMEETA GHAI

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

SUSAN BULLER

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

TIMOTHY O'KEEFE

ENERGY SOLUTIONS

EMAIL ONLY

EMAIL ONLY, CA 00000

TOM STEPIEN

PRIMUS POWER CORPORATION

EMAIL ONLY

EMAIL ONLY, CA 00000

TONY CATENACCI

SOUTHERN CALIFORNIA GAS COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000

UDI HELMAN

HELMAN ANALYTICS EMAIL ONLY

EMAIL ONLY, CA 00000

WAYNE BISHOP **PRESIDENT** CH4 ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000

WILLIAM MARIN ITRON, INC. EMAIL ONLY

EMAIL ONLY, CA 00000

DAVIS WRIGHT TREMAINE LLP

EMAIL ONLY

EMAIL ONLY, CA 00000

MRW & ASSOCIATES, LLC

EMAIL ONLY

EMAIL ONLY, CA 00000

CROWELL & MORING

EMAIL ONLY

EMAIL ONLY, CA 00000

MASH COALITION EMAIL ONLY

EMAIL ONLY, CA 00000

CAMERON-DANIEL, P.C.

EMAIL ONLY

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY, CA 00000 EMAIL ONLY

BENJAMIN AIRTH

EAMIL ONLY, CA 00000-0000

EMAIL ONLY, CA 00000-0000

FOR: GRID ALTERNATIVES

ERICA MACKIE, P.E.

GRID ALTERNATIVES

EMAIL ONLY

DAN PERKINS

VP-NATIONAL PRESEVATION PARTNERS

ENERGYSMARTHOMES.NET

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

FOR: CLEANTECH ENERGY SOLUTIONS INC.

IRENE M. STILLINGS EXECUTIVE DIRECTOR

CALIF. CTR. FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

JESSICA TELLEZ

PACIFIC GAS AND ELECTRIC COMPANY

EXECUTIVE DIRECTOR & CO-FOUNDER

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

KATRINA MORTON

CENTER FOR SUSTAINABLE ENERGY

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

NICK STIMMEL

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

RICHARD W. RAUSHENBUSH

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

ROBERT CHAN

PACIFIC GAS AND ELECTRIC COMPANY

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

SCOTT J. SACHS

ATKINSON ANDELSON LOYA RUUD & ROMO

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

THOMAS MILLHOFF

HELIO MICRO UTILITY, INC.

EMAIL ONLY

EMAIL ONLY, CA 00000-0000

ROBERT BENZ

EMAIL ONLY

EMAIL ONLY, CA 00000-2096

HEATHER TAKLE

DIRECTOR, CORPORATE DEVELOPMENT

AMERESCO INC EMAIL ONLY

EMAIL ONLY, MA 01701

ROBERT PANORA TECOGEN, INC. 45 FIRST AVENUE

WALTHAM, MA 02451

JENNIFER ARASIMOWICZ DIR - GOVN'T AFFAIRS

FUELCELL ENERGY, INC. 3 GREAT PASTURE ROAD

DANBURY, CT 06810

AMANDA CLEVINGER

SALES ANALYST BRIGHT POWER

11 HANOVER SQUARE, 21ST FL.

NEW YORK, NY 10005

PATRICK JOBIN

CREDIT SUISSE SECURITIES (USA) LLC

ONE MADISON AVENUE

NEW YORK, NY 10010

ASHLEY ARMSTRONG

A.O. SMITH

400 NORTH CAPITOL STREET, NW SUITE 585

WASHINGTON, DC 20001

EMILY N. SMITH ATTORNEY AT LAW

MCDERMOTT WILL & EMERY LLP 600 THIRTEENTH ST., NW

WASHINGTON, DC 20005

KATHERINE HOFFMASTER

AL LUNA

LITIGATION ASSISTANT, CLEAN ENERGY TEAM

EARTHJUSTICE

1625 MASSACHUSETTS AVE., N.W., STE 702

WASHINGTON, DC 20036-2243

SVEN L. ANDEN VP SALES AND MARKETING

ZEROPEX A.S. 368 WASHINGTON RD. SR. REGULATORY AFFAIRS ANALYST NEXTERA ENERGY RESOURCES 700 UNIVERSE BLVD., FEJ/JB

JUNO BEACH, FL 33405

KATRINA M. FRITZ KM FRITZ LLC PO BOX 79

OLD MISSION, MI 49673

GROSSE POINTE, MI 48230-1616

JIM ROSS RCS, INC. 266 PENNINGTON LANE CHESTERFIELD, MO 63005

JAMES HARRIS UPSTAR ENERGY LLC 4131 CENTURION WAY ADDISON, TX 75001

THOMAS MULHOLLAND
CARBON CYCLE ENERGY, LLC
5755 CENTRAL AVE., STE. C
BOULDER, CO 80301
FOR: CARBON CYCLE ENERGY, LLC (C2E)

LORI SMITH SCHELL EMPOWERED ENERGY 174 N. ELK RUN DURANGO, CO 81303

ADRIAN MARTINEZ SOUTHERN CALIFORNIA GAS COMPANY 555 W. 5TH STREET LOS ANGELES, CA 90013

DALE FONTANEZ PROJECT MGR. SOUTHERN CALIFORNIA GAS COMPANY 555 W. FIFTH STREET LOS ANGELES, CA 90013

PAMELA WU REGULATORY CASE MGR. SOUTHERN CALIFORNIA GAS COMPANY 555 W. FIFTH STREET, GT14D6 LOS ANGELES, CA 90013

SID NEWSOME REGULATORY MGR. SOUTHERN CALIFORNIA GAS COMPANY 555 WEST FIFTH STREET, GT14D6 LOS ANGELES, CA 90013-1011

AARON DYER BUSINESS UNIT MANAGER ROMEO SYSTEMS, INC. 4380 AYERS AVE VERNON, CA 90058

SID CARTER ROMEO POWER TECHNOLOGY 4380 AYERS AVENUE VERNON, CA 90058

NORMAN A. PEDERSEN, ESQ. ATTORNEY HANNA AND MORTON LLP 444 S. FLOWER STREET, SUITE 1500 LOS ANGELES, CA 90071-2916 MIKE BERGEY
PRESIDENT
DISTRIBUTED WIND ENERGY ASSOCIATION
2200 INDUSTRIAL BLVD.
NORMAN, OK 73069

KELLY CRANDALL EQ RESEARCH, LLC 1580 LINCOLN STEET, SUITE 880 DENVER, CO 80203

KEVIN COONEY
PRINCIPAL/CEO
SUMMIT BLUE CORPORATION
1375 WALNUT ST., STE. 200
BOULDER, CO 80302-5242

MONA TIERNEY-LLOYD SR. DIR., WESTERN REG. AFFAIRS ENEL X NORTH AMERICA, INC. 2071 ALTAIR LAVE RENO, NV 89521

BLAINE WAYMIRE SOUTHERN CALIFORNIA GAS COMPANY 555 W. 5TH STREET, GT20B8 LOS ANGELES, CA 90013

EDITH MORENO SOUTHERN CALIFORNIA GAS COMPANY 555 W. 5TH STREET, GT14D6 LOS ANGELES, CA 90013

REGINA LUGANI MANAGER, PROGRAM LEVERAGING GRP SOUTHERN CALIFORNIA GAS COMPANY 555 WEST FIFTH STREET, GT-2089 LOS ANGELES, CA 90013

BYRON CHAN
ASSOCIATE ATTORNEY
EARTHJUSTICE
800 WILSHIRE BOULEVARD, SUITE 1000
LOS ANGELES, CA 90017

RAVI SANKARAN DIR - BUSINESS DEVELOP. ROMEO SYSTEMS, INC. 4380 AYERS AVENUE VERNON, CA 90058

JOHN A. BRIDGE HOGAN LOVELLS US LLP 1999 AVENUE OF THE STARS, SUITE 1400 LOS ANGELES, CA 90067

RICHARD R. KRIETE
CALNETIXPS
324 MARINE AVENUE
MANHATTAN BEACH, CA 90266

ARLEEN NOVOTNEY SOUTHERN CALIFORNIA FORUM 941 PALMS BLVD. VENICE, CA 90291 YING-NIEN YU, P.E. CONSERVATION TECHNOLOGY 1018 OXFORD AVENUE MARINA DEL REY, CA 90292

ANDREW MEYER
CO-FOUNDER
SWELL ENERGY
1515 7TH ST., NO. 049
SANTA MONICA, CA 90401

RICK RUIZ ZENVIRONMENT 209 MONTANA AVENUE, NO. 307 SANTA MONICA, CA 90403

TED BARDACKE
SENIOR ASSOCIATE
GLOBAL GREEN USA
2218 MAIN STREET, 2ND FLOOR
SANTA MONICA, CA 90405

TONJA WICKS
TLW LEGAL & GOV'T. SUPPORT SVCS.
5318 E. 2ND STREET, NO.703
LONG BEACH, CA 90803

REN ZHANG CITY OF PASADENA WATER AND POWER 150 S. LOS ROBLES AVENUE, SUITE 200 PASADENA, CA 91101 ERIC KLINKNER
PASADENA DEPARTMENT OF WATER AND POWER
150 SOUTH LOS ROBLES AVENUE, SUITE 200
PASADENA, CA 91101-2437

DANIEL W. DOUGLASS ATTORNEY DOUGLASS, LIDDELL & KLATT 4766 PARK GRANADA, SUITE 209 CALABASAS, CA 91302 FOR: EDF RENEWABLES, INC. GREGORY KLATT DOUGLASS & LIDDELL 4766 PARK GRANADA, STE. 209 CALABASAS, CA 91302

DEBORAH SCHLAFF
OPERATIONS MANAGER
GREEN SOLAR TECHNOLOGIES
6400 LAUREL CANYON BLVD.
NORTH HOLLYWOOD, CA 91606

MICHAEL TEN EYCK RANCHO CUCAMONGA MUNICIPAL UTILITY 10500 CIVIC CENTER DRIVE RANCHO CUCAMONGA, CA 91730

AILEEN M. LAGBAO SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVE. ROSEMEAD, CA 91770 CASE ADMINISTRATION SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE, RM. 370 ROSEMEAD, CA 91770

GARY BARSLEY SOUTHERN CALIFORNIA EDISON COMPANY 2244 WANUT GROVE AVE., PO BOX 800 ROSEMEAD, CA 91770 MICHAEL D. MONTOYA SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE, PO BOX 800 ROSEMEAD, CA 91770

MICHAEL TOMLIN SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE ROSEMEAD, CA 91770 NICOLE REICHENBERGER SOUTHERN CALIFORNIA EDISON COMPANY 8631 RUSH STREET ROSEMEAD, CA 91770

REBECCA MEIERS-DE PASTINO SR. ATTORNEY SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE / PO BOX 800 ROSEMEAD, CA 91770 FOR: SOUTHERN CALIFORNIA EDISON COMPANY SHIELA LINAO SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVENUE ROSEMEAD, CA 91770

JENNIFER TSAO SHIGEKAWA ASSISTANT GENERAL COUNSEL SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVE. / PO BOX 800 ROSEMEAD, CA 91770-3714 RONALD K. ISHII
AESC, INC.
5927 BALFOUR COURT, SUITE 213
CARLSBAD, CA 92008

DHAD CHAHBAZI BAP POWER CORP. D/B/A/CENERGY POWER 3176 LIONSHEAD AVE., NO. 11 CARLSBAD, CA 92010

ADAM RHEAD
MILHOLLAND ELECTRIC INC.
1475 NORTH CUYAMACA STREET
EL CAJON, CA 92020

KEITH DAVIDSON DE SOLUTIONS 732 VAL SERENO DRIVE ENCINITAS, CA 92024

JOE SILVA CALIFORNIA POWER PARTNERS, INC. 13000 DANIELSON ST., STE. J POWAY, CA 92029

FREDERICK M. ORTLIEB
DEPUTY CITY ATTORNEY
CITY OF SAN DIEGO
1200 THIRD AVENUE, SUITE 1100
SAN DIEGO, CA 92101

MORGAN LEE U-T SAN DIEGO 350 CAMINO DE LA REINA SAN DIEGO, CA 92108

MARCIE A. MILNER
VP - REG AFFAIRS
SHELL ENERGY NORTH AMERICA (US), L.P.
4445 EASTGATE MALL, STE. 100
SAN DIEGO, CA 92121

LISA BICKER CLEANTECH SAN DIEGO 9191 TOWNE CENTRE DRIVE. STE. 410 SAN DIEGO, CA 92122-1227

CHRISTA M. LIM
REGULATORY CASE MGR.
SAN DIEGO GAS & ELECTRIC COMPANY
8330 CENTURY PARK COURT, CP32D
SAN DIEGO, CA 92123

JAMES STRANGE
CENTER FOR SUSTAINABLE ENERGY
EMAIL ONLY
EMAIL ONLY, CA 92123

LAURA EARL SR. COUNSEL - REGULATORY SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTURY PARK CT., CP32D SAN DIEGO, CA 92123

ROSALINDA MAGANA SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTRY PARK COURT, CP32D RICHARD T. SPERBERG ONSITE ENERGY CORPORATION 2701 LOKER AVE. WEST., SUITE 107 CARLSBAD, CA 92010

FRANK J. MAZANEC BIOFUELS ENERGY LLC 2211 ENCINITAS BLVD. ENCINITAS, CA 92024

JEFF COX 1557 MANDEVILLE PLACE ESCONDIDO, CA 92029

TY TOSDAL ATTORNEY TOSDAL LAW FIRM 777 S. HIGHWAY 101, SUITE 215 SOLANA BEACH, CA 92075

PETER HAMILTON
BETTER ENERGIES LLC
3044 IVY ST.
SAN DIEGO, CA 92104
FOR: SHARP ELECTRONICS CORPORATION

TRACY TSUI SOLAR EDGETECH 7710 BALBOA AVE., SUITE 218-B SAN DIEGO, CA 92111

PILOT POWER GROUP, INC. 8910 UNIVERSITY CENTER LANE, STE 520 SAN DIEGO, CA 92122

CENTRAL FILES SDG&E / SOCALGAS 8330 CENTURY PARK COURT, CP31-E SAN DIEGO, CA 92123

CLAY FABER SOUTHERN CALIFORNIA GAS CO. 8330 CENTURY PARK CT., CP32D SAN DIEGO, CA 92123

JOSEPH BICK CENTER FOR SUSTAINABLE ENERGY EMAIL ONLY EMAIL ONLY, CA 92123

MICHELLE SOMERVILLE REGULATORY CASE MGR. SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTURY PARK COURT, CP32F SAN DIEGO, CA 92123

SHEWIT WOLDEGIORGIS
MGR - REGULATORY AFFAIRS
SAN DIEGO GAS & ELECTRIC COMPANY

SAN DIEGO, CA 92123

8330 CENTURY PARK CT., CP32F SAN DIEGO, CA 92123

STEVE LANGO SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTURY PARK CT., CP32E SAN DIEGO, CA 92123 WILLIAM FULLER
CALIFORNIA REGULATORY AFFAIRS
SAN DIEGO GAS & ELECTRIC COMPANY
8330 CENTURY PARK COURT, CP32F
SAN DIEGO, CA 92123

CAROL MANSON
SAN DIEGO GAS & ELECTRIC CO.
8330 CENTURY PARK COURT CP32D
SAN DIEGO, CA 92123-1530

DAVID HENRY
CITY OF SAN DIEGO
MS 1101B
9601 RIDGEHAVEN COURT, SUITE 120
SAN DIEGO, CA 92123-1636

LORIE COSIO-AZAR
INTERIM DEPUTY DIRECTOR
CITY OF SAN DIEGO
9601 RIDGEHAVEN COURT, SUITE 120
SAN DIEGO, CA 92123-1636

DAVID DEMAREST SOLARE ENERGY, INC. 9520 PADGETT STREET, SUITE 107 SAN DIEGO, CA 92126

JONATHAN HART
MGR - VIRTUAL POWER PLANT
EDF RENEWABLES, INC.
15445 INNOVATION DRIVE
SAN DIEGO, CA 92128

LAURA J. MANZ LJ MANZ CONSULTING 12372 AVENIDA CONSENTIDO SAN DIEGO, CA 92128

ROBERT A. LAURIE
ASSISTANT GENERAL COUNSEL - ENERGY
IMPERIAL IRRIGATION DISTRICT
333 E.BARIONI BLVD.
IMPERIAL, CA 92251

JAMES A. WELLS III DIR - ENGINEERING SHOREBREAK ENERGY DEVELOPERS, LLC 1 PETERS CANYON RD., STE. 110 IRVINE, CA 92606

JAY PURCELL
PRESIDENT
ENGENUUS ENERGY LLC
31741 VIA PERDIZ
COTO DE CAZA, CA 92679

STEVEN SCIORTINO CITY OF ANAHEIM 200 SOUTH ANAHEIM BOULEVARD ANAHEIM, CA 92805

SHAUN YEAGER MOHR POWER SOLAR, INC 1454 POMONA ROAD CORONA, CA 92882 DAVID ALBERS
BIOENERGY SOLUTIONS, LLC
PO BOX 2282
BAKERSFIELD, CA 93303-2282

KEVIN FLANAGAN THINKWIRE ENERGY SERVICES, INC. PO BOX 16015 FRESNO, CA 93755 KEN DURSA RECORDING SECRETARY CCCCWE 145 HARVEST STREET SALINAS, CA 93901

THOMAS R. ADCOCK
PRESIDENT
ALISAL WATER CORPORATION
249 WILLIAMS ROAD
SALINAS, CA 93905
FOR: ALISAL WATER CORPORATION D/B/A
ALCO WATER SERVICE/CALIFORNIA UTILTIES
SERVICE, INC.

ARTHUR MCLOUGHLIN
CO-CHAIR
CCCCWE
9540 S. CENTURY OAK RD.
SALINAS, CA 93907
FOR: CENTRAL COAST COALITION OF
COMMUNITIES FOR WASTEWATER EQUITY

JOHN PIMENTEL FOUNDATION WINDPOWER 431 BURGESS DR., STE. 200 MENLO PARK, CA 94025-3478 KEVIN HAUCK FOUNDATION WINDPOWER 431 BURGESS DR., STE. 200 MENLO PARK, CA 94025-3478

JULIE STEURY STEM, INC. 100 ROLLINS ROAD MILLBRAE, CA 94030 MATTHEW PERRY STEM, INC. 100 ROLLINS ROAD MILLBRAE, CA 94030

MANI THOTHADRI COGENRA SOLAR 365 E. MIDDLEFIELD RD. MOUNTAIN VIEW, CA 94040 BRUCE KARNEY 833 BUSH STREET MOUNTAIN VIEW, CA 94041

JOHN KILKENNY SKYWATCH ENERGY 365 E. MIDDLEFIELD ROAD MOUNTAIN VIEW, CA 94043 RATSON MORAD COGENRA SOLAR, INC. 365 E. MIDDLEFIELD RD. MOUNTAIN VIEW, CA 94043

SUE MARA CONSULTANT RTO ADVISORS, LLC 164 SPRINGDALE WAY REDWOOD CITY, CA 94062 HILARY STAVER MGR - REGULATORY & LEGISLATIVE AFFAIRS SILICON VALLEY CLEAN ENERGY AUTHORITY 333 W. EL CAMINO REAL, STE. 290 SUNNYVALE, CA 94087 FOR: SILICON VALLEY CLEAN ENERGY AUTHORITY

PAUL DETERING REDWOOD INSIGHTS LLC 1160 SHENANDOAH DR. SUNNYVALE, CA 94087

ALEC WARD CALIF PUBLIC UTILITIES COMMISSION ELECTRICITY PRICING AND CUSTOMER PROGRAM AREA 505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

CATHLEEN A. FOGEL CALIF PUBLIC UTILITIES COMMISSION ADMINISTRATIVE LAW JUDGE DIVISION **ROOM 5101**

CHARLES WARD CALIF PUBLIC UTILITIES COMMISSION DEMAND RESPONSE, CUSTOMER GENERATION, AN AREA 505 VAN NESS AVENUE

505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

SAN FRANCISCO, CA 94102-3214

CHRISTOPHER WESTLING CALIF PUBLIC UTILITIES COMMISSION DEMAND RESPONSE, CUSTOMER GENERATION, AN COMMISSIONER RECHTSCHAFFEN AREA 505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

SANDY GOLDBERG CALIF PUBLIC UTILITIES COMMISSION ROOM 5202 505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

STEPHEN CASTELLO CALIF PUBLIC UTILITIES COMMISSION ELECTRICITY PRICING AND CUSTOMER PROGRAM ADMINISTRATIVE LAW JUDGE DIVISION

CALIF PUBLIC UTILITIES COMMISSION ROOM 5005

505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

VALERIE KAO

JAMES R. GROW

CANDICE YU

ANGLEA PATANE S.F. PUC POWER ENTERPRISE 525 GOLDEN GATED AVE. SAN FRANCISCO, CA 94102-3220 FRASER D. SMITH CITY AND COUNTY OF SAN FRANCISCO SAN FRANCISCO PUBLIC UTILITIES COMM 525 GOLDEN GATE AVE., 7TH FLOOR SAN FRANCISCO, CA 94102-3220

ADAM GERZA SULLIVAN SOLAR POWER OF CALIFORNIA, INC. 169 11TH STREET SAN FRANCISCO, CA 94103

DEPUTY DIRECTOR NATIONAL HOUSING LAW PROJECT 703 MARKET ST., STE. 2000 SAN FRANCISCO, CA 94103

ANDRE DEVILBISS ASSOCIATE, DEVELOPMENT

ASSOCIATE, ADVANCED SOLUTIONS

RECURRENT ENERGY 300 CALIFORNIA STREET, 8TH FLOOR SAN FRANCISCO, CA 94104

JULIA KANTOR ATTORNEY KEYES & FOX LLP 580 CALIFORNIA STREET, 12TH FL. SAN FRANCISCO, CA 94104

NOEMI GALLARDO ATTORNEY SUNRUN INC. 225 BUSH ST., STE 13 SAN FRANCISCO, CA 94104

SRINIDHI S. KUMAR
PROGRAM MGR.
CALIFORNIA HOUSING PARTNERSHIP
369 PINE STREET, STE. 300
SAN FRANCISCO, CA 94104
FOR: CALIFORNIA HOUSING PARTNERSHIP

ANA GONZALEZ
SR. CASE MGR - REGULATORY
PACIFIC GAS AND ELECTRIC COMPANY
77 BEALE STREET, MC B9A / PO BOX 770000
SAN FRANCISCO, CA 94105

BUCHALTER DOCKET BUCHALTER 55 SECOND STREET, SUITE 1700 SAN FRANCISCO, CA 94105

DAVID POSTER
PACIFIC GAS AND ELECTRIC COMPANY
245 MARKET STREET
SAN FRANCISCO, CA 94105

HAROLD HIRSCH PACIFIC GAS AND ELECTRIC COMPANY 245 MARKET STREET SAN FRANCISCO, CA 94105

JOSEPHINE WU PACIFIC GAS AND ELECTRIC COMPANY 77 BEALE STREET, MC B9A, RM. 975 SAN FRANCISCO, CA 94105

LAMINE AKABA PACIFIC GAS AND ELECTRIC COMPANY 245 MARKET STREET, MC N4Q SAN FRANCISCO, CA 94105

MICHAEL CADE
ANALYST
BUCHALTER, A PROFESSIONAL CORPORATION
55 SECOND STREET, SUITE 1700
SAN FRANCISCO, CA 94105

RYAN CHAN PACIFIC GAS AND ELECTRIC COMPANY SUNEDISON 44 MONTGOMERY STREET, 2200 SAN FRANCISCO, CA 94104

MERRIAN BORGESON SR. SCIENTIST, ENERGY PROGRAM NATURAL RESOURCES DEFENSE COUNCIL 111 SUTTER ST., 21ST FL. SAN FRANCISCO, CA 94104

PIERRE DELFORGE SENIOR SCIENTIST NATURAL RESOURCES DEFENSE COUNCIL 111 SUTTER ST., 21ST FLOOR SAN FRANCISCO, CA 94104

ADAM FORNI SENIOR RESEARCH ANALYST NAVIGANT CONSULTING, INC. ONE MARKET STREET, SUITE 1200 SAN FRANCISCO, CA 94105

ARCADY SOSINOV SECRETARY REPURPOSED BATTERY ALLIANCE 403 MAIN ST., NO. 306 SAN FRANCISCO, CA 94105 FOR: REPURPOSED BATTERY ALLIANCE

CHRIS KATO
PACIFIC GAS AND ELECTRIC COMPANY
245 MARKET STREET, N6G
SAN FRANCISCO, CA 94105

DEREK JONES
NAVIGANT CONSULTING, INC.
ONE MARKET ST., SPEAR TOWER, SUITE 1200
SAN FRANCISCO, CA 94105

JEFFREY KLOMPUS NEWCOMB ANDERSON MCCORMICK 201 MISSION ST., STE. 2000 SAN FRANCISCO, CA 94105

KARIN CORFEE
MANAGING DIRECTOR
NAVIGANT CONSULTING INC
1 MARKET ST., SPEAR ST. TWR, ST. 1200
SAN FRANCISCO, CA 94105

LISA LONG
PACIFIC GAS AND ELECTRIC COMPANY
245 MARKET STREET
SAN FRANCISCO, CA 94105

PETER VAN MIEGHEM, ESQ. ATTORNEY PACIFIC GAS AND ELECTRIC COMPANY 77 BEALE STREET, B30A, RM 3165 SAN FRANCISCO, CA 94105

SARINA URIZA
PACIFIC GAS AND ELECTRIC COMPANY

245 MARKET STREET SAN FRANCISCO, CA 94105 77 BEALE ST., B9A SAN FRANCISCO, CA 94105

SAVI ELLIS

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE STREET

SAN FRANCISCO, CA 94105

SHANNON GRAHAM ASSOCIATE DIRECTOR NAVIGANT CONSULTING, INC.

ONE MARKET, SPEAR STREET TOWER, STE 1200

SAN FRANCISCO, CA 94105

STEPHANIE YANG

PACIFIC GAS AND ELECTRIC COMPANY 245 MARKET STREET, MC N4Q

SAN FRANCISCO, CA 94105

TESSA CARLBERG

COUNSEL

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE STREET, B30A SAN FRANCISCO, CA 94105

BUCHALTER, A PROFESSIONAL CORPORATION

55 SECOND STREET, SUITE 1700 SAN FRANCISCO, CA 94105

EVELYN KAHL ATTORNEY

BUCHALTER, A PROFESSIONAL CORPORATION

55 SECOND STREET, SUITE 1700 SAN FRANCISCO, CA 94105-3493 FOR: ENERGY PRODUCERS AND USERS

COALITION

LILLIAN RAFII

ATTORNEY

BUCHALTER, A PROFESSIONAL CORPORATION 55 SECOND STREET, STE. 1700 SAN FRANCISCO, CA 94105-3493

COLIN EYKAMP TIOGA ENERGY

123 MISSION STREET, FLOOR 89 SAN FRANCISCO, CA 94105-5124

PAUL DETERING

TIOGA ENERGY

123 MISSION STREET, FLR. 9 SAN FRANCISCO, CA 94105-5124 ANDY SCHWARTZ TESLA, INC.

444 DE HARO ST., STE. 101 SAN FRANCISCO, CA 94107

CHESTER A. ROAMAN 1180 DE HARO

SAN FRANCISCO, CA 94107

DANIEL CHIA DIR.

TESLA, INC.

444 DE HARO ST., STE. 101 SAN FRANCISCO, CA 94107

FRANCESCA WAHL

SR. POLICY ASSOCIATE, BUS. DEVELOPMENT

TESLA, INC.

444 DE HARO STREET, STE. 101 SAN FRANCISCO, CA 94107

JEFF PALMER TESLA, INC.

444 DE HARO ST., STE. 101 SAN FRANCISCO, CA 94107

NICHOLAS CONNELL

SR. MGR - GOV'T AFFAIRS ADVANCED MICROGRID SOLUTIONS 25 STILLMAN STREET, SUITE 200 SAN FRANCISCO, CA 94107

SANJINA MALPANI

ADVANCED MICROGRID SOLUTIONS

25 STILLMAN STREET SAN FRANCISCO, CA 94107

STEVEN MOSS

SAN FRANCISCO COMMUNITY POWER CO-OP 2325 THIRD STREET, STE 344 SAN FRANCISCO, CA 94107

BRIAN T. CRAGG

GOODIN, MACBRIDE, SQUERI, DAY & LAMPREY

505 SANSOME STREET, SUITE 900 SAN FRANCISCO, CA 94111

DAVID SIDDIQUI ORACLE / OPOWER

475 SANSOME ST 15TH FLOOR SAN FRANCISCO, CA 94111

JILL N. JAFFE NOSSAMAN LLP

50 CALIFORNIA STREET, 34TH FLOOR

SAN FRNACISCO, CA 94111

JOHN CLARK

SETH D. HILTON GOODIN MACBRIDE SQUERI DAY & LAMPREY LLP ATTORNEY

https://ia.cpuc.ca.gov/servicelists/R1211005_80602.htm

505 SANSOME STREET, SUITE 900 SAN FRANCISCO, CA 94111 FOR: EVERYDAY ENERGY, INC.

STEVE SHERR SVP BUSINESS AFFAIRS & GEN. COUNSEL

FOUNDATION WINDPOWER, LLC 505 SANSOME STREET, STE 450 SAN FRANCISCO, CA 94111 FOR: FOUNDATION WINDPOWER, LLC

MARTIN MATTES NOSSAMAN LLP 50 CALIFORNIA STREET, STE. 3400 SAN FRANCISCO, CA 94111-4799

MANAL YAMOUT VP - POLICY & MARKETS ADVANCED MICROGRID SOLUTIONS 25 STILLMAN STREET SAN FRANCISCO, CA 94114

BRIAN ORION LAWYERS FOR CLEAN ENERGY 656A CLAYTON STREET SAN FRANCISCO, CA 94117

NADIA PABST WEST ROAD STRATEGIES 507 HAIGHT STREET SAN FRANCISCO, CA 94117

BETSY WILKINS WILKINS COMMUNICATIONS 815 25TH AVE., NO. 204 SAN FRANCISCO, CA 94121

BRIAN BISHOP PACIFIC GAS AND ELECTRIC COMPANY 245 MARKET ST., MC N3F SAN FRANCISCO, CA 94177

RON MORENO, JR. PACIFIC GAS AND ELECTRIC COMPANY 245 MARKET STREET, MC N3F SAN FRANCISOC, CA 94177

CHRIS KING SIEMENS SMART GRID SOLUTIONS 4000 E. THIRD AVE., STE. 400 FOSTER CITY, CA 94404

PHILIP GLECKMAN CHIEF EXECUTIVE OFFICER SUNVAPOR INC. 1911 SECOND ST. LIVERMORE, CA 94550

MATTHEW BARMACK DIR. - MARKET & REGULATORY ANALYSIS STOEL RIVES, LLP THREE EMBARCADERO CENTER, STE. 1120 SAN FRANCISCO, CA 94111

TARA S. KAUSHIK HOLLAND & KNIGHT, LLP 50 CALIFORNIA STREET, 28TH FLOOR SAN FRANCISCO, CA 94111

ALLIE DETRIO CHIEF STRATEGIST REIMAGINE POWER 77 SALA TERRACE SAN FRANCISCO, CA 94112

MEGHA LAKHCHAURA DIR. PUBLIC POLICY SUNRUN INC 595 MARKET STREET SAN FRANCISCO, CA 94115

CHRIS RAPHAEL EDITOR CALIFORNIA ENERGY MARKETS 425 DIVISADERO ST., STE. 303 SAN FRANCISCO, CA 94117

HILLARY CORRIGAN CALIFORNIA ENERGY MARKETS 425 DIVISADERO ST. STE 303 SAN FRANCISCO, CA 94117-2242

MEGAN M. MYERS **ATTORNEY** LAW OFFICES OF SARA STECK MYERS 122 - 28TH AVENUE SAN FRANCISCO, CA 94121

CASE COORDINATION PACIFIC GAS AND ELECTRIC COMPANY PO BOX 770000, MC B9A SAN FRANCISCO, CA 94177

GRANT KOLLING SENIOR ASSISTANT CITY ATTORNEY CITY OF PALO ALTO PO BOX 10250 PALO ALTO, CA 94303

JAY C. MCLAUGHLIN FATHOM ENGINEERING 4843 SOUTH POINT DISCOVERY BAY, CA 94505

JEAN MERRIGAN ADVOCATE WOMEN'S ENERGY MATTERS PO BOX 2615 MARTINEZ, CA 94553

SCOTT WAYLAND, P.E. WYLAND ENGINEERING, INC.

CALPINE CORPORATION 4160 DUBLIN BLVD., SUITE 100 DUBLIN, CA 94568

424 MELROSE COURT SAN RAMON, CA 94582

TRACY PORTER PREMIERE SOLUTIONS, LLC 4695 CHABOT DRIVE, STE. 200

PLEASANTON, CA 94588

ADAM ROSS BROWN AND CALDWELL 201 N CIVIC DR., STE. 115 WALNUT CREEK, CA 94596

GAVIN MCCORMICK WATT TIME 1111 BROADWAY, FLOOR 3 OAKLAND, CA 94607

JERRY LAHR PROGRAM MANAGER ABAG POWER 101 EIGHT STREET OAKLAND, CA 94607-4756

CATHLEEN MONAHAN SR. PROGRAMS DIR. **GRID ALTERNATIVES** 1171 OCEAN AVENUE, SUITE 200 OAKLAND, CA 94608

ERANDI RATNAYAKE PROGRAM MANAGER **GRID ALTERNATIVES** 1171 OCEAN AVE STE 200 OAKLAND, CA 94608

STEPHEN CAMPBELL PROJECT MGR. GRID ALTERNATIVES 1171 OCEAN AVE., SUITE 200 OAKLAND, CA 94608

STANLEY GRESCHNER GRID ALTERNATIVES 1171 OCEAN AVENUE, NO. 200 OAKLAND, CA 94608-1147

MONIKA WEISS ERGSOL 10033 BROADWAY TERRACE OAKLAND, CA 94611

WILLIAM L. SCOTT 6895 EXETER DRIVE OAKLAND, CA 94611

ANDREA MERCADO **ENERGY SOLUTIONS** 449 15TH STREET SUITE 400 OAKLAND, CA 94612

W. BRYCE CHASTAIN ATKINSON ANDELSON LOYA RUUD & ROMO 5075 HOPYARD RD., STE. 210 PLEASANTON, CA 94588-2797

KATE PISCOPO SENIOR ANALYST TIERRA RESOURCE CONSULTANTS, LLC 1501 BROADWAY, SUITE 300 WALNUT CREEK, CA 94596

HENRY RICHARDSON WATTIME 1111 BROADWAY, 3RD FLOOR OAKLAND, CA 94607

ADAM BUCHHOLZ **GRID ALTERNATIVES** 1171 OCEAN AVE., SUITE 200 OAKLAND, CA 94608

ELISE HUNTER DIR - POLICY & REGULATORY AFFAIRS **GRID ALTERNATIVES** 1171 OCEAN AVE., STE. 200 OAKLAND, CA 94608

KRISTIN CHARIPAR **GRID ALTERNATIVES** 1171 OCEAN AVE., SUITE 200 OAKLAND, CA 94608

ZOEY BURROWS PROGRAM MGR **GRID ALTERNATIVES** 1171 OCEAN AVENUE OAKLAND, CA 94608

ZACH FRANKLIN CHIEF STRATEGY OFFICER GRID ALTERNATIVES 1171 OCEAN AVE., STE. 200 OAKLAND, CA 94608-1147

REBECCA LANGER **GRID ALTERNATIVES** 1171 OCEAN AVE. OAKLAND, CA 94611

ADAM BROWNING VOTE SOLAR 360 22ND STREET, SUITE 730 OAKLAND, CA 94612

ANNE DOUGHERTY OPINION DYNAMICS CORPORATION 1999 HARRISON ST. FLOOR 6, STE 650 OAKLAND, CA 94612

JASON B. KEYES KEYES & FOX LLP 436 14TH ST., STE.1305 OAKLAND, CA 94612 FOR: TESLA, INC.

JASON HUFFINE INFORMATION SYSTEMS MGR. II **ENERGY SOLUTIONS** 449 15TH STREET, STE. 400 OAKLAND, CA 94612

KATHERINE RAMSEY STAFF ATTORNEY SIERRA CLUB 2101 WEBSTER STREET, SUITE 1300

COUNSEL KEYES & FOX LLP 436 14TH STREET, STE. 1305

OAKLAND, CA 94612

OAKLAND, CA 94612

LAURENCE G. CHASET

RACHEL BIRD DIR - POLICY & BUS. DEVELOPMENT, WEST BORREGO SOLAR SYSTEMS, INC. 360 22ND STREET, SUITE 600 OAKLAND, CA 94612

CURRY STYPULA COMMERCIAL ENERGY 7677 OAKPORT STREET, STE. 525 OAKLAND, CA 94621

RON PERRY CE₀ COMMERCIAL ENERGY 7767 OAKPORT ST., STE. 525 OAKLAND, CA 94621

CHRISTOPHER GALLU TERRAVERDE RENEWABLE PARTNERS 110 LARKSPUR LANDING CIRCLE, SUITE 155 LARKSPUR, CA 94639

GREG SAN MARTIN PO BOX 5114 BERKELEY, CA 94705 GREG SAN MARTIN PO BOX 5114 BERKELEY, CA 94705

R. THOMAS BEACH PRINCIPAL CONSULTANT CROSSBORDER ENERGY 2560 NINTH STREET, SUITE 213A GALEN BARBOSE LAWRENCE BERKELEY NATIONAL LABORATORY

BERKELEY, CA 94710

1 CYCLOTRON RD., MS90R4000 BERKELEY, CA 94720

FOR: THE CALIFORNIA COGENERATION COUNCIL

GERALD T. ROBINSON LAWRENCE BERKLEY NATIONAL LABS ONE CYCLOTRON ROAD BERKLEY, CA 94720

RYAN WISER BERKELEY LAB 1 CYCLOTRON ROAD, MS-90-4000 BERKELEY, CA 94720

VINCE YOUNG UC BERKELEY ENVIRONMENTAL LAW CLINIC 356 BOALT HALL BERKELEY, CA 94720

RICK BROWN, PH.D PRESIDENT TERRAVERDE RENEWABLE PARTNERS 1100 LARKSPUR LANDING CIRCLE, STE. 155 LARKSPUR, CA 94939

NICOLE SANTOS SOLAR POWER PARTNERS, INC. 80 E SIR FRANCIS DRAKE BLVD., STE. 2A LARKSPUR, CA 94939-1709

ATTORNEY AT LAW JOHN NIMMONS & ASSOCIATES, INC. 175 ELINOR AVE., SUITE G MILL VALLEY, CA 94941-0000 FOR: ENPHASE ENERGY, INC.

JOHN NIMMONS, ESQ.

JASON SIMON DIR - POLICY STRATEGY **ENPHASE ENERGY** 1420 N. MCDOWELL BLVD. PETALUMA, CA 94954

NEIL MAGUIRE CEO AND FOUNDER JUICEBOX ENERGY, INC. 1650 CENTER POINTE DRIVE MILPITAS, CA 95035 FOR: JUICEBOX ENERGY, INC.

JIAN ZHANG CEO GRIDX, INC. MAHLON ALDRIDGE VP - STRATEGIC DEVELOPMENT

ECOLOGY ACTION

19122 BLUE LYNCH CT.
MORGAN HILL, CA 95037-9051

LORI MITCHELL
DIR - COMMUNITY ENERGY
SAN JOSE CLEAN ENERGY

200 E. SANTA CLARA ST., 14TH FL.

C.SUSIE BERLIN LAW OFFICES OF SUSIE BERLIN 1346 THE ALAMEDA, SUITE 7 NO.141

SHERIE BLAKE ALL N 1 INC. 5425 FRASCHINI CIRCLE SAN JOSE, CA 95136

SAN JOSE, CA 95113 FOR: SAN JOSE CLEAN ENERGY

SAN JOSE, CA 95126

WES MONIER
STRATEGIC ISSUES AND PLANNING MANAGER
TURLOCK IRRIGATION DISTRICT
333 EAST CANAL DRIVE, PO BOX 949
TURLOCK, CA 95381-0949

FOR: TURLOCK IRRIGATION DISTRICT

DUANE CAMPBELL UMASOLAR 499 EDISON CT., STE. A FAIRFIELD, CA 95434

RICHARD MCCANN, PH.D M. CUBED 2655 PORTAGE BAY, SUITE 3 DAVIS, CA 95616

GEORGE SIMONS
DIRECTOR
ITRON, INC.
330 MADSON PLACE
DAVIS, CA 95618-6599
FOR: ITRON

KENNETH SWAIN
MANAGING CONSULTANT - ENERGY
NAVIGANT CONSULTING, INC.
35 IRON POINT CIRCLE, STE. 225
FOLSOM, CA 95630

STUART CHANG ENERGY SYSTEMS DEVELOPMENT, LLC 5420 DOUGLAS BLVD., STE.F GRANITE BAY, CA 95746

ASISH GAUTAM
CALIFORNIA ENERGY COMMISSION
1516 NINTH STREET, MS-22
SACRAMENTO, CA 95814

877 CEDAR STREET, STE. 240 SANTA CRUZ, CA 95060-3938

LUISA F. ELKINS SR DEPUTY CITY ATTORNEY CITY OF SAN JOSE 200 EAST SANTA CLARA ST., 16TH FL SAN JOSE, CA 95113

IABBIE LAUGTUG
POLICY MANAGER
BLOOM ENERGY
4353 N 1ST ST, PO BOX 3785
SAN JOSE, CA 95134

NANCY FOLLY TURLOCK IRRIGATION DISTRICT PO BOX 949 TURLOCK, CA 95381

RICH BORBA JKB ENERGY 941 E MONTE VISTA AVE. TURLOCK, CA 95382

LINDA KELLY ELECTRICITY ANALYSIS OFFICE CALIFORNIA ENERGY COMMISSION 1516 9TH STREET, MS 20 SACRAMENTO, CA 95614

ANN PETERSON ITRON, INC. 330 MADSON PLACE DAVIS, CA 95618-6599

SMITA GUPTA SR. ENERGY CONSULTANT ITRON, INC. 330 MADSON PLACE DAVIS, CA 95618-6599

LEGAL DEPARTMENT
CALIFORNIA ISO
250 OUTCROPPING WAY
FOLSOM, CA 95630

CAROLYN M. KEHREIN PRINCIPAL ENERGY MANAGEMENT SERVICES 2602 CELEBRATION WAY WOODLAND, CA 95776

AUTUMN BERNSTEIN
DIRECTOR
CLIMATE PLAN
717 K STREET, SUITE 330
SACRAMENTO, CA 95814

DAN L. CARROLL
ATTORNEY AT LAW
DOWNEY BRAND, LLP

621 CAPITOL MALL, 18TH FLOOR SACRAMENTO, CA 95814

DAVID PEFFER ATTORNEY

BRAUN BLAISING SMITH WYNNE, P.C. 915 L STREET, SUITE 1480 SACRAMENTO, CA 95814

FOR: SILICON VALLEY CLEAN ENERGY AUTHORITY / SONOMA CLEAN POWER AUTHORITY / PENINSULA CLEAN ENERGY

AUTHORITY

DELANEY L. HUNTER
MANAGING PARTNER
GONZALEZ, QUINTANA, HUNTER & CRUZ, LLC
915 L STREET, STE. 1270
SACRAMENTO, CA 95814

JAN MCFARLAND CAEATFA

915 CAPITOL MALL, RM. 468 SACRAMENTO, CA 95814

JOHN SHEARS

CEERT

1100 11TH STREET, SUITE 311 SACRAMENTO, CA 95814

KEVIN WOODRUFF CONSULTANT

WOODRUFF EXPERT SERVICES, INC. 1127 - 11TH STREET, SUITE 514

SACRAMENTO, CA 95814

MATT KLOPFENSTEIN
GONZALEZ QUINTANA & HUNTER LLC

915 L STREET, SUITE 1270 SACRAMENTO, CA 95814 PAYAM NARVAND

CALIFORNIA ENERGY COMMISSION 1516 NINTH STREET, MS -45 SACRAMENTO, CA 95814

ROBERT PETERSON

CALIF PUBLIC UTILITIES COMMISSION INFRASTRUCTURE PLANNING AND PERMITTING B 300 Capitol Mall

300 Capitol Mall Sacramento, CA 95814 SCOTT MURTISHAW

SENIOR ADVISOR, REGULATORY AFFAIRS
CALIFORNIA SOLAR & STORAGE ASSOCIATION

1107 9TH STREET, SUITE 820 SACRAMENTO, CA 95814

TIM SCHMELZER

DIR. - LEGISLATIVE & REGULATORY AFFAIRS

WINE INSTITUTE 915 L STREET, STE. 1400 SACRAMENTO, CA 95814

FOR: WINE INSTITUTTE

THOMAS A. ENSLOW

ADAMS BROADWELL JOSEPH & CARDOZO 520 CAPITOL MALL, SUITE 350 SACRAMENTO, CA 95814-4715 FOR: IBEW-NECA LMCC

BRYAN NEFF

CALIFORNIA ENERGY COMMISSION 1516 9TH STREET, MS-20 SACRAMENTO, CA 95814-5512 DAVID VIDAVER

CALIFORNIA ENERGY COMMISSION 1516 NINTH STREET, MS-20 SACRAMENTO, CA 95814-5512

CHRISTIAN BRIGGS

ELLISON SCHNEIDER HARRIS & DONLAN LLP 2600 CAPITOL AVENUE, SUITE 400 SACRAMENTO, CA 95816 RONALD LIEBERT ATTORNEY AT LAW

ELLISON SCHNEIDER HARRIS & DONLAN LLP 2600 CAPITOL AVENUE, STE. 400

SACRAMENTO, CA 95816

GREGGORY L. WHEATLAND

ATTORNEY

ELLISON SCHNEIDER HARRIS & DONLAN LLP 2600 CAPITOL AVENUE, SUITE 400 SACRAMENTO, CA 95816-5905 DENNIS PETERS GOV'T AFFAIRS REP.

SACRAMENTO MUNICIPAL UTILITY DISTRICT

6301 S STREET, MS A311 SACRAMENTO, CA 95817

MARISSA O'CONNOR

ATTORNEY

SACRAMENTO MUNICIPAL UTILITY DISTRICT

6301 S STREET, MS B406 SACRAMENTO, CA 95817

MARTHA HELAK

GOV'T AFFAIRS REP, REGULATORY SACRAMENTO MUNICIPAL UTILITY DISTRICT

6201 S STREET, MS B404 SACRAMENTO, CA 95817

SCOTT BLUNK STRATEGIC BUSINESS PLANNER MICHAEL S. DAY PRINCIPAL

SACRAMENTO MUNICIPAL UTILITY DISTRICT 6301 S STREET, MS A311 SACRAMENTO, CA 95817 ROCKWOOD CONSULTING 2701 2ND AVE. SACRAMENTO, CA 95818

STEVE NEBOZUK 10060 GOETHE RD. SACRAMENTO, CA 95827

JAYSON WIMBLEY
. MGR. OF ENERGY AND ENVIRONMENTAL SVCS.
95827 DEPT. OF COMMUNITY SERVICES & DEVELOP.
2389 GATEWAY OAKS DRIVE
SACRAMENTO, CA 95833

RON THORN

VP - DIR OF SALES

RON THORN CONSTRUCTION INC.

4013 VISTA PARK COURT

SACRAMENTO, CA 95834

MARC DAVIS BARNUM & CELILLO ELECTRIC, INC. 3900 TAYLOR STREET SACRAMENTO, CA 95838

CHUCK SOLT LINDH & ASSOCIATES 7909 WALERQA RD., STE 112, PMB 119 ANTELOPE, CA 95843 KAREN LINDH
CALIFORNIA ONSITE GENERATION
7909 WALERGA ROAD, NO. 112, PMB 119
ANTELOPE, CA 95843

JIM BARNETT
SACRAMENTO MUNICIPAL UTILITY DISTRICT
PO BOX 15852, MS A203
SACRAMENTO, CA 95852-1803

RACHEL HUANG
SACRAMENTO MUNICIPAL UTILITY DISTRICT
PO BOX 15852, MS A353
SACRAMENTO, CA 95852-1803

WADE HUGHES
SACRAMENTO MUNICIPAL UTILITY DISTRICT
PO BOX 15852, MS A203
SACRAMENTO, CA 95852-1803

CRAIG ANDERSON
YOUTH WITH A MISSION
ADMINISTRATION
15850 RICHARDSON SPRINGS ROAD
CHICO, CA 95973

LIBERTY UTILITIES (FORMERLY CALPECO) LLC MIKE DOOLEY
933 ELOISE AVENUE VP OF MARKE
SOUTH LAKE TAHOE, CA 96150 AE SOLAR ENI

MIKE DOOLEY
VP OF MARKETING
AE SOLAR ENERGY, INC.
PO BOX 7348
BEND, OR 97708-7348

TOM ECKHART
CEO
CAL-UCONS
10612 NE 46TH STREET
KIRKLAND, WA 98033
FOR: CAL-UCONS

HEATHER CURLEE WILSON SONSINI GOODRICH & ROSATI 701 FIFTH AVENUE, SUITE 5100 SEATTLE, WA 98104

JENNIFER HOLMES ENERGY MARKET INNOVATIONS INC. 83 COLUMBIA STREET, SUITE 400 SEATTLE, WA 98104-1416 KURT SCHEUERMANN ITRON, INC. 603 OFFICERS ROW VANCOUVER, WA 98661

MATT HARPER AVALON BATTERY CORPORATION 1226 FRANCES STREET VANCOUVER, BC V6A 1Z5 CANADA

State Service

HELENA OH
CPUC
EMAIL ONLY
EMAIL ONLY, CA 00000

NICHOLAS FUGATE SUPPLY ANALSYS OFFICE CALIFORNIA ENERGY COMMISSION EMAIL ONLY EMAIL ONLY, CA 00000

SHERRILL NIEDICH RENEWABLE ENERGY OFFICE CALIFORNIA ENERGY COMMISSION

EMAIL ONLY EMAIL ONLY, CA 00000

ANNE E. SIMON CALIF PUBLIC UTILITIES COMMISSION

ADMINISTRATIVE LAW JUDGE DIVISION

ROOM 5118

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

ASAL ESFAHANI

XIAN MING LI

EMAIL ONLY

ORA - ELECTRICITY PRICING

EMAIL ONLY, CA 00000

CALIF PUBLIC UTILITIES COMMISSION DEMAND RESPONSE, CUSTOMER GENERATION, AN

CALIFORNIA PUBLIC UTILITIES COMMISSION

AREA

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

CHRISTOPHER HOGAN

CALIF PUBLIC UTILITIES COMMISSION ELECTRICITY PRICING AND CUSTOMER PROGRAM

AREA

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

EDWARD F. RANDOLPH

CALIF PUBLIC UTILITIES COMMISSION

ENERGY DIVISION ROOM 4004

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

EHREN SEYBERT

CALIF PUBLIC UTILITIES COMMISSION ADMINISTRATIVE LAW JUDGE DIVISION

ROOM 5041

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

ELIZABETH FOX

CALIF PUBLIC UTILITIES COMMISSION ELECTRICITY PLANNING & POLICY BRANCH

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

FOR: CAL PA (PUBLIC ADVOCATES OFFICE) FORMERLY THE OFFICE OF RATEPAYER

ADVOCATES (ORA)

GABRIEL PETLIN

CALIF PUBLIC UTILITIES COMMISSION DEMAND RESPONSE, CUSTOMER GENERATION, AN

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

JEANNE MCKINNEY

CALIF PUBLIC UTILITIES COMMISSION ADMINISTRATIVE LAW JUDGE DIVISION

ROOM 5015

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

KARI SMITH

CALIF PUBLIC UTILITIES COMMISSION INFRASTRUCTURE PLANNING AND PERMITTING B DEMAND RESPONSE, CUSTOMER GENERATION, AN

AREA

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

KERRY FLEISHER

CALIF PUBLIC UTILITIES COMMISSION

AREA

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

LISA PAULO

CALIF PUBLIC UTILITIES COMMISSION

ENERGY EFFICIENCY BRANCH

AREA 4-A

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

MARNA ANNING

CALIF PUBLIC UTILITIES COMMISSION UTILITY & PAYPHONE ENFORCEMENT BRANCH

AREA

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

PATRICK DOHERTY

CALIF PUBLIC UTILITIES COMMISSION ADMINISTRATIVE LAW JUDGE DIVISION

ROOM 5044

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

RADU CIUPAGEA

CALIF PUBLIC UTILITIES COMMISSION **ELECTRICITY PLANNING & POLICY BRANCH**

ROOM 4104

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

RAHMON MOMOH

CALIF PUBLIC UTILITIES COMMISSION UTILITY & PAYPHONE ENFORCEMENT BRANCH

AREA 2-E

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

SEAN A. SIMON

CALIF PUBLIC UTILITIES COMMISSION

COMMISSIONER RECHTSCHAFFEN

ROOM 5201

505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

SHANNON O'ROURKE CALIF PUBLIC UTILITIES COMMISSION PRESIDENT BATJER AREA 4-A 505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

TIM G. DREW CALIF PUBLIC UTILITIES COMMISSION ENERGY INFRASTRUCTURE BRANCH AREA 4-A 505 VAN NESS AVENUE

SAN FRANCISCO, CA 94102-3214

FRANZ CHENG CALIF PUBLIC UTILITIES COMMISSION MARKET STRUCTURE, COSTS AND NATURAL GAS 300 Capitol Mall Sacramento, CA 95814

LE-QUYEN NGUYEN CALIFORNIA ENERGY COMMISSION 1516 NINTH STREET SACRAMENTO, CA 95814

OSTAP LOREDO-CONTRERAS ENERGY SYSTEM RESEARCH OFFICE CALIFORNIA ENERGY COMMISSION 1516 9TH STREET, MS-43 SACRAMENTO, CA 95814

THOMAS ROBERTS CALIF PUBLIC UTILITIES COMMISSION ENERGY INFRASTRUCTURE BRANCH ROOM 4108 505 VAN NESS AVENUE SAN FRANCISCO, CA 94102-3214

GARY COLLORD STATIONARY SOURCE DIVISION CALIFORNIA AIR RESOURCES BOARD 1001 I STREET, PO BOX 2815 SACRAMENTO, CA 95812

LAWRENCE LINGBLOOM ASSEMBLY NAT. RESOURCES COMMITTEE 1020 N STREET, ROOM 164 SACRAMENTO, CA 95814

MATT COLDWELL CALIFORNIA ENERGY COMMISSION **ELECTRICITY ANALYSIS OFFICE** 1516 NINTH STREET, MS-20 SACRAMENTO, CA 95814

BRIAN MCCOLLOUGH **ENERGY ASSESSMENTS DIVSION** CALIFORNIA ENERGY COMMISSION 0203 9TH STREET SACRAMENTO, CA 95818

TOP OF PAGE BACK TO INDEX OF SERVICE LISTS

PG&E Gas and Electric Advice Submittal List General Order 96-B, Section IV

AT&T

Albion Power Company Alcantar & Kahl LLP

Alta Power Group, LLC Anderson & Poole

Atlas ReFuel BART

Barkovich & Yap, Inc.
California Cotton Ginners & Growers Assn
California Energy Commission
California Public Utilities Commission
California State Association of Counties
Calpine

Cameron-Daniel, P.C.
Casner, Steve
Cenergy Power
Center for Biological Diversity

Chevron Pipeline and Power City of Palo Alto

City of San Jose
Clean Power Research
Coast Economic Consulting
Commercial Energy
Crossborder Energy
Crown Road Energy, LLC
Davis Wright Tremaine LLP
Day Carter Murphy

Dept of General Services Don Pickett & Associates, Inc. Douglass & Liddell Downey & Brand
East Bay Community Energy
Ellison Schneider & Harris LLP
Energy Management Service
Engineers and Scientists of California

GenOn Energy, Inc.
Goodin, MacBride, Squeri, Schlotz &
Ritchie
Green Power Institute
Hanna & Morton

IGS Energy

ICF

International Power Technology Intestate Gas Services, Inc.

Kelly Group
Ken Bohn Consulting
Keyes & Fox LLP
Leviton Manufacturing Co., Inc.

Los Angeles County Integrated

Waste Management Task Force MRW & Associates Manatt Phelps Phillips Marin Energy Authority McKenzie & Associates

Modesto Irrigation District NLine Energy, Inc. NRG Solar

Office of Ratepayer Advocates OnGrid Solar Pacific Gas and Electric Company Peninsula Clean Energy Pioneer Community Energy

Redwood Coast Energy Authority Regulatory & Cogeneration Service, Inc. SCD Energy Solutions

SCE

SDG&E and SoCalGas

Tiger Natural Gas, Inc.

SPURR

San Francisco Water Power and Sewer Seattle City Light Sempra Utilities Southern California Edison Company Southern California Gas Company Spark Energy Sun Light & Power Sunshine Design Tecogen, Inc. TerraVerde Renewable Partners

TransCanada
Troutman Sanders LLP
Utility Cost Management
Utility Power Solutions
Water and Energy Consulting Wellhead
Electric Company
Western Manufactured Housing
Communities Association (WMA)
Yep Energy